

### **University of Huddersfield Repository**

Karolia, Mohammed

A study into the use of metaphors, the narratives they represent, and the application of the Clean Language questioning model as a method of reflection within an HE context.

#### **Original Citation**

Karolia, Mohammed (2022) A study into the use of metaphors, the narratives they represent, and the application of the Clean Language questioning model as a method of reflection within an HE context. Doctoral thesis, University of Huddersfield.

This version is available at http://eprints.hud.ac.uk/id/eprint/35752/

The University Repository is a digital collection of the research output of the University, available on Open Access. Copyright and Moral Rights for the items on this site are retained by the individual author and/or other copyright owners. Users may access full items free of charge; copies of full text items generally can be reproduced, displayed or performed and given to third parties in any format or medium for personal research or study, educational or not-for-profit purposes without prior permission or charge, provided:

- The authors, title and full bibliographic details is credited in any copy;
- A hyperlink and/or URL is included for the original metadata page; and
- The content is not changed in any way.

For more information, including our policy and submission procedure, please contact the Repository Team at: E.mailbox@hud.ac.uk.

http://eprints.hud.ac.uk/

MH Karolia



# A study into the use of metaphors, the narratives they represent, and the application of the Clean Language questioning model as a method of reflection within an HE context.

Mohammed Karolia

School of Education and Professional Development

University of Huddersfield

A thesis submitted to the University of Huddersfield in partial fulfilment of the requirements for the degree of Doctor of Education.

January 2022

### **Copyright Statement**

- The author of this thesis (including any appendices and/ or schedules to this thesis) owns any copyright in it (the "Copyright") and s/he has given The University of Huddersfield the right to use such Copyright for any administrative, promotional, educational and/or teaching.
- Copies of this thesis, either in full or in extracts, may be made only in accordance with the regulations of the University Details of these regulations may be obtained from the Librarian. Details of these regulations may be obtained from the Librarian. This page must form part of any such copies made.
- iii. The ownership of any patents, designs, trademarks and any and all other intellectual property rights except for the Copyright (the "Intellectual Property Rights") and any reproductions of copyright works, for example graphs and tables ("Reproductions"), which may be described in this thesis, may not be owned by the author and may be owned by third Such Intellectual Property Rights and Reproductions cannot and must not be made available for use without permission of the owner(s) of the relevant Intellectual Property Rights and/or Reproductions.

### Abstract

This thesis explores the relevance and significance of the metaphors used to convey a story and the broader narratives these metaphors represent, specifically within the context of reflective practice in higher education (HE). By situating the research in relation to traditional educative cyclical process models of reflective practice, this study identifies how alternative models of engagement could be used to promote academics' reflective practice, and how the metaphors employed to tell a story can give insights and inform narratives of teaching practice.

The study employed the Clean Language questioning model (Lawley & Tompkins, 2000; Sullivan & Rees, 2008) to encourage participants to recount their stories of teaching and elicit the metaphors representative of their teaching practice. The effectiveness as a research tool of the Clean Language questioning model, traditionally used in therapeutic contexts, was also evaluated as part of this study, given an emerging trend utilising Clean Language within research interviews. Eleven academics from different schools within a northern English university were interviewed to encourage stories of their teaching experiences. Furthermore, a participant questionnaire and a focus group discussion were used alongside the interview data to interrogate and substantiate the research conclusions.

The three strands of the study – metaphors, narratives and Clean Language questions – were synthesised via a conceptual framework, which informed the research's approach to analysis and findings. More specifically, content analysis coding of the interview transcripts was carried out to identify participants' metaphors against Kövecses (2010) list of source and target domains and evaluate the use of the Clean Language questions within the interviews. Thematic analysis of the data was also employed to gain an understanding of the broader narratives the conceptual metaphors might represent via frames of reference corresponding Tompkins & Lawley's (2006), to the Problem, Remedy and Outcomes (PRO) model and at a deeper level with McAdams and McLean's (2013) Narrative Identity Constructs and Schwartz's (2012) Universal Values Theory.

Findings established two dominant conceptual metaphors symbolic of HE academics' perception of their teaching practice: 'teaching is a journey' and 'teaching is a performance'. Participants corroborated the relevance and significance of these conceptual metaphors as representative of their focus on issues linked to learning and progression, and associated with

aspirations to teach well and improve aspects of their pedagogy. Additionally, the two key dominant conceptual metaphors espoused narratives representing exploratory narrative processing, agency and redemptive stories in relation to McAdams and McLean's (2013) Narrative Identity Constructs and embraced values of self-direction, stimulation and achievement when analysed for significance against Schwartz's (2012) Universal Values Theory.

The study's findings also determined the value of adopting a conversational storytelling approach to reflective practice, as opposed to cyclical process-driven models of reflection. More specifically, the value of identifying the dominant metaphors referred to by academics to reflect on their teaching practice was evident in giving insights into participants' pedagogical understanding of their teaching practice.

From a research methodological viewpoint, the positive potential of the Clean Language questions to elicit metaphors and encourage reflection emerged. Furthermore, the findings suggest that a limited range of Clean Language questions, in conjunction with reflective summaries and probing questions, could be used to promote effective reflective practice within HE.

From a theoretical perspective, the study suggested how the conceptual framework had relevance in synergising the study's three strands (metaphors, narratives and the Clean Language questioning model). More specifically, the findings also identified how an emphasis on exploring metaphors and analysing conceptual metaphors could assist in ascertaining broader narratives that might inform academics' pedagogy, especially within the environment of HE.

Practically, the outcome of the findings indicated an alternative approach and model to promote reflective practice. This model of reflective practice amalgamates and utilises a focus on exploring metaphors through a limited range of Clean Language questions to identify frames of reference and raise awareness of the broader narratives representing academics' approaches towards their teaching practice and pedagogy.

### Contents

COPYRIGHT STATEMENT	2
ABSTRACT	3
CONTENTS	5
LIST OF FIGURES AND TABLES	8
ACKNOWLEDGEMENTS	9
1. CHAPTER 1 - INTRODUCTION	10
1.1 Starting Point	
1.2 Stories and Narratives	11
1.3 Focus on Metaphors	14
1.4 Use of the Clean Language Questioning Model to Explore Metaphors	16
1.5 Research Questions	
1.6 Contribution to Knowledge	20
1.7 Structure of the Thesis	21
Chapter Summary	
CHAPTER 2 – LITERATURE REVIEW	24
2.1 EXPLORATION OF STORIES WITHIN THE CONTEXT OF REFLECTIVE PRACTICE	25
2.2 THE SIGNIFICANCE OF METAPHORS IN CONVEYING INSIGHTS AND UNDERSTANDING	29
2.3 Reading (the Narratives) Between the Lines	
2.4 The Use of the Clean Language Questions	45
2.5 Three Pieces of a Jigsaw – the Conceptual Framework and Chapter Summary	53
2.5.1 The key concepts	53
2.5.2 Relationships between the concepts	
2.5.3 The theoretical lens	
2.5.4 Integrating elements	
CHAPTER 3 – METHODOLOGY	60
3.1 Research Paradigms – The Starting Point	60
3.1 Research Paradigms – The Starting Point 3.1.1 Embracing a phenomenological approach to the study	

3.2.2 The research interviews	
3.2.3 Questionnaire evaluation of the interview and Clean Language questions	
3.2.4 Use of focus group	
3.2.5 Pilot surveys	
3.3 DATA ANALYSIS	76
3.3.1 Coding decisions	
3.3.2 Research validity	
3.4 Етніся	
Chapter Summary	
CHAPTER 4 – ANALYSIS OF FINDINGS INTO PARTICIPANTS' USE OF METAPHORS TO REFLECT	<b>F ON TEACHING</b>
PRACTICE	90
4.1 GENERAL OBSERVATIONS EMERGING FROM THE ANALYSIS OF FINDINGS	
4.2 Analysis of Source Domains	
4.3 Findings Target Domains	
4.4 DISCUSSION POINTS EMERGING FROM THE ANALYSIS OF FINDINGS	
Chapter Summary	
CHAPTER 5 – ANALYSIS OF PARTICIPANTS' USE OF METAPHORS AND THEIR NARRATIVES OF	F TEACHING105
5.1 FINDINGS OF NARRATIVE CODING ANALYSIS OF CONCEPTUAL METAPHORS	
5.1.1 Reporting on focus group reflections of conceptual metaphors identified	
5.1.2 Findings of analysis of PRO categories	
5.1.3 Findings of analysis of McAdams and McLean's Narrative Constructs	
5.1.4 Findings of analysis of Schwartz's Universal Values list	
5.2 DISCUSSION INTO THE SIGNIFICANCE OF NARRATIVES EMERGING FROM THE FINDINGS	
Chapter Summary	
CHAPTER 6 – ANALYSIS OF USE OF THE CLEAN LANGUAGE QUESTIONS WITHIN THE RESEAR	CH INTERVIEWS
	126
6.1 Analysis into the Use of Clean Language Questions	
6.2 Analysis of Interviewers' Conduct in the Use of the Clean Language Questions	
6.3 Analysis into Clean Language Questions' Ability to Promote Reflective Practice	
Chapter Summary	
CHAPTER 7 – CONCLUSION	139
7.1 REVIEW OF THE RESEARCH QUESTIONS	
7.2 Contribution to Knowledge	

7.3 AN EMERGING MODEL OF REFLECTION	147
7.3.1 Eliciting metaphors	
7.3.2 Identify conceptual metaphors	
7.3.3 Explore frames	
7.3.4 Links and transferability to practice	
7.4 LIMITATIONS AND CAVEATS OF THE STUDY	152
Concluding Summary	153
REFERENCES	155
APPENDIX 1 – CLEAN LANGUAGE QUESTIONS	167
APPENDIX 2 - PARTICIPANT CONSENT FORM	168
APPENDIX 3 – PARTICIPANTS PROFILE	172
APPENDIX 4 – PILOT INTERVIEW SAMPLE TRANSCRIPT	173
APPENDIX 5 – PILOT SURVEY SAMPLE RESPONSES	
APPENDIX 6 – ANLP CONFERENCE PRESENTATION POWERPOINT	
APPENDIX 7 – METAPHOR CODING CATEGORIES	
APPENDIX 8 – NARRATIVE ANALYSIS CODING CATEGORIES	
APPENDIX 9 – CLEAN LANGUAGE CODING CATEGORIES	
APPENDIX 10 - INTERVIEW TRANSCRIPT AND CODING (RESEARCHER 1 & 2)	
APPENDIX 11 – QUESTIONNAIRE FINDINGS	209
APPENDIX 12 – NVIVO ANALYSIS OF SOURCE DOMAINS	223
APPENDIX 13 – NVIVO ANALYSIS OF TARGET DOMAINS	224
APPENDIX 14 – ANALYSIS OF USE OF CLEAN LANGUAGE QUESTIONS	225

## List of Figures and Tables

Figure 1 – Summary diagram of Conceptual Framework	58
Figure 2 – Interview Transcripts Coding Process Summary	82
Figure 3 – Schwartz's (2012) Universal Values Framework	117
Figure 4 - Proposed Model of Reflection	148

Table 1 – Use of Research Methods and Analysis against the research questions
Table 2 - Examples of Coded Sense and Nature/Environment Domains
Table 3 - Examples of Coded Behaviour, Emotions and Environment Target Domains95

### Acknowledgements

First and foremost, I would like to give special thanks to my Supervisors, Professor Ann Harris and Dr Wayne Bailey, for guiding the journey of completing this thesis; a journey where I have had to complete the doctorate under the most challenging of circumstances. I cannot express my appreciation and gratitude enough for all your help and support – thank you.

Secondly, I would like to extend my thanks to Dr Rob Burton and Professor Kevin Orr for their encouragement, valuable insights and guidance in helping me to progress the thesis over the years.

Thank you also to the participants who agreed to contribute to the study and willingly gave their time to engage with the research, and Tamsin Hartley, who assisted in completing the research interviews, Andrea Norton for her assistance with the transcription of the interviews and Becky Ashwell for proofreading the thesis.

In particular, a special thanks goes to my friends within and outside of the university. You know who you are, for your endless stream of support and encouragement, for which I am forever grateful.

And finally, my parents and sisters, for their love and support, and for always being there for me through the peaks and valleys of life; and my partner Fatima for helping me to smile again.

This thesis is dedicated to my son Sufyaan, and all who have supported me in my time of need, and, more importantly, for being part of my story and narrative. Especially my son Sufyaan to remember that whatever the next chapter of our story holds, "you will never walk alone".

### 1. Chapter 1 - Introduction

#### 1.1 Starting Point

The foundation of the thesis stems from my voluntary counselling work and my interest in how clients reflect on and share very personal perspectives through analogies that can reveal how they have conceptualised their experiences. This process of recalling experience from a personal perspective within a counselling context recognises the distinction between what happened and the subsequent story of what happened from the individual viewpoint. As a higher education (HE) tutor, I inevitably considered how my wider experience might inform my professional practice and understanding by appreciating how teachers share and reflect on their stories as part of becoming a 'reflective practitioner' (Schön, 1983) in order to develop their practice. This symmetry between my voluntary and professional roles provided the initial stimulus behind the rationale for this study which explores how academics recount stories of their teaching practice with a particular focus on the metaphors employed to articulate their experiences.

However, what further informed my initial thinking was recognising that teachers are encouraged to tell their stories of teaching through the lens of traditional and established reflective practice methods, predominantly process-driven cyclical models of reflection developed by Kolb (1984) and Gibbs (1988). As an alternative, exploring the value of allowing participants to freely reflect on aspects of their teaching, with a research focus on exploring metaphors to encourage reflective practice, helped to mould the key research questions of this thesis. Within the context of the study, this translated into encouraging HE academics, through interviews, to reflect on aspects of their teaching practice and to analyse the relevance and significance of the metaphors they use in order to identify their narratives for teaching.

The research utilised a questioning technique, known as the Clean Language questioning model, to interview participants in order to encourage them to identify metaphors and to reflect on and explore symbolic representations of their teaching experiences. The origins of the Clean Language model emerged out of the work of the New Zealand-based psychotherapist David Grove (1950-2008), whose observations into the use of metaphors mirrored my own experiences of how clients, within my voluntary counselling work, often referred to analogies and metaphors to tell their stories of their experiences (Grove, 1996). Grove developed a set of questions, codified by Lawley and Tompkins (2000), and subsequently referred to as the 'Clean Language model', to encourage and allow clients to express and explore the metaphors

symbolic of their experiences to raise their self-awareness and conceptualise experiences. Over time, the Clean Language model has evolved and progressed from its origins in psychotherapy to being utilised as a personal development and coaching tool for individuals (Dunbar, 2017; Hartley, 2020) and teaching children (McCracken, 2016). More recently, the potential of Clean Language as a phenomenological research tool and interviewing method has been recognised by Lawley and Small (2018), Lawley (2017) and, Tosey (2011). To further explore and validate the use of the Clean Language model within a research context, this study employed the use of Clean Language questions when interviewing participants about their reflections on their teaching practice (see sections 1.4 and 2.4 for further details of the Clean Language model).

The study was located within a post-92 university in the north of England where a sample of 11 participants (see section 3.2.1 for further details of the sample surveyed), across a range of subject specialisms, were interviewed in order to elicit and analyse the metaphors to which they referred when reflecting on aspects of their teaching practice. Participants' comments were subsequently followed up and explored through post-interview questionnaires and focus group discussion.

#### **1.2 Stories and Narratives**

In our everyday lives, people tell stories about their life experiences. Stories about their triumphs and tragedies, their heroes and foes, their families, their friends, their thoughts and their feelings are conveyed through the tales we reflect upon within personal, professional, social, formal and informal situations. Gaining an understanding of how phenomena, as experienced by individuals, create stories that communicate and reflect their sense of experience forms the foundations of research into narratives. This is a feature of the study into academics' reflections into aspects of their teaching practice (Bold, 2012).

The allure and attraction of reflecting on and telling stories are historically resonant through oral and family histories. Lynch (1991), for example, sees the story as something which "grows out of life, reflects it and enters into dialogue with it. All life is in a story so that, there, we find our experience confirmed, challenged and broadened" (Lynch, 1991, p. 5); this is a notion Garnet et al. (2018) exemplify in relation to how Paulo Freire argued that the telling of stories could represent an act of empowerment by speaking up and by having our voice

listened to and respected by others. Ollerenshaw and Creswell (2002) contended that telling a story aids in enabling individuals to explore and understand their thoughts, behaviour and actions, while Bold (2012) further asserts that the act of narration gives individuals a creative licence to express emotions, and provides insights into the sense made of experience within the realm and context of the story. Regarding teaching, educators are regularly required to reflect on their practice to raise awareness and maximise opportunities to enhance practice by exploring stories of teaching and sharing experiences with peers and the wider teaching community (Moen, 2006).

Stories – the tales we tell of our experiences – evolve into narratives through a process of associations, reconstructions and interpretation. Storr (2020) succinctly summarises how the processing of experiences is associated and interpreted against previous experiences, thus creating and reinforcing patterns that form the narratives that can influence and guide thoughts, behaviours and actions. Therefore, the perception of an experience is a reconstructed and processed narrative, whereby differences exist between what happened and the interpretation of the story of what happened in relation to previous experiences. The associations between experiences and conceptualised memories allow us to think metaphorically and generate metaphors that convey how experiences are conceptualised. Recognition of this forms a bedrock of this study regarding how academics' articulation of their teaching experiences and, more specifically, within the context of this research, the metaphors they use to tell their stories, gives insights into their narratives for teaching.

To help locate this study within the fields of reflective practice, the process of recounting experiences as a story acknowledges the value and importance of storytelling to aid understanding in contrast to reflective practice encouraged by cyclical models of reflective practice (i.e., Kolb, 1984; Gibbs, 1988). Therefore, the process of telling stories of experiences is seen as the prime source of encouraging reflection via the research interviews and specifically exploring the relevance and significance of the metaphors referred to in telling a story in keeping with the research questions (see section 1.5) of this study.

Within this research into narratives and the study's context of academics reflecting on the stories of their teaching practice, consideration was given to acknowledge the differences between a story and a narrative. Squire et al. (2014), citing the work of early Russian linguists, refer to the terms 'fabula' and 'syuzhet' to distinguish between a story and a narrative, where a 'fabula' denotes the factual description of events in contrast to the reflective recalling of an

event signified by a 'syuzhet'. Within the context of this study, a working definition of narratives as 'developing an understanding of how metaphors (i.e., meaningful units) within a story aids in reflecting participants' narratives of their teaching practice' leans towards the notion of identifying stories as 'syuzhet' (Squire et al., 2014), and more specifically towards an analysis of the metaphors used to convey a story and the meanings people attach to a story, in order to make sense of and organise their lives (see section 2.3 for further discussions of definitions of research into narratives).

Analysis of narratives aids in giving voice to experiences by furnishing a story with culturally resonant meaning (Garnet et al., 2018). The appeal in researching narratives lies in gaining insights into how individuals create the stories which communicate and reflect their sense of the experience (Bold, 2012). The analysis and study of stories, within a research context, referred to as narratives, aids in developing and documenting an understanding of phenomena as experienced by individuals; a process which moves away from what might purport to be a factual account of a phenomenon towards recognition that the interpretive representation of phenomena, as experienced by individuals, actually reveals how experiences are conceptualised and acted upon (Fivush, 2008). Moen (2006) stresses how narratives, far from being abstract thoughts, are located by context, time and audience, and the narrator's agency. Narratives have an emergent quality in giving valuable insights and understanding into how individuals process phenomena by recalling and reminiscing events. Furthermore, the transformational potential of narratives in encouraging change through an exploration of the past, and by promoting a more empowering narrative of the future, establishes an aspect of this study whereby narratives may develop and impact teaching practice (Burchell & Dyson, 2005). Within the context of this study, the stories academics tell of their teaching experiences are viewed as acts of reflective practice. More specifically, the metaphors referred to convey a story are meaningful units used by individuals to help them make sense of the complexities of understanding experiences, which can further give insights into their narratives of their teaching practice.

Given the range of potential insights offered by a study of narratives, this research sought to establish, through its analysis of interview transcripts, to explore how microelements, such as the frame chosen to project metaphors, can influence thoughts and behaviour via Tompkins and Lawley's (2006) Problem, Remedy, Outcomes (PRO) frames, and McAdams and McLean's (2013) theory of Narrative Identities, specifically regarding the seven categories of Narrative Constructs identified by McAdams and McLean (2013). A further level of analysis

explored how the values fit within Schwartz's (2012) theory of Universal Values was utilised to identify the values emerging out of the academics' narratives of teaching and to identify potential congruence or conflict between Schwartz's (2012) 'openness to change', 'self-transcendence', 'conservation' and 'self-enhancement' categories of values.

#### 1.3 Focus on Metaphors

Exploring metaphors to encourage individuals to tell their stories formed a central tenet of this thesis. This section highlights the importance of exploring metaphors as linguistic phrases used by individuals, conveying deeper meaning and the implications embedded in what they wish to say. It draws upon Lakoff and Johnson's (1980; 2003) seminal work on the theory of conceptual metaphors, delineating the inherent and subconscious meaning contained within metaphors, especially within a teaching context.

Lakoff and Johnson (1980) initially promoted the broader significance and importance of exploring metaphors by their key text *Metaphors We Live By*, and the theory of 'conceptual metaphors'. In defining metaphors as "describing one thing in terms of another", Lakoff & Johnson (2003, p. 5) promote metaphors as a way of understanding an unfamiliar idea, event or state of affairs in relation to something more familiar, mainly when restraints imposed by language, vocabulary, context, emotions, cultural norms, etc. inhibit a person from openly expressing their thoughts or a phenomenon. This, they explain, is a process whereby metaphors referred to by a person to convey a concept gives insights into their cognitive schema and conceptual systems, which suggest how preference is given to one perception over another.

Subsequently, a range of authors (e.g., Grove, 1996; Battino, 2005; Kövecses, 2010; Geary, 2011; Gibbs, 2017), amongst others, have concurred and promoted the use and value of 'conceptual metaphors' to raise awareness of how a person's thoughts and schemas are constructed and structured. This broader value of exploring metaphors is reiterated and summarised by Saban (2006), who describes metaphors as:

"Far from being a figurative device or simply an elliptical simile, metaphors structure our perception, thoughts and actions. For better or worse, they fundamentally affect our way of conceptualising the world and reality whether we are aware of this phenomenon or not" (p. 299). Thus, metaphors, as patterns of thoughts linked to Lakoff and Johnson's (2003) work on 'conceptual metaphors', offer an alternative variable and unit of reflection more closely resembling and aligned with the natural thought processes of individuals. Consequently, exploration of the conceptual metaphors, representative of a person's underlying thought processes, gives metaphors a heuristic quality potentially worthy of study, and hence was chosen as a focal point for this thesis.

Various authors have further emphasised the fundamental essence of metaphors representing the broader cognitive processes at work when reflecting on and analysing experiences. For example, in support of Lakoff and Johnson's (1980; 2003) theory of conceptual metaphors, Mahlios et al. (2010) accentuate the significance of how a person's beliefs impact the metaphors individuals choose and use to describe and reflect on their experiences. Yero (2010) further emphasises how metaphors impact the "judgements and evaluations we make about ourselves, others and the world around us" (p. 28) and help give a person a sense of stability, consistency and predictability in the way they think and behave.

The use and role of metaphors as a method of reflection within a teaching context has been cited as having value. For example, an exploration of metaphors within the context of reflective practice taps into Schön's (1987) concept of 'appreciative systems' of how teachers frame and interpret their teaching experiences in relation to their values, beliefs, knowledge and prior experiences (Bullough et al., 1992; Pajares, 1992; Zeichner & Liston, 2014). Cook-Sather (2003) and Mahlios et al. (2010) further suggest that to better understand teachers' 'appreciative systems', it is necessary to research the metaphors representative of their teaching practice. Therefore, an analysis of the metaphors academics use can be employed to gain insights into the narratives underpinning their teaching practice.

In summary, this thesis examines how metaphors, far from being mere linguistic terms, are fundamental in mirroring individuals' underlying thought processes and cognitive schema (Lakoff & Johnson, 1980; 2003), their narratives (McAdams et al, 2006; McAdams & McLean, 2013) and values (Schwartz, 2012). They can be explored to identify how individuals conceptualise experiences and linked to Schön's (1987) theory of appreciative systems, within the context of this study, to encourage academics to reflect on their teaching experiences through the stories they tell.

# **1.4 Use of the Clean Language Questioning Model to Explore Metaphors**

This section briefly introduces the origins, history and theory of the Clean Language model to explore metaphors, before explaining why the model is used as an alternative framework, in place of process-driven cyclical models of reflection, to complete the research interviews and encourage participants to more freely reflect on their teaching experiences via discussion and the metaphors used to describe their teaching practice.

The origins of the Clean Language model emerged from New Zealand-based psychotherapist David Grove (1950-2008). Grove, while working with victims of trauma and traumatic memories, observed how:

"Many clients naturally described their symptoms in metaphor and found that when he enquired about these using their exact words, their perception of the trauma began to change" (Lawley & Tompkins, 2000, p. xiii).

This observation into the use of metaphors by clients gave rise to the early origins of the 'Clean Language' model, founded on the premise of how 'dirtying' a conversation by questions loaded with interpretations and assumptions (of the interviewer) negatively influenced the discussion and outcomes of conversations with clients. In contrast, allowing clients during counselling sessions both the space and opportunity to express themselves purposefully revealed their metaphors and exposed the underlying significance of these metaphors in raising self-awareness and comprehending experiences.

The Clean Language model was further developed by Lawley and Tompkins (2000), who on observing the work of David Grove, codified the set of questions (see Appendix one) Grove used when working with and analysing clients' metaphors. These questions form the cornerstone of the Clean Language model, and within this study will be referred to as the Clean Language questions, in contrast to the Clean Language model, which signifies a focus on wider use and explorations of metaphors within other contexts (see section 2.4 for further details of how Clean Language questions are distinguished from other forms of questioning participants within this thesis). While the roots of the Clean Language model had its origins in psychotherapy and the work of David Grove, the model has evolved and progressed from its origins, and, in addition to a coaching and personal development context, has been employed as a research method in social sciences and as a phenomenological research tool used by interviewers (Tosey, 2011). Within this study, the Clean Language questioning model

was specifically aimed at encouraging reflective practice within a teaching context to explore its depth and resonance as a qualitative research method within a HE environment.

The decision to utilise the Clean Language questioning model was reinforced by the four key strengths and features of the model. Firstly, the use of the Clean Language model in offering an alternative approach to reflective practice would counter criticisms of existing models of reflection, as identified by Boud and Walker (1998), Johns (2000) and Bulman and Schutz (2013), who highlight how popular models of reflection potentially diminish the quality and specific nuances of the way individuals reflect on their practice. The Clean Language model, in contrast, aimed to allow participants to freely reflect on aspects of their teaching practice and explore the intricate dimensions of an experience without the intervention or influence of the questioner/researcher. Secondly, intrinsic to the Clean Language model is the significance of the word 'clean' within the title of the approach, in contrast to a 'dirty' questioning approach whereby an interviewer may actively decode or subconsciously interpret the matters discussed and subliminally ask potentially leading questions as a result. Thirdly, the Clean Language model focus of enabling participants to steer the discussion and respond openly to interviewer questions would tap into Boud and Walker's (1998) suggestion of the need to adopt a 'coaching' approach to encouraging reflective practice. Finally, by the use of the Clean Language model to explore academics' reflections on their teaching practice, a novel and innovative reflection method may provide a more dynamic and flexible model of reflection more closely aligned to academic experiences.

A study of the literature suggests that much of writing about the Clean Language model originates from the experiences of the principal advocates of the model, as opposed to a critical analysis of the validity and application of the model. Hence, evaluating the efficacy and effectiveness of the Clean Language model as a method of reflection informed the third research question of the thesis. This is an aim that James Lawley, co-author of the key text *Metaphors in Mind* (Lawley & Tompkins, 2000), is keen to promote in giving the Clean Language model some rigour through doctorate level and academic studies into the use of the Clean Language model within a range of different contexts.

In summary, the Clean Language questioning model, despite its origins being rooted within the fields of psychotherapy and coaching, has more latterly been noted for its potential use as a research interviewing tool and is used within this study to promote a more conversational coaching approach to reflective practice and to explore the relevance of specific nuances of metaphors expressed by a participant in more depth.

#### **1.5 Research Questions**

Having explained the key strands underlining the rationale and motivation of this thesis, the research questions of the study focus on gaining an understanding of the use of metaphors referred to by academics while reflecting on aspects of their teaching practice and the broader significance of these metaphors, and an evaluation of the Clean Language questioning model as a research tool within a qualitative research context. Therefore, the specific questions which drive the research are:

- 1. What are the dominant metaphors used in the way academics recall, reflect and describe their teaching experiences?
- 2. What are the relationships between academics' use of metaphors and their narratives of teaching?
- 3. To what extent do the Clean Language questions allow participants to reflect on the metaphors representative of their teaching experiences?

Attention is also drawn in determining the relevance and significance of the metaphors expressed and conveyed by participants with regards to underlying narratives inherent in the metaphors expressed by participants. These metaphors are analysed to determine the deeper cognitive schemas and narratives that locate and inform academics' teaching practice.

The context of the study centred on the HE sector. This sector was chosen due to recognising how reflective practice is encouraged within the profession, though not a mandatory requirement within the sector. Chan and Lee (2021) specifically stress the lack of focus on reflective practice within HE via the following quote:

"Current literature on reflective practice has focused on its uses in primary and secondary education, as well as in teacher, medical and language education; little has been done in context of higher education" (p. 2).

The lack of emphasis on reflective practice within HE rallies against the proposed benefits of reflective practice in allowing academics to think, evaluate and examine their professional judgements, enhance teaching practice, student learning, and promote personal and professional development (Ashwin & Boud, 2020). Furthermore, the absence of the mandatory need to reflect on practice within HE has led to a lack of direction of how academics

could reflect on their teaching practice and result in academics referring to popular cyclical models of reflection (Kolb, 1984; Gibbs, 1988) to reflect on their teaching practice. This lack of direction in how to facilitate reflective practice within HE is further exasperated via a scenario whereby academics are trained as researchers and scientists who may not be required to complete pedagogical qualifications (Chan & Lee, 2021). However, the willingness of academics' commitment to research to promote effective teaching practices, my familiarity of working in the sector and the availability of a purposive sample of academics willing to participate in the study, aided in further promoting the motivation for focusing the study within a HE setting.

A purposive sample of 11 academics working across various disciplines (see section 3.2.1) within the HE sector was interviewed as part of this study. In keeping with the phenomenological approach, the use of interviews allowed participants to identify the relevance and significance of the metaphors they use to describe their teaching practice (DiCicco-Bloom & Crabtree, 2006).

With regards to the research questions, combined use of qualitative and quantitative research methods was used as detailed below:

- What are the dominant metaphors used in the way academics recall, reflect and describe their experiences of teaching? – data captured via the use of open interviews and analysis of interview transcripts
- What are the relationships between academics' use of metaphors and their narratives of teaching? – determined by analysis of findings and further corroborated via focus group discussion
- 3. To what extent do the Clean Language questions allow participants to reflect on the metaphors representative of their teaching experiences? – evaluated via analysis of questions asked by interviewers, the distribution of questionnaire after completion of research interviews and focus group discussion with participants

A more detailed discussion on the relevance and significance of the research methods used within the study can be found in section 3.2 of the Methodology chapter.

MH Karolia

#### **1.6 Contribution to Knowledge**

The key strands of this thesis – storytelling within reflective practice, metaphors, narratives and application of the Clean Language questions – make a contribution to knowledge in four ways:

This research investigates an approach to reflective practice not previously studied within the context of how HE academics could reflect on aspects of their teaching practice or how reflective stories and the metaphors referred to by academics within the stories of their teaching practice link to how they conceptualised their teaching experiences (see chapters 4, 5 and 7 for further details). Within the study, this emerges by identifying the dominant metaphors referred to by academics to reflect on their teaching practice and by interpreting the significance of these metaphors in terms of the narratives the metaphors represent within broader teaching and reflective practice contexts.

From a methodological viewpoint, the study evaluates the use of the Clean Language questioning model to complete qualitative phenomenological research interviews. Historically, the Clean Language model application has been traditionally entrenched within the fields of counselling, psychotherapy and coaching disciplines. While Lawley and Small (2018), Lawley (2017) and, Tosey (2011) recognised the potential of the use of the Clean Language model as a research method, further studies were required into its potential and efficacy within a research context. This thesis adds to the empirical evidence supporting the extended use of the Clean Language model in alternative contexts (see chapters 6 and 7).

Theoretically, the relevance of Lakoff and Johnson's (1980; 2003) theory of conceptual metaphors in the telling of stories (of personal experiences) within the realm of narrative studies reinforces the role of metaphors in conveying a greater understanding, specifically with regards to the work of McAdams (2006), and McAdams and McLean (2013) on narrative identities and represents a range of broader values and beliefs in relation to Schwartz's (2012) theory of Universal Values.

Practically, the study sees the development of a new model of reflection based on the research findings. This is a model of reflection based on empirical evidence of how academics conceptualise their experiences, that acknowledges the complexity and individual nature of the reflective practice, adopting a model of reflection that allows for the individual agency

through an exploration of metaphors and the use of the Clean Language questioning model (see section 7.3 for further details). The model of reflection would also help to provide a method of reflection and encourage reflective practice within a HE setting, where there is no obligatory requirement to reflect on practice.

#### **1.7 Structure of the Thesis**

This thesis comprises seven chapters. Each chapter includes an introduction that gives an overview of the chapter and how it contributes to the study, and ends with a summary discussion of the key themes and issues relevant to the research.

This introductory chapter establishes the rationale and context of the study. It details the research questions and ends with an overview of the structure of the thesis and its subsequent chapters.

Chapter 2, the Literature Review, identifies and discusses the most significant ideas and issues relevant to the research: storytelling, reflection, the use of metaphors, narratives and the Clean Language model. With these elements in mind, chapter 2 focuses on discussing 'the aim and role of stories and reflection in conceptualising experiences', 'the use of metaphors in understanding experiences' and 'exploration into the narratives behind metaphor', and the utilisation of the 'Clean Language' model as broad themes and subheadings of the Literature Review chapter. By adopting a questioning approach to exploring the key themes, the chapter evaluates earlier studies and discusses writing that identifies points of convergence and divergences in the literature relevant to the research.

Chapter 3, the Methodology chapter, discusses the underpinning philosophies, approaches, strategies and choice of research study methods. The chapter explores and details issues associated with research methodology, methods, validity, sampling, ethics and data analysis. The chapter summarises the relevance and appropriateness of the methods used to complete the study and how these could be evaluated against the research questions.

The examination of the findings and the analysis of data are discussed over a series of three chapters. Chapter 4 identifies the dominant metaphors participants used to recall and reflect on their teaching practice, focusing on determining the source and target domains of these metaphors. Chapter 5 discusses the relationships between academics' use of metaphors and their narratives of teaching by identifying the conceptual metaphors emerging from the

analysis of the data and the narrative frames the metaphors represent. Chapter 6 considers the effectiveness of the Clean Language questions in allowing the participants to reflect on their teaching practice in relation to the research findings. Within each of the findings chapters, data collected from both the one-to-one interviews, questionnaires and the focus group were considered together to stress the appropriateness and limitations of the significant findings of the study and, where relevant, to discuss the themes linked to the literature review in order to locate the findings within the broader context of the research questions and study.

Finally, chapter 7, the Conclusion, summarises the extent to which the study has effectively addressed the research questions and reiterates the nature of the work's original contribution. The limitations of the research are also discussed, in addition to an evaluation of the relevance and appropriateness of the methods used to complete the study.

#### Chapter Summary

In introducing the thesis, this chapter appreciates how the value of reflective practice and the importance of telling stories of experiences contrast with how teachers are encouraged to reflect on their teaching practice predominantly via process-driven cyclical models (e.g., Kolb, 1984; Gibbs, 1988). Specifically, within the HE sector, HE academics are not mandated to reflect on practice and hence lack direction of how to employ academic reflective practice approaches; a notion further expanded on by acknowledging how metaphors conveying a story could be studied as meaningful units that give insights into academics' narratives of their teaching practice and elucidate participants' frames of processing and teaching values.

The significance of metaphors in the communication of ideas and the pervasiveness of metaphors in everyday conversation is highlighted within the chapter and helped to inform the first research question, particularly with regard to Lakoff and Johnson's (1980; 2003) theory of conceptual metaphors being symbolic and representative of a person's cognitive schemas and conceptual systems. Within this study, the significance of metaphors within the reflective practice context, was explored to encourage HE academics, through interviews, to reflect on aspects of their teaching practice and determine their narratives for teaching.

The emergent quality of narratives' ability to give valuable insights into how individuals framed and processed phenomena within this study was the focus of the second research question. The exploration of the relationships between academics' use of metaphors and their narratives of teaching was explored by analysis of conceptual metaphors referred to by participants against Tompkins and Lawley's (2006) PRO frames, McAdams and McLean's (2013) theory of narrative identities, and Schwartz's (2012) theory of Universal Values.

The research specifically employed a questioning technique known as the Clean Language questioning model as a research method to interview participants to encourage them to reflect on their teaching experiences and explore conceptual metaphors symbolic of their teaching practice. The Clean Language questioning model was specifically developed to give clients the space and opportunity to express themselves and allow researchers to explore participants' metaphors via a codified set of questions (Appendix one) developed by Lawley and Tompkins (2000), and Sullivan and Rees (2008). To further explore and evaluate the use of the Clean Language questioning model within a research context, this study employed the use of Clean Language questions when interviewing participants about their reflections on their teaching practice and helped inform the third research question.

The three concepts of the study, the use of metaphors, analysis of conceptual metaphors to determine participants' narratives of teaching and the use of the Clean Language questions, helped structure the chapters of this thesis. The three concepts are referred to in the same order to formulate the research questions, align with the themes explored as part of the Literature Review chapter, and subsequently discuss the study's findings within chapters 4, 5 and 6.

### **Chapter 2 – Literature Review**

The following chapter explores and discusses the relevance and significance of the key concepts informing the thesis and the conceptual framework of the study. Four key concepts were identified as entities to review in order to address the research questions and to help locate the study's conceptual framework. Those concepts were:

- 2.1 Exploration of stories within the context of reflective practice
- 2.2 The significance of metaphors referred to within a story in conveying insights and understanding
- 2.3 The interpretation and significance of the narratives embedded within a story
- 2.4 The use of the Clean Language questioning model to locate and explore metaphors

These concepts will be referred to as themes to structure the following literature review. The first theme within the chapter (2.1) contends that stories are integral to how individuals reflect on teaching experiences and how recounting a story frames and conceptualises the way in which phenomena associated with teaching experiences are perceived and interpreted. The second theme (2.2) explores how metaphors help convey a story and are symbolic and significant in conveying the deeper aspects of the cognitive processes regarding how individuals reflect on their experiences. The third theme debates the use of narrative research to gain a broader understanding of how stories, as narratives, may impact a person's thoughts, feelings and behaviours (2.3). Section 2.4 reviews the application and relevance of the Clean Language questioning model as an integral element of the research method used to explore the metaphors uttered by participants during the research interviews, and later evaluated as the third research question of the study.

Finally, the chapter concludes with a discussion of how the above themes informed the conceptual framework and identified how the concepts investigated connected to the study and aligned with the research questions (Leshem & Trafford, 2007).

#### 2.1 Exploration of Stories Within the Context of Reflective Practice

Within everyday lives, people tell stories of their experiences, a common phenomenon whereby:

"We rely on stories to sort out the world ... storytelling is a universal habit, a part of our common humanity ... we discover in stories ways of saying and telling that let us know who we are" (Meek, 1991 cited in Burchell & Dyson, 2000, p. 437).

Through the stories of our experiences, individuals understand their lives (Garnet et al., 2018) as stories can act to raise awareness of a person's thoughts, feelings and behaviours (Ollerenshaw & Creswell, 2002). Furthermore, stories can covertly communicate a person's values, beliefs and understanding of the world around them (Clandinin & Connelly, 2000; Ollerenshaw & Creswell, 2002). These ideas of how stories of our experiences have more profound significance are of key relevance to this study and tie into the notion of how stories can confer meaning on people's lives and act as a 'narrative' through which they think, perceive, imagine, behave and make choices (Polkinghorne, 1989; Moen, 2006). Moen (2006) further elaborates this by saying how narrating our stories can be pivotal in helping individuals appreciate and understand their everyday experiences by providing a lens through which experiences are construed, interpreted, contextualised and conceptualised. These different facets of stories are further stressed by Lynch (1991), who suggests stories:

"grow out of life, reflects it and enters into dialogue with it. All life is in a story ... there we find our experience confirmed, challenged and broadened [by the stories of our experiences]" (p. 5).

In considering the importance of telling stories, this study is situated and focused on telling stories of experiences as an act of reflection of experiences. More specifically, how the metaphors referred to within the story have significance in informing academics' broader teaching narratives.

The recalling and telling of individual stories of teaching experiences are cited as a crucial ingredient needed to aid personal and professional development within education (Surgenor, 2011). The act of recounting experiences actively encourages practitioners to ponder, analyse and question their experiences and decisions, and a process often referred to as 'reflective practice' (Pollard, 2008; Ashwin & Boud et al., 2020). The value and need to adopt reflective practices within education has been put forward by Pollard (2008), who draws attention to how teachers, as part of their practice, are continually required to manage dilemmas in and out of

the classroom and ponder over the 'inner discomforts' they experience in their teaching, a term used by Brookfield (1987) to describe aspects of teaching practice that appear incongruent or problematic. Pollard (2008) also identifies how the need to raise selfawareness and explore values and beliefs as critical aspects of the reflective process, particularly by adopting an open-minded, reflective mindset, can lead to a more practical approach to enhancing teaching practice. Furthermore, the link between stories and reflective practice has been emphasised by a range of authors. For example, Schön (1988), a key proponent of the theory of reflective practice, emphasises the need to utilise stories as 'objects of reflection' rather than regard them as a by-product of reflection within the reflective process. As an act of reflective practice, stories have significance in enabling individuals to understand and conceptualise their experiences (Clandinin & Connelly, 2000; Ollerenshaw & Creswell, 2002; Bold, 2012). Within the context of this thesis, the link between the telling of stories of teaching experiences and reflective practice can be explained by Armstrong's (2020) analogy of scaffolding as a metaphor for locating storytelling within a reflective practice context. A scaffold, generally referred to as a structure used to aid the construction of buildings, can be considered as stories used to aid reflective practice. The process of recounting experiences as a story can be seen as the raw materials on which aspects of reflective practice processes can be understood in keeping with Schön (1988), who emphasises the need to utilise stories as 'objects of reflection', rather than regard them as a by-product of the reflective process. Furthermore, the process of recalling and the telling of individual stories of their teaching experiences moves away from following a structural approach to reflective practice advocated and encouraged by cyclical models of reflective practice (i.e., Kolb, 1984; Gibbs, 1988) towards a more naturalistic conversational approach to reflecting on experiences (Guo, 2021). Therefore, within this thesis, the process of telling stories of experiences is seen as the prime source of encouraging reflection, and, in keeping with the first research question of this study, exploring the relevance and significance of the metaphors referred to in telling a story.

The act of reflecting on and exploring storylines and narratives underpinning a story can help inform the way academics make decisions and lean towards particular ways of being and behaving. Jääskelä et al. (2017), in referring to agency as a way individuals can engage in autonomous or self-defined actions that are meaningful to them, help to identify a notion whereby the stories academics tell can inform and give insights into the choices, approach to teaching and ways of working with students they adopt within their teaching practice. Awareness and reflection into the stories that inform this cyclical process can act as a catalyst and be considered a learning process that transforms the way academics teach and adapt their own teaching practices (Ashwin & Boud, 2020). Thinking of stories as narratives that could be explored and examined to encourage greater awareness and agency is an idea in keeping with Pollard's (2008) suggestion of considering how teachers' teaching roles, perceptions and personal biographies can be enhanced via reflective practice, given the breadth of opportunities afforded by exploring stories and narratives as part of the reflective process. Within the scope of this study, of particular importance is how the stories of the experiences academics tell themselves can give insights into and inform their teaching practice.

To understand and appreciate the emphasis and importance placed on promoting reflective practice within education and the context of this study, it is necessary to acknowledge the early work of Dewey (2008) in the 1930s, which helped to establish the contemporary concepts of reflective practice and further reinforced by the work of Schön (1983; 1987), which helped to establish the importance of reflective practice in encouraging personal and professional development within education and other allied professions today. Dewey's (2008) fundamental distinction between 'routine action' and 'reflective action' provided the initial stimulus whereby reflective action formed a unit of analysis of experiences and the stories people tell. For Dewey (2008), reflective action is distinguished as a way of being where reason and emotion are engaged to holistically reflect on and respond to problems, instead of 'routine action' guided by tradition, authority or impulse. Within the context of this study, Dewey's concept of reflective action specifically focused on the use of metaphors referred to by individuals when reflecting on experiences and formed a key theme and concept of this study.

Since Dewey (2008) and Schön (1983; 1987) further popularised the concept of reflective practice, various authors have emphasised particular elements of the reflective process to promote reflective practice. Broadly, these approaches distinguish between components promoting routine/dialogic and critical reflection (Dewey, 2008; Schön, 1987); cyclical process-driven models of reflection (Kolb, 1984; Gibbs, 1988); and, finally, those encouraging a questioning approach to examine experiences (Brookfield, 1995; Revans, 2017). Of these approaches, the use of cyclical models of reflective landscape. This gives rise to two issues that need to be acknowledged and considered regarding this study: firstly, relating to adopting a structured/cyclical process-based approach to reflection, and, secondly, the quality of the reflections resulting from the use of these reflective methods. With regards to adopting a structured/cyclical process-based approach to reflection, the prevalence of models advocating

a sequential approach to reflection, such as those promoted by Kolb (1984) and Gibbs (1988), run contrary to Boud and Walker's (1998) criticism of process-driven models, which encourages a mechanical reflect on-demand approach, primarily as a solitary activity in keeping with rigid stages of the models to reflection, is at odds with the practice of reflection advocated by Schön (1987). Secondly, the pervasiveness of the reflect on-demand culture promoted by structured models of reflective practice restricts and impacts the quality of reflections due to the level of awareness and critical analysis needed to review and appraise experiences objectively at different stages of the reflective process (Surgenor, 2011). Additionally, several models of reflective practice also fail to acknowledge the value of dialogue and interaction, which, when purposeful, can maximise the learning to be gained via discussion with others (Hoffman-Kipp et al., 2003). Moreover, Bulman and Schutz (2008) identify how not acknowledging the specific distinctions and nuances of how people think and learn differently are further limitations of extant approaches and models of reflective practice.

These criticisms of cyclical models of reflection within this study are offset via a focus of moving away from adopting process-driven models of reflective practice promoted by Kolb (1984) and Gibbs (1988) to encourage participants to recount their experiences of teaching as stories and specifically explore the metaphors referred to in telling a story as an aspect of reflective practice. This process of encouraging reflection via stories is aided by the use of interviews and the Clean Language questions, whereby academics would be encouraged to freely reflect on their practice and allow participants to engage in dialogue with the interviewer, as proposed by this thesis. Furthermore, this study's focus, exploring the value of metaphors uttered within the stories academics tell of their teaching practice, encourages a move away and counters the use of cyclical process-driven models of reflection as articulated by Boud and Walker (1998) and Surgenor (2011).

In summary, the value of telling stories of experiences within this study is acknowledged in aiding understanding and, within a teaching context, is seen as an act of reflective practice. The ability for stories to confer meaning, whereby consistent storylines and narratives could be explored, can give insights into and inform practice, and helps feed into the first research question linked to exploring the relevance and significance of metaphors referred to in the telling of stories. This approach to the telling of stories as a way of scaffolding reflective learning can help counter criticism of traditional cyclical process-driven models of reflection.

#### 2.2 The Significance of Metaphors in Conveying Insights and Understanding

Studies into metaphors overlap across a range of disciplines. For example, over time, metaphors have been examined and researched for relevance and significance in the use of words, language, meaning, etc., across various contexts, including philosophy, psychology, sociology and cognitive linguistics. An integral focus of this thesis rests on the concept of metaphors being symbolically significant and representative of a person's thoughts, and suggests that, where metaphors are utilised to convey a story, they may give insights into the narratives of how experiences are interpreted and perceived (Carpenter, 2008). Within the thesis' context, this theme builds on the conceptual framework in discussing the relevance of metaphors as a key concept and corresponds with how the metaphors referred to by academics, while reflecting on their teaching practice, can give insights into their narratives for teaching. The theme briefly explores definitions and the prevalence of metaphors in everyday life, and in expressing thoughts in order to stress the importance of researching metaphors. The theme progresses to focus on considering how Lakoff and Johnson's (1980; 2003) seminal work into how conceptual metaphors mirror and give insights into individuals' conceptual systems is in keeping with the focus on researching academics' conceptual metaphors of their teaching practice. Finally, the theme concludes by exploring how theories of analysing and mapping metaphors informed the analysis of this study's findings and notes some caveats associated with researching metaphors.

The word metaphor can be traced back to the Latin word 'metaphora' and Greek word 'metapherein', with 'meta' referring to being beyond or over, and 'pherein' denoting 'bring or bear'. This etymology denotes how a metaphor can carry meaning beyond the literal and refer to a new or different meaning (Battino, 2005). A contemporary definition of metaphors from the Oxford Dictionary presents metaphors as: "A figure of speech in which a word or phrase is applied to an object or action to which it is not literally applicable" and "A thing regarded as representative or symbolic of something else". Therefore, in keeping with the historical roots of the word metaphor and dictionary definitions of metaphor, this study, via its research questions, explores how the metaphors referred to within stories symbolically represent academics' teaching practice and 'brings to bear' how they conceptualise their teaching practice.

The interest and importance of studying metaphors lie in appreciating the pervasive and ubiquitous presence of metaphors in everyday language, and metaphors' significance in giving

insights into an individual's conceptual systems. For example, Geary (2011) suggests how people could, on occasions, utter up to six metaphors a minute. This is a claim for which Tosey et al. (2013) concluded some evidence exists, as metaphors are more likely to be used when an individual, challenged by limitations of vocabulary, resorts to metaphors to draw on what is familiar to express aspects of experience that they may find difficult to articulate (Barry et al., 2009). Yob (2003) adds to this by drawing attention to how a metaphor might be employed when "one wants to explore and understand something esoteric, abstract, novel or highly speculative" (p. 134) and further emphasises how metaphors can simplify the understanding of concepts, facilitate communication, and rationalise and illuminate conceptual thought. Metaphorical analogies are significant in conveying how individuals perceive phenomena, rationalise how they make sense of the world within social and cultural contexts, and communicate this understanding within a personal and professional setting. The degree and frequency of the comparisons and associations conveyed by metaphors and the conceptualisations of their experiences can give insights into, inform and reinforce a person's thoughts, and generate new ideas. Geary (2011) specifically stresses this link between metaphor and thought via the following quotation:

"There is not an aspect of our experience not moulded in some way by metaphor's almost imperceptible touch ... Metaphorical thinking – shapes our view of the world and is essential to how we communicate, learn, discover and invent" (p. 3).

Geary (2011) here helps to emphasise an appreciation of metaphors' pervasive presence in everyday language. The significance of metaphors to convey a more profound understanding of a person's thoughts helped form the bedrock of this study into participants' use of metaphors. This is primarily with regards to exploring the value of conceptual metaphors' ability to raise awareness and give insights into participants' thoughts, schemas (Grove, 1996; Battino, 2005; Kövecses, 2010) and narratives (Moen, 2006; Squire et al., 2014) into their teaching practice; which, within this study, are linked to the second research question studying how elements and attributes of metaphors represent the broader narratives of how HE academics conceptualise their experiences of teaching (see section 1.3 for further details).

Initially, the interest in exploring metaphors centred on appreciating and analysing the language and prose of metaphors. This meant examining precisely how words and semantics conveying metaphors substitute, compare and offer analogies to understand concepts. Of fundamental interest to this study is the role of metaphors in conceptualising thoughts; a trajectory in the study of metaphors first muted by Ortony's (1979; 2008) publication of

*Metaphor and Thought*, marking a 'cognitive turn' wherein Ortony (1979; 2008) claimed metaphors are a matter of thought and create meaning, distinct from the everyday semantic meanings of metaphors. Ortony's (1979) initial work lay the foundations of postulating the role of metaphors signifying thoughts and was closely followed and furthered by Lakoff and Johnson's (1980) seminal work *Metaphors We Live By*, advancing the contemporary understanding of the role of metaphors and introduced the theory of conceptual metaphors as a way of understanding how thoughts are communicated via metaphors. Primarily, the work of Lakoff and Johnson (1980), introducing the role of conceptual metaphors, has been instrumental and referred to in defining and understanding the importance of metaphors in conveying an understanding of a person's conceptual system, and, within the context of this study, academics' perceptions and conceptualisations of their teaching practice.

Lakoff and Johnson's (1980) theory of conceptual metaphors posits the universal roles metaphors have in people's lives by stressing how:

"Not just in language but in terms of which we both think and act, is fundamentally metaphorical. The concepts that govern our thought are not just matters of the intellect. They also govern our everyday functioning, down to the most mundane details. Our concepts structure what we perceive, how we get around in the world, and how we relate to other people" (p. 3).

This pioneering view of the conceptual role of metaphors, a move away from the traditional linguistic views of metaphors, echoes the dominant discussions surrounding the role of metaphors today. Primarily, establishing the popular definition of metaphors as "describing one thing in terms of another", Lakoff and Johnson (1980, p. 5) identify how metaphors are used to simplify understanding of an unfamiliar notion or explain a state, events or a concept with aspects that are more familiar. They argued that metaphors, far from being a set of linguistic phrases, represent aspects of a person's thoughts. More importantly, Lakoff and Johnson (1980) claimed both thought processes and human conceptual systems are metaphorically structured and propose "that if we are right in suggesting that our conceptual system is largely metaphorical, then the way we think what we perceive and what we do every day is a matter of metaphor" (Lakoff & Johnson, 1980, p. 3). Therefore, Lakoff and Johnson suggest that the linguistic words and phrases used to express an idea through metaphor mirror and express the person's conceptual system and are fundamental in structuring how individuals think about abstract events and concepts (Gibbs, 2017). Saban (2006) further emphasised this view:

"Far from being a figurative device or simply an elliptical simile, metaphors structure our perception, thoughts and actions. For better or worse, they fundamentally affect our way of conceptualising the world and reality whether we are aware of this phenomenon or not" (p. 299).

Lakoff and Johnson's (1980; 2003) seminal work on conceptual metaphors has been instrumental in driving this thesis via recognising metaphors as being more than simply expressing one thing in terms of another (Lakoff & Johnson, 1980, p. 5) but also symbolic of a person's conceptual system, and also functioning as a lens through which preferred perceptions of phenomena are expressed and give preference to specific ways of thinking and behaving (Cook-Sather, 2003; Saban, 2006; Gibbs, 2017). However, in researching metaphors, it is important to consider the constraints of exploring and analysing metaphors, and the potential drawbacks of researching metaphors as caveats of this study. Gibbs (2017) asserts the need to acknowledge significant debates in the study of metaphors regarding the premise and controversy surrounding the link between the analysis of metaphors and inferences about a person's thoughts. For example, Weade and Ernst (1990) contend that a significant drawback of studying metaphors lies in how metaphors are selective. They represent a part, but not the whole, of the phenomena they describe (p. 133) and stress how metaphors only represent elements of an experience. Furthermore, while analysing the metaphors used by individuals can give insights into their thoughts and beliefs, Trick and Katz (1986) argue how the interpretation given to a metaphor, both by the person conveying the metaphor and the listener, depends on the meaning attached to the specific characteristics of the metaphor at a particular time, and question, therefore, whether the use of a specific metaphor has any more significance to the person employing them as they do to the researchers analysing and interpreting them. This is a view supported by Perrin (1987), who states how:

"Metaphors open us to experience in specific ways and closes us in others. It invites us to participate in the constitution of reality while at the same time barring us from the consideration of rival alternatives" (p. 265).

Hence, any analysis of metaphors within a research study requires careful consideration of the context and the interpretations attached to the metaphor to avoid presenting a lack of rigour in the analysis of metaphors. Carpenter (2008) elaborates by stressing how "at best, metaphors illuminate, at their worst, they distort and obscure" (p. 281) and result in "casting shadows" on findings potentially trivialising or betraying the essence of the phenomena being researched. Within this study, to help counter criticisms of exploring the significance of

metaphors, the interpretation of metaphors as part of the coding and analysis of the interview transcripts is safeguarded via the specific focus on identifying and mapping the metaphors referred to within the interview transcripts against Kövecses' (2010) list of 'source' and 'target' domains in preference to the subjective interpretation of a specific metaphor uttered by participants. Furthermore, participants verified the findings for relevance and significance individually, and via subsequent focus group discussion reflecting on the findings (further discussed within sections 3.2.4, 4.4 and 5.2). Additionally, identifying the narratives of academics' teaching practice emerging from the metaphors comprises a cumulative analysis of the primary conceptual metaphors identified from the analysis of the findings and not discrete scrutiny of the metaphors uttered by individual participants. Thus, this safeguards to some extent against criticism by Trick and Katz (1986), Perrin (1987) and Carpenter (2008) relating to the dangers of incorrectly interpreting metaphors conveyed by individuals.

In keeping with the conceptual theory of metaphors being symbolic of a person's conceptual system, by extension, academics' metaphors of their teaching experiences represent the conceptualisation of their teaching practice and give insights into their teaching narratives within this study's context. Similarly, the metaphors referred to by academics when reflecting on their teaching practice can frame and guide their approach to teaching (Martínez et al., 2001), where effective teaching practice assumptions mirror the language frame and metaphors chosen to convey experiences (Saban, 2006). For example, when an academic's conceptual metaphor of effective teaching mirrors and is framed around a 'factory' metaphor this conceptualisation of their teaching may perceive students as the 'raw material', the curriculum as production guidelines and assessments as output and guality control checks of the production line (Botha, 2009). Alternatively, Martínez et al. (2001) offer a conceptual metaphor whereby, if teachers framed their teaching practice as being a 'captain' in their teaching role, it might imply the presence of elements of authority and control in steering the direction of a ship (i.e., students). Within this study's context, this notion of conceptual metaphors' potential to frame academics' narratives of their teaching practice helped form the premise of this study and the first research question into identifying the dominant metaphors used in the way academics recall, reflect and describe their experiences of teaching, and relationships between academics' use of metaphors and their narratives of teaching.

Language, and how people use language, explicitly and implicitly communicate their underlying values and beliefs about the experience, and potentially, in relation to academics, their teaching practice (Yero, 2010). The value of exploring the language used to describe

experiences via metaphor, within the context of this study, specifically the metaphors referred to by academics to reflect on their teaching practice, is in keeping with Lakoff and Johnson's (1980; 2003) theory of conceptual metaphors' ability to communicate and convey the underlying conceptual systems which go on to inform academics' teaching practice. More specifically, Mahlios et al. (2010) further propose the need to research the metaphors representative of teachers' values and beliefs about their teaching to identify the larger constructs through which they organise their thinking and plan their teaching actions (Cook-Sather, 2003). Studies have further examined and reinforced the ability of metaphors to convey preferences towards ways of thinking. For example, Thibodeau and Boroditsky (2011, 2013, 2015) and Steen et al. (2014), investigating the role of metaphors and thoughts, researched how an individual's perception of concepts was influenced by the use of metaphors to describe the concept and concluded how evidence exists regarding metaphors' ability to guide the interpretation of a phenomenon. Within these specific studies, the example used to demonstrate metaphors' ability to frame and influence phenomenon referred to how participants' concepts of perceiving 'crime to be a beast' and 'crime to be a virus' influenced how the sample surveyed, viewed and responded with ideas of how to manage crime. These studies determined that those adopting the 'crime is a beast' metaphor resulted in participants identifying aggressive solutions to managing crime compared to the more progressive solutions put forward by participants framing crime as a virus. The notion of framing concepts, and the potential these frames may have on contextualising and conceptualising phenomena, have been adopted within this study with regards to analysing the interview transcripts to determine the frames adapted by participants while reflecting on aspects of their teaching practice (see section 2.3 for further details).

The anatomy of metaphors and how metaphors precisely "describe one thing in terms of another" (Lakoff & Johnson, 1980, p. 5) need examining to inform the analysis of metaphors as part of this study's research methodology, and in order to explain the coding decisions taken to analyse the interview transcripts (see section 3.3 of the Methodology chapter for further details). Understanding the inference contained within a metaphor and how one concept is understood in terms of another provides a starting point for understanding and analysing metaphors (Lakoff & Johnson, 2003; Geary, 2011). For example, Lakoff and Johnson (2003) initially referred to the term 'embodied metaphors' to describe the ways metaphors are conveyed via an analogy drawing on the notion of experiences being conceptualised in terms of senses and bodily perceptions (warm-cold, in-out, front-back, up-down, etc.). These embodied metaphors illustrate how individuals communicate their

understanding of phenomena regarding how they perceived and how they experienced the phenomena, and, of interest to this study, also insights into how experiences have been interpreted and understood (Gibbs, 2017).

The inferences contained within a metaphor were initially explored by Richards in 1936 in his The Philosophy of Rhetoric via the phrases 'tenor' (renaming of a subject) and 'vehicle' (analogy associated with the abstract thought being expressed). These terms help promote the concept of exploring the composition of a metaphor by mapping each one through the words and phrases used to convey a specific metaphor and comparing that with the concept being expressed in order to give insights into individuals' conceptual systems due to the range of possibilities of determining the inferences within metaphors, a well-defined process of mapping tenor, vehicle and insinuations of a metaphor needed to be established to examine metaphors as part of this study, and, further, given the challenge of determining how inferences are mapped in the analysis of metaphors, detailed within section 3.3 and elaborated in Appendix seven. Lakoff and Johnson (2003) further extend the need to examine metaphors, promoting the necessity to understand and analyse the words used to convey a metaphor. They refer to the term 'container' to describe the symbolic meaning attached to a metaphor by an individual as elements of a metaphor that could be explored and examined for relevance Kövecses (2010), building on the notion of metaphor 'containers', and significance. categorised the anatomy of a metaphor in terms of 'source' and 'target' domains, whereby 'source domain' refers to the variables (predominantly a concrete, physical or tangible element) used to describe a concept, and the 'target domain' (more abstract in nature) identifies the concept being conveyed by the source domain. In keeping with Lakoff and Johnson's (1980) definition of metaphors "describing one thing as another" (p. 5), the source and target domains describe the projection of one corresponding schema on to another. The designation of source and target domains is a process that can be mapped to aid understanding and analysis of a metaphor and simplify the process of dissecting the anatomy of a metaphor (Moser, 2000; Lakoff & Johnson, 2003; Kövecses, 2010). Within this study, the concept of source and target domains was referred to and used to map the metaphors in the interview transcripts when identifying the conceptual metaphors representative of participants' teaching practice. More specifically, the categories of 'source' domains identified by Kövecses (2010) consist of those pertaining to the use of the human body, health, animals, plants, buildings, machinery, sports, money, food, temperature, light/darkness, forces, movements, etc.
Similarly, Kövecses' (2010) target domains were categorised as those relating to emotions, thoughts, behaviour, relationships, communications, life/death, desire, morality, society, politics, economy, religion, etc. (see Appendix seven for a complete list of Kövecses' (2010) source and target domains used as codes within this study). References to Kövecses' (2010) 'source' and 'target' domains were integral in being referred to as the initial codes to isolate and map participants' conceptual metaphors within the interview transcripts and aided in determining the broader narratives of how HE academics conceptualise the narratives of their teaching experiences. Furthermore, the decision to refer to Kövecses' (2010) list of source and target domains rests on Lakoff and Turner's (1989) 'invariance principle', whereby the need to maintain the inference of a metaphor is preserved in the analysis of metaphors (cited in Gibbs, 2017). Within this thesis, given the range of variables available to analyse component elements of a metaphor, this aided in mapping the inferences of a metaphor between a significant range of source and specific target domains linked to the context of the study. However, recognition related to the difficulties in identifying, categorising and capturing the extensive range of source and target domain and metaphors uttered by individuals within the context is acknowledged. Therefore, the use of Kövecses' (2010) source and target domains as codes in the analysis of the data remained flexible to allow for the addition and analysis of supplementary source or target domains uttered by individuals not covered by Kövecses' (2010) original list of domains.

In summary, within this study, conceptual metaphors are deemed to be symbolically significant and representative of a person's thoughts and conceptual systems (Lakoff & Johnson, 1980; 2003), and within the context of this study, academics conceptualise their teaching practice. While criticism exists into the relevance of exploring and analysing specific metaphors (Trick & Katz, 1986; Weade & Ernst, 1990; Gibbs, 2017), the pervasiveness of metaphors in everyday life (Geary, 2011) and metaphors' ability to convey and inform insights (Cook-Sather, 2003; Saban, 2006; Mahlios et al., 2010) have informed the focus of this study.

In determining the inferences of the conceptual metaphors, Kövecses' (2010) list of source and target domains aided in simplifying the process of dissecting a metaphor and from which frames and narratives of conceptual metaphors referred could be determined.

### 2.3 Reading (the Narratives) Between the Lines

As discussed in section 2.1, stories are pivotal in helping individuals understand and appreciate the complexity of everyday experiences and have the capacity to confer meaning on people's lives. More specifically, of interest to this study is how the meaning attached to a story and the conceptual metaphors used to convey a story can act as a 'narrative' through which individuals think, perceive, imagine, interact, make choices and behave (Moen, 2006). Exploring stories and the metaphors used to express a story can give insights into how individuals interpret and frame experiences, and uncover the underpinning narratives influencing a person's conceptual systems and actions (Squire et al., 2014).

In keeping with research question two – "What are the relationships between academics' use of metaphors and their narratives of teaching?" – the following theme briefly explores how definitions of narratives apply to this study before surveying the different approaches to researching narratives and the need for approaches to researching narratives to be fit for purpose and correspond to the methods of analysing narratives employed by this study. In acknowledging the subjectivity associated with researching narratives, the theme concludes by examining how this study managed issues linked to the interpretation of working with narratives.

Given the depth and breadth of differences within approaches to researching stories and the narrative they convey, it is important to locate the definition of narratives identified with the research question in order to understand how narratives align with this study. Considerable diversity exists as to what a narrative is, resulting in a lack of consensus on what constitutes research into narrative stories. As a starting point, acknowledging the differences between a story and a narrative helps advance the understanding of the narratives embedded within stories, and this study's context and focus on exploring academics' reflections of their stories about their teaching practice. As initially discussed within section 1.2, Squire et al. (2014), citing the work of early Russian linguists, refer to the terms 'fabula' and 'syuzhet' to distinguish between a story and a narrative, where a 'fabula' denotes the factual description of events in contrast to the reflective recalling of an event signified by a 'syuzhet'. This distinction between a 'fabula' and 'syuzhet' aids in drawing a contrast between analysing the specific content of a story against the structure and substance of a story described by Squire et al. (2014), as organised, plotted, interpreted accounts of events and termed as a 'narrative'. The distinctions between a fabula and syuzhet tie with Bruner's (1990) notions of how individuals interpret personal experiences in terms of stories and two modes of thought, namely paradigmatic and narratives. Bruner (1990) recognised how experiences are comprehended through reasoned analysis, empirical observations and facts within paradigmatic modes of thinking. By contrast, narrative modes of thoughts and experiences are not explained in terms of facts but rather stories of intentions and actions. More recently, Kahneman (2012), in keeping with notions of differences between a fabula and syuzhet, and Bruner's (1990) concept of paradigmatic and narrative modes of thoughts, coined the phases 'system 1' and 'system 2' thinking to differentiate between automatic modes of thinking (i.e., system 1) against a thoughtful subjective approach to making choices and decisions of experiences (system 2). In taking account of and furthering the distinction between a 'fabula' and 'syuzhet', paradigmatic/narrative and system 1/2 modes of thoughts within elements of research into narratives, suggest the scope of research into a narrative's focus on understanding how individuals interpret and attach meanings to their stories. They tell of their experiences, broadly in keeping with Bolton's (2005) notion of how stories and the meaning attached to a story help individuals make sense of and organise their lives. Bold (2012), in providing a more contemporary definition of narrative, loosely defines narratives as being structured around events (phenomena experienced by individuals at a particular time and context) and reflecting the sense they and the narratives make of their lives. In the absence of a generally agreed definition of research into narratives, the common denominator evident within the above examples is the need to make sense of phenomena by analysing the critical role stories play in people's lives. Moen (2006) succinctly and concisely highlights this by asserting:

"As we make our way through life, we have continuous experiences and dialogic interactions both with our surrounding world and with ourselves. All of these are woven together into a seamless web, where they might strike one as being overwhelming in their complexity. One way of structuring these experiences is to organise them into meaningful units. One such meaningful unit could be a story, a narrative" (p. 59).

Within the context of this study, advancing Moen's (2006) notion of referring to 'meaningful units', and specifically stories and narratives as a way of making sense of complexities of understanding experiences, the metaphors used to convey the story of experiences are considered as the 'meaningful unit' worthy of consideration and exploration as part of this study's second research question into what the relationships are between academics' use of metaphors and their narratives of teaching. The focus on determining academics' narratives of teaching as part of this study further stresses the significance of exploring metaphors, as indicated within section 2.2, while the stories academics tell of their teaching experiences correspond to a 'syuzhet' (Squire et al., 2014) and incorporate elements of narrative modes of

thought (Bruner, 1990), and system 2 thinking (Kahneman, 2012). More specifically, in combining these elements, this study adopts a definition of research into narratives as focused on developing an understanding of how metaphors (i.e., meaningful units) within a story aid in reflecting participants' narratives of their teaching practice to analyse the significance of the metaphors referred to by participants within interview transcripts and the narratives they represent.

In keeping with the second research question, this study focuses on research into narratives centred on analysing data to determine academics' narratives of their teaching practice, in contrast, to the use of narrative research as a method of inquiry. In studying, analysing and interpreting narratives, approaches to determining narrative within research studies, as with other methods of inquiry, need to be fit for purpose and justified as an appropriate method and methodology to study a phenomenon (Clandinin & Connelly, 2000). Bold (2012) stresses that a process should be in the researcher's mind before commencing the research and requires a consistent approach to analysing data at all stages of the research (Riessman, 1993; Czarniwaska, 2004). However, this process is complicated by the extensive range of approaches and interests in researching aspects of a narrative. For example, approaches to research into narratives could include phenomenology, grounded theory and ethnography to understand the context of a phenomenon (Clandinin & Connelly, 2000).

Furthermore, the foci and interest in studying narratives could concentrate on examining a range of variables, for example, how individuals think and feel (Chamberlayne et al., 2000); the way narratives are performed, shaped and co-constructed for and by different audiences (Georgakopoulou, 2007); or how narratives are situated within a context or associated with social identities (Riessman, 2008; Salmon & Riessman, 2013). Given the diverse range of approaches and variables available to analyse narratives, Clandinin and Connelly (2000) call for and stress the need for narrative approaches to be fit for purpose. Therefore, within the context of this study, in adopting an interpretive and classical phenomenological approach to this study (see section 3.1.1 for further details) to answer Moustakas' (1994) question: "What have they (the participant) experienced in terms of the phenomenon?", was addressed by identifying the meaningful units (i.e., the conceptual metaphors emerging out of the interview transcripts), to interpret and determine the narratives representative of participants' conceptualisation of their teaching experiences.

A range of theoretical approaches to analyse and interpret narratives were considered as part of this study in keeping with Gudmundsdottir's (2001) notion of using theories and frameworks to safeguard against the open analysis and interpretation of data into a narrative. For example, initially, Lieblich et al.'s (1998) early work on identifying narrative elements (holistic, categorical, content and form) was deliberated on as possible codes to analyse the interview transcripts. Morson's (1994) concept of temporal coherence of narrative via the terms 'back shadows', 'side shadows' and 'foreshadowing' to denote how the telling of a story casts shadows over the temporal elements of a narrative was further considered with regards to how temporal coherences could inform broader influences of narratives. Given this study's links with the Clean Language model, Tompkins and Lawley's (2006) concept of framing metaphors as expressing a 'problem', 'remedy' or 'outcome' (PRO) was also reviewed as potential methods of identifying frames of reference conveyed by a conceptual metaphor. Elliott (2005), in advancing Morson's (1994) work on temporal coherence of narrative, further identified the 'meaningful' and 'social' elements of a narrative to denote the inherent complexity of how different facets of a narrative interweave in creating an understanding of experiences for both the narrator and audience for whom the narrative is intended were also considered as possible method of analysing and interpreting narratives. From a storytelling perspective, Booker's (2004) seven basic story plots and Dan Harmon's story circle (Fechter, 2021) were also briefly considered as possible approaches to determining narratives from the analysis of data. McAdams and McLean's (2013) work on 'narrative identities' was also evaluated as an approach to analyse the findings, given that 'narrative identities' theory focuses on determining the impact stories have on the thoughts and actions of individuals. Finally, Mahlios et al.'s (2010) notion of how metaphors represent overt and covert values and beliefs that teachers hold about themselves, and about aspects of their teaching practice, aided in reviewing the relevance of Schwartz's (2012) Universal Values Theory as an approach to determining the values expressed within participants' conceptual metaphors. In reviewing the relevance of these theories to this study, key considerations around the need for the theoretical lens to function as code to analyse and interpret data, identify inferences and form conclusions in keeping with research questions, acted as a filter in reviewing theories of narrative research that could be applied to this study.

After some consideration, McAdams and McLean's (2013) Narrative Identity theory was adopted as the primary theoretical lens to analyse the data to determine academics' narratives of their teaching practice. McAdams and McLean's (2013) theory of narrative specifically appealed and lent itself to this study due to the theory's ability to act as frames to identify and

solidify the link between the conceptual metaphors and the narratives they represent via the list of seven Narrative Constructs (to denote how narrative meaning can be constructed via an analysis of stories) helping to establish a coding strategy for the interview transcripts with regards to how metaphors framed participants' narratives of their teaching practice (see Appendix eight). Goffman (1974), an early pioneer of the theory of framing, described how the act of placing frames around concepts and arguments consciously and unconsciously layers a schema over the way experiences are perceived, interpreted and, of importance to this study, understood by individuals (Borah, 2011). This notion of identifying frames acting as schemas through which experiences are conceptualised, further ties with Thibodeau and Boroditsky's (2011, 2013, 2015) and Steen et al.'s (2014) studies into how people's perceptions of concepts were impacted by the use of specific metaphors to describe the concept (see sections 2.2 and 3.3.1 for further details). Additionally, the similarities in viewpoints between the theory of narrative identities and significance of reflective stories are evident within McAdams' (2006) quotation referring to narratives as "an internalised and evolving cognitive structure or script that provides an individual's life with some degree of meaning and purpose" (p. 11), drawing on the premise of how people naturally tell stories of their experiences that impact on their thoughts and behaviour (as previously discussed in sections 2.1 and 2.2 of this chapter). Within this storytelling process, narrative choices are made whereby individuals construct the story to fit their interpretations of the experience and existing cognitive schemas. The disparate shared stories aid in crafting a narrative identity in itself a schema – communicating to others who they are, how they came to be and where they envision themselves (McAdams & McLean, 2013), and are explored further within the findings of this thesis (see chapter 5). This premise regarding individuals constructing stories of their interpretations of experiences against existing schemas also corresponds to Lakoff and Johnson's (1980) theory of conceptual metaphors, whereby metaphors used to describe the experience are symbolic of deeper cognitive schemas and conceptual systems. Within this study, it is important to note how McAdams' (2006) theory of 'narrative identities' differs from traditional notions of studies into academic identities, which typically centre on studies and writing into being an academic and working in academia, and the dynamic complexities of working within the sector (Clegg, 2008; Billot, 2010). McAdams' (2006) use of the word 'identities' denotes a focus on the psychology of life stories, and how these convey to themselves and to others who they are, how they came to be and where they think their lives may be going in the future (McAdams & McLean, 2013). Within the context of this study, McAdams and McLean's (2013) theory of Narrative Identity is extended to denote how the

reflective stories academics tell of their practice are symbolic and representative in giving insights into aspects of their teaching practice and not notions of study into being an academic and working in academia. To avoid confusion with the use of the term 'identities' within this thesis, McAdams and McLean's (2013) theory of Narrative Identity Constructs will be referred to as 'Narrative Constructs' within the following chapters.

To gain a deeper understanding of the narratives emerging from the conceptual metaphors, in addition to McAdams' (2006; 2013) Narrative Constructs, further layers of analysis into participants' narratives of their teaching practice were considered with reference to Tompkins and Lawley's (2006) PRO model and Schwartz's (2012) theory of Universal Values helping to provide additional codes of analysis of the interview transcripts, and identify the frames and narratives representative of participants' teaching practice (see section 3.3.1 and chapter 5). The use of Tompkins and Lawley's (2006) PRO model assisted in identifying broad frames representative of the metaphors referred to by participants in a bid to determine how participants perceived and framed their experiences. Tompkins and Lawley's (2006) use of three broad variables, indicating a clear delineation of how individuals may construe their experiences as a conundrum, challenge, obstacle or set (i.e., problem), or a solution, answer or a resolution (i.e., remedy), or an aim, purpose, goal or ambition to achieve (i.e., outcome) aided in giving initial insights into how participants conceptualised their experiences and stories of their teaching practice. The decision to refer to Schwartz's (2006) Universal Values list was taken because the model identified a list of comprehensive recognisable values and the model's ability to understand the motivations impacting behaviour, thus allowing a deeper exploration into the participants' narratives of teaching practice.

In studying how stories and the conceptual metaphors referred to within stories inform narratives, three key issues impacting the inferences drawn from the analysis of narrative needed to be considered to help consolidate the conclusion drawn from the findings. Namely, issues linked to appreciating the expression of a narrative, the perception of a narrative and the interpretation involved in analysing academics' narratives of their teaching practice (Riessman, 2008; Squire, et al., 2014).

Regarding the expression of a narrative, Bold (2012), in coining the term 'representative construction', emphasises how stories are subject to decisions made by the narrator to present aspects of themselves and their story in keeping with their schemas and expectations, and, in doing so, present difficulties in determining the 'narrative voice' of the participants. The telling

of a story does not happen in isolation but in the audience's proximity and context where the narrative is conveyed and is further influenced by how aspects of the story are generalised and deleted due to memory limits. An individual's values, beliefs, previous experiences, levels of creativity and language further distort and create distance between facts and fiction of a narrative. Therefore, the analysis of narratives needed to consider the impact of factors linked to narrative expression and completed within the study by corroborating the findings with participants individually and via focus group discussions (see section 3.3 for further details).

In working with perceptions of narratives, it is important to note how stories in themselves are not representative of reality, an essential element of narrative research emphasised by Squire et al. (2014), who state: "your stories about your life are not the same as the life you live" (p. 110). Moen (2006), in citing Jerome Bruner, further distinguishes between a 'life lived' (what happened), a 'life experienced' (thoughts, feelings, images, sensations, etc.), and a 'life told' (narrative of the life lived to a particular audience and context) to underline the differences between what happened and the story of what happened, which Moen (2006) refers to as 'facts', 'fallacies' and 'fiction', whereby 'facts' refer to events believed to have occurred, 'fallacies' describe how those facts were lived and experienced by individuals and 'fiction' considered as how facts and fallacies are construed and woven together in keeping with a person's conceptual system and cognitive schemas. Therefore, when researching narratives, "there are blurred lines between what is said and unsaid, what is heard and not heard, what is analysed and not analysed" (Squire et al., 2014, p. 99). These 'blurred lines', within the context of this study, needed to acknowledge the differences between facts, fictions and the subjectivity linked to academics' stories of their teaching practice and taken into account by acknowledging how conclusion drawn from the analysis of the findings cannot be generalised to a broader population, but limited to giving insights into the narratives of the participant surveyed, and elude to how other academics may have similar perceptions of their teaching practice. The caveats associated with the differences between a 'life lived', a 'life experienced' and a 'life told' are further discussed as part of the conclusion of this study with regards to the narratives emerging from the findings.

The interpretation and analysis of narrative data are further complicated by the meaning attached to a story by both the narrator and those involved in examining the narratives to determine the dominant 'narrative voice' (Squire et al., 2014). Bold's (2012) use of the term 'representative construction' could equally apply to the role of the researcher analysing narratives, and how the narratives concluded from analysis of the findings by the researcher,

potentially could be distant from the original 'narrative voice' implied by the participant (Gudmundsdottir, 2001; Squire et al., 2014). Andrews et al. (2013), as mentioned previously, further stress the need for researchers to interpret and analyse narratives to guard against the tendency to identify cause and effect patterns by not acknowledging the complexities, ambiguities and differing versions of reality inherent in the narration or interpretation of narrative data. This subjectivity involved in the interpretation of narrative data has been cited by Gudmundsdottir (2001) and Moen (2006) as issues whereby narrative researchers need to validate the claims established in the analysis of the narratives (Bleakley, 2005) - especially concerning the influences and the co-construction of the narratives resulting from the analysis of the data. This is a process also referred to as 'restorying' by Ollerenshaw and Creswell (2002) to describe the practice of gathering stories and analysing data to construct a narrative representative of participants' experience in keeping with Squire et al.'s (2014) assertion of the need to identify the dominant internalised script of the stories. Sometimes it is also known as the narrative or 'master voice' permeating across the narratives and is considered to detail the narratives emerging from the research findings. With regards to how the gap between the expression, perception and interpretation of the story can result in a vast range of understandings of a narrative, the need exists to determine and clarify the interpretation of the narrative emerging from the analysis of the data by reference to a theoretical narrative framework to interpret and determine the narrative voice emerging from the findings (Gudmundsdottir, 2001; Squire et al., 2014). Within this study, this was managed by isolating the dominant conceptual metaphors emanating from the interview transcripts, which were explored in relation to Tompkins and Lawley's (2006) PRO model, McAdams and McLean's (2013) Narrative Constructs and Schwartz's (2012) theory of Universal Values frames, as discussed earlier to determine the relationships between academics' use of metaphors and their narratives of teaching.

In summary, the above themes identify how, in the absence of agreed definitions of narratives and research into narratives, this study draws on the distinctions between a 'fabula' and 'syuzhet' to distinguish between how stories, and the way a story is expressed via metaphors, can be studied to determine how academics make sense of their teaching experiences. Within this thesis, this study adopts a definition of research into narratives as focused on 'developing an understanding of how metaphors within a story (i.e., meaningful units) to aid in reflecting an understanding of participants' narratives of their teaching practice' to inform the second research question linked to determining relationships between academics' use of metaphors and their narratives of teaching. In considering the diversity and range of approach, and researching narratives and theoretical lenses to analyse and interpret narratives, McAdams and McLean's (2006) narratives identity theory was referred to due to the theory's ability to offer a method of both coding the interview transcripts and a method of interpreting the findings to determine the narratives representative and symbolic of metaphors referred to by participants, which was in keeping with Lakoff and Johnson's (1980; 2003) theory of conceptual metaphors, and relevant to the frames of references used to interpret the findings and to form conclusions. Furthermore, a deeper level of analysis and insight into frames of reference and narratives emerging from the findings was provided by Tompkins and Lawley's (2006) PRO model and Schwartz's (2012) Universal Values Theory.

In analysing narratives, key issues linked to appreciating the influence of perception, expression and interpretation of a narrative have all been considered and countered by isolating the dominant conceptual metaphors emanating from the interview transcripts, as well as a broad-based approach to determine the narrative voice via the use of an appropriate theoretical lens as discussed above.

### 2.4 The Use of the Clean Language Questions

This theme introduces the Clean Language questioning model to locate the use of the model within this study. Further to the introduction of the development of the Clean Language model within section 1.4 of the thesis, this section gives an overview of the model, explaining how the application and use of these Clean Language questions, specifically the Clean Language syntax, were addressed as part of this study. Finally, regarding the use of Clean Language questions within interviews, the use of Lawley's (2014) *Protocol for Validating the 'Cleanness' of an Interview* to evaluate the use of the Clean Language questions as part of this study and the third research question is reviewed and discussed within this section.

While the origins of the Clean Language model lay in the observations of David Grove's therapeutic work with clients, many of the contemporary advocates of the Clean Language movement have earlier links with the field of Neuro-Linguistic Programming (NLP) given the shared interest between the NLP and Clean Language model in working with and developing people. Hence, comparisons between Clean Language and NLP questioning models (O'Connor & Seymour, 1994; 2002), amongst other similar models such as Gendlin's (1982)

Focusing approach and Cooperrider et al.'s (2007) Appreciative Inquiry model, have been made. However, Clean Language is distinguished from other questioning models by focusing on the role of metaphors in analysing experiences and perceptions in ways other models do not.

Over time, the Clean Language model has evolved, developing theories associated with selfawareness and therapeutic change work. For example, 'Symbolic Modelling' (Lawley & Tompkins, 2000), 'Clean Space' (Lawley & Way, 2017) and the 'Listening Space' (Hartley, 2020) are amongst the more prevalent theories utilising the role of metaphors and Clean Language within personal development and therapeutic settings. In particular, Lawley and Tompkins (2000) developed the theory of 'Symbolic Modelling' as "a process which uses Clean Language to facilitate people's discovery of how their metaphors express their way of being in the world – including how that way of being evolves" (p. 23). Symbolic Modelling is a means of identifying and exploring an individual's 'metaphor landscapes' through Clean Language questions, based on the notion that all metaphors contain 'symbols' that have significance for individuals at both a conscious and unconscious level. The concept of 'metaphor landscapes' suggests that metaphors reflect mental models and a map of the world through which a person interacts with their everyday experiences, and it links to the theory of conceptual metaphors developed by Lakoff and Johnson (1980; 2003). The Clean Space (Lawley & Way, 2017) theory, in keeping with the Clean Language model, aims to remove the influence of the facilitator/interviewer within the conversation, but with the added element of utilising the physical space around a person as a further avenue from which to explore their metaphors and perception of the world. The 'Listening Space' model developed by Hartley (2020) aimed to combine the exploration of metaphors with mindfulness to raise selfawareness and experience of the world. While both 'Clean Space' and the 'Listening Space' have their origins within a therapeutic setting, they have more recently been utilised within a coaching context, though not as a research interviewing method. As the use of the 'Clean Space' and the 'Listening Space' models fall outside the remit of the research questions, they have not been considered or utilised as part of this study. However, the use of the Clean Language questions, originating from the work of David Grove and central to the Clean Language model, was of key interest in the study in a bid to explore the metaphors expressed by participants to help cover the first research question: "What are the dominant metaphors used in the way academics recall, reflect and describe their experiences of teaching?" Within the context of this study, the phrase 'Clean Language questions' is used to differentiate and distinguish how the study focused on the use of the Clean Language list of questions, and not the term 'Clean Language model', identifying the other strands of the model discussed above. Furthermore, in defining research boundaries, it is important to note how this study wholly focused on utilising the Clean Language questions within researching the interviewing context. Therefore, the emergence of additional branches of the Clean Language model, such as Clean Space (Lawley & Way, 2017) and Listening Space (Hartley, 2020), was not considered and is outside the remit of this study.

Although the Clean Language model and theory have their origins in psychotherapy and the work of David Grove, the model has since evolved and is popular within coaching contexts (Dunbar, 2017; Way, 2013) and as a facilitation method for working with groups (Walker, 2014), and, of particular interest to this study, as a social sciences research method and a potential phenomenological research tool (Tosey, 2011; Lawley & Small, 2018). Within this study, Clean Language's emphasis on metaphors was especially relevant to the first research question, probing: "What are the dominant metaphors used in the way academics recall, reflect and describe their experiences of teaching?" Specifically, the Clean Language model also promotes a conversational approach to reflection promoted by Hoffman-Kipp et al. (2003) and Guo (2021), and acknowledges how the presence of the metaphors conveyed within a discussion are symbolic in communicating conscious and unconscious thoughts associated with reflecting on a phenomenon (Lakoff & Johnson, 2003; Yob, 2003; Ho, 2005; Saban, 2006).

Central to the Clean Language model is the use of the Clean Language questions used by David Grove and latterly codified by Lawley and Tompkins (2000), and therefore these questions needed to be relevant and appropriate within the context of the research interviews. Loftus (2005), cited in Lawley (2014) via their research studies, identified how a question was asked and why the words used had overt and covert influences on the participants' memories. The view was that a change in a single word within the question could affect or 'contaminate' how a person would respond to the question. Thibodeau and Boroditsky (2011, 2013, 2015) concurred and expanded to suggest how the presence of a metaphor further influenced the frame of a question and impacted how participants responded to a question. The work of David Grove and the Clean Language model acknowledge the possible effects, and 'contamination' an interviewer's wording of a question can have on the participants and, therefore, the model aims to minimise, if not eliminate, an interviewer's influence and assumptions on the issues discussed. By encouraging interviewers and facilitators to listen and respond to participants' responses without allowing their reactions, judgement,

expectations, values and beliefs to impact the dialogue, the Clean Language questions allow participants to determine metaphors unfettered by Loftus' (1975, 2005) concerns. Furthermore, the Clean Language model also acknowledges how the metaphors conveyed within a discussion are symbolic in communicating the conscious and unconscious thoughts of phenomena and conceptual systems (Lakoff & Johnson, 2003; Yob, 2003; Ho, 2005; Saban, 2006).

However, diversity exists in the range and number of Clean Language questions promoted by various authors. For example, while Lawley and Tompkins (2000) identify nine core Clean Language questions, Sullivan and Rees (2008) promote 12 core Clean Language questions and the potential of 20 other 'specialised questions'. More latterly, Rees (2016) encourages the use of two lazy Jedi questions, with the term 'jedi' metaphorically representative of how the 'force' of the Clean Language questions can be channelled through the use of two dominant Clean Language questions (i.e., "What kind of X (is that x)?" and "Is there anything else about x?"). Alternatively Hartley (2020) identifies four questions linked to the new Lighthouse Coaching model to simplify the use and application of the Clean Language questions. Thus, suggesting how the use and application of the Clean Language model and questions is not an exact science and that practitioners are encouraged to be creative in using the model in relation to meeting the needs of those being interviewed within the context and aim of the specific interview. After some deliberations, Sullivan and Rees' (2008) 12 core questions were used to complete the research interviews as part of this study and the coding categories for the analysis of the interview transcripts, based on Sullivan and Rees' (2008) list of questions, offered a middle ground that categorised the questions into Developing, Sequence and Source, and Intention Questions (see Appendix one), which allowed for a more structured, yet straightforward way of employing the use of the questions during the research interviews. Furthermore, Sullivan and Rees' (2008) list of core questions mirrored the questions listed by Lawley and Tompkins (2000), and also Hartley's (2020) variations in the use of Clean Language questions.

While a definitive manual of how to apply the use of the Clean Language questions does not exist, the application of the model consists of the use of a range of Clean Language questions (Appendix one), repeated several times to direct participants' attention towards the metaphor used to convey an experience, allowing it to be explored while discouraging the inferences which might potentially emerge in the interviewer's responses to participants' answers. A process referred to as a 'syntax' (to denote an arrangement of phrases asked in a specific

48

sequence and manner) indicates where a strong preference exists within the Clean Language community in employing the Clean Language questions. The syntax comprises of the formula (as listed below), which an interviewer would follow in responding to what the participant has to say:

And ... (repeat clients' words and gestures) and when ... (some/select words and gestures) [ask a Clean Language question] (Sullivan & Rees, 2008, p.127)

The use of a specific 'syntax' can result in a mechanical approach to applying the Clean Language model, and, to mitigate this, there is a necessity for good rapport, while skilled use of the interviewer's tone of voice is encouraged to add variety to how Clean Language questions are asked. The use of the Clean Language syntax within a specific research context has been debated within the Clean Language community and in professional social media forums, particularly in relation to criticism of the repetitiveness in the way in which the syntax is employed and the recognition that this could result in a mechanical approach in the application of the Clean Language model. While a lack of agreement and consensus of the use of the syntax exists, Lawley (2014), as part of the social media discussion into the use of the syntax, acknowledged how a move away from the pure form of applying the syntax would make more sense. This is a view further elaborated by Meese (2014), who, as part of the same online discussion, stressed how the aim of the discussion/interview could have a bearing on the use of the syntax, for example, the use of the syntax in its pure form might be beneficial if the focus of the discussion was for the client to become more aware. More pragmatic use of the syntax could be employed where the aim of the discussion was for the researcher/interviewer to become more closely acquainted with the client's comprehension and understanding of experiences. Evidence from the current study's pilot interviews confirmed how employing the 'syntax' within the confines of a research interview could prove counter-intuitive in building rapport with the participants, and therefore was not employed within the research interviews completed as part of this study.

Within a research context, and regarding the third research question – "To what extent do the Clean Language questions allow participants to reflect on the metaphors representative of their teaching experiences?" – the need existed to identify measures to evaluate the effectiveness of the use of Clean Language questions. Lawley (2014), acknowledging the issues around evaluation of the use of Clean Language questions, presents criteria and a

*Protocol for Validating the 'Cleanness' of an Interview*. In the absence of other potential evaluation measures, Lawley's (2014) protocol was adopted by this study to determine the extent the questions asked during the research interviews were 'clean', given the detailed criteria proposed for determining the cleanness of a question and building on adopting a uniformed approach in measuring the effectiveness of Clean Language interviews. Lawley's (2014) protocol precisely consists of measuring Clean Language questions against the following criteria:

- **Classically clean:** a Clean Language question that substantially refers to the interviewee's words, without introducing the interviewer's own words, concepts, opinions, metaphors, etc.
- **Contextually clean:** a 'clean' question tailored to the interviewee's experiences and may introduce words related to the context and purpose of the research or interview.
- Mildly/potentially leading: ask questions and/or introduce words not used by the interviewee, are metaphorical or have a discernible impact on the interviewee.
- **Strongly leading:** interviewer uses words, concepts, opinions, metaphors, etc., not referred to by the interviewer and has a discernible impact on how the interviewee responds.

While there may be some subjectivity which exists in determining how accurately a question can be considered 'clean', the formulation of criteria, against which Clean Language questions can be measured, aids in validating the 'cleanness' of interviews regarding the questions asked, while acknowledging how the use of the Clean Language questions varies in relation to the context of the discussion. Considering the criteria as a continuum of Clean Language questions within a research or therapeutic setting, Lawley (2014) suggests that a rate of 90% (approx.) of questions asked should be either 'classically clean' or 'contextually clean' within an interview for it to be considered 'clean'. A measure of the cleanness of the interview against which the findings and analysis of the interview transcripts within this study were compared.

It is important to note, however, that Lawley (2014) is keen to stress how the *Protocol for Validating 'Cleanness' of an Interview* presents an attempt to establish an empirical research strategy using the Clean Language model and questions, and is not a defined criterion into the use of the Clean Language questions. Hence, this study interrogated Lawley's (2014) protocol alongside its other findings while acknowledging that a lack of empirical research appears to exist into the validity of the Clean Language model's use. Within a specific context, this study attempts to address this and to contribute to the developing research base around

the use of the Clean Language model across a range of contexts that is evident within the Clean Language community. This base is more evident within networking communities such as the 2014 and 2015 International Clean Language Conferences' 'Research Sandbox' events and, more recently, the annual online 'Metaphorium' forum founded by Judy Rees and subsequently facilitated by Sharon Small. Additionally, Lawley (2021) identifies nearly 100 publications that link to the use of the Clean Language model to establish further the validity of the Clean Language questions' use across a range of contexts. However, a cursory survey of these publications suggests how, amongst a selection of doctoral-level studies incorporating the use of the Clean Language model, much of what has been written about the Clean Language model originates from advocates of the model as opposed to a critical analysis of research into the validity and application of the model, especially within a research context.

In summary, in identifying the importance of exploring metaphors as part of this study, the Clean Language model, and specifically the Clean Language questions (see Appendix one), help to establish a framework for completing the research interviews given the Clean Language questions' focus and ability to identify and probe metaphors. More specifically, the use of the Clean Language questions as a research method helped to inform the third research question: "To what extent do the Clean Language questions allow participants to reflect on the metaphors representative of their teaching experiences?"

Given the Clean Language origins in therapeutic practice, adjustments are needed to use the Clean Language questions within the research interview context. This is specifically given the issue linked to the repetitive use of the Clean Language syntax, related to the requirement for interviewers to continually repeat specific phrases to elicit and explore participants' responses. Drawing on Lawley (2014) and Meese's (2014) discussions on using the Clean Language syntax, a more flexible use of the syntax was adopted during the research interviews completed as part of this study.

To address the thesis' third research question into identifying the value of using the Clean Language questions in enabling participants to reflect on the metaphors representative of their teaching experiences, in the absence of any other measures, Lawley's (2014) protocol was used to evaluate the research interviews' effectiveness as part of this study. By adopting Lawley's (2014) criteria, this study builds on the growing number of postgraduate and doctoral studies using the Clean Language questions in different contexts.

MH Karolia

# 2.5 Three Pieces of a Jigsaw – the Conceptual Framework and Chapter Summary

In studying aspects of the role and use of metaphors within reflective stories, narratives and the Clean Language questioning model within this chapter, a conceptual framework was developed and utilised to stress how the different concepts link together into a coherent study and summarise the Literature Review chapter (Thomson, 2018).

While differences exist as to definitions and explanations of what constitutes a conceptual framework, Ravitch and Riggan's (2017) observation of a conceptual framework comprising of "an argument about why the topic one wishes to study matters" (p. xv), and Miles and Huberman's (1984) definition of a conceptual framework consisting of "the researcher's map of the territory being investigated" (p. 33) succinctly explain the underlying purpose of developing a conceptual framework to underpin this study. Specifically, the relationship between key concepts and the theoretical lens employed to research these concepts, as well as how this informed parts of the research design, analysis of findings and interpretation of conclusions (Wald & Daniel, 2020), is discussed further below.

Taking into account Ravitch and Riggan's (2017) need for a conceptual framework comprising of "an argument about why the topic one wishes to study matters" (p. xv), it is important to acknowledge McAdams' (2009) notions of how a study into personality traits, behaviour characteristics, motivations, etc., while helpful, lacks insight and understanding of how individuals themselves have understood, interpreted and conceptualised their experiences, and how this in turn informed their learning and development. Therefore, this study into researching the conceptual metaphors symbolic and representative of academics' narratives of their teaching practices, sought to investigate and give an insight into understanding academic perceptions of their teaching practice within the context of storytelling, reflective practice and narratives in keeping with contributions to knowledge discussed in section 1.6.

### 2.5.1 The key concepts

Within conceptual frameworks, Wald and Daniel (2020) stress the need to go beyond definitions of concepts to determine the characteristics and attributes of the concepts to be examined. Three varied concepts were explored in this literature review and study: the metaphors embedded within reflective stories, the Clean Language questioning model, and narratives, all in keeping with the research questions. Within these three concepts, it was key to move beyond definitions and exploration, and acknowledge the role of metaphors in being

symbolic and representative of how academics interpreted and conceptualised their teaching experiences (Lawley & Tompkins, 2000; Kövecses, 2010; Geary, 2011). This could be accomplished, specifically, by drawing on Lakoff and Johnson's (1980; 2003) seminal work on conceptual metaphors signifying how metaphors represent a person's underlying schema and conceptual systems which relates to the aspect of the study informing the first research question: "What are the dominant metaphors used in the way academics recall, reflect and describe their teaching experiences?"

In determining the relationships between academics' use of metaphors and their narratives of teaching (research question two), the concept of 'stories' conferring meaning on people's lives and acting as a narrative (McAdams, 2006) has been key to informing how this study examines the significance of metaphors in conveying the deeper meaning of schema underpinning academics' teaching practice.

The focus on exploring participants' metaphors within the study was completed via the Clean Language model (Lawley & Tompkins, 2000; Sullivan & Rees, 2008) as a research interviewing tool given the model's specific focus and emphasis on probing metaphors. Thus, the opportunity to evaluate the Clean Language model with regards to its effectiveness in allowing participants to express and examine their metaphors formed the second concept of this study and feeds into the third research question: "To what extent do the Clean Language questions allow participants to reflect on the metaphors representative of their teaching experiences?"

### 2.5.2 Relationships between the concepts

From a theoretical perspective, considerations were given to how theories discussed within this chapter informed the relationships between the concepts, underpinned the research and acted as a lens to analyse the data in developing the conceptual framework.

The glue binding the three concepts of metaphors, Clean Language model and narratives together within this thesis consists of a belief or assumptions that connect the various ideas and notions (Miles et al., 2014). This, as mentioned previously, consisted of appreciating the value of metaphors as being symbolic and characteristic of a person's conceptual systems (Lakoff & Johnson, 1980; 2003). This is particularly significant regarding how stories of personal experiences, or within the context of this study, the metaphors used by academics

to reflect on their teaching practice, can give insights into how experiences are perceived and interpreted and inform future practice (Carpenter, 2008).

Within the conceptual framework, concepts linking to narratives follow Bolton's (2005) and Salmon and Riessman's (2013) notion of how stories and the meaning attached to a story help individuals make sense of and organise their lives. Within the context of this study, the significance attached to a story focuses on exploring conceptual metaphors, as the 'meaningful units' which Moen (2006) alludes to in identifying how individuals understand and structure experiences of teaching from an analysis of the interview transcripts (see sections 2.2 and 3.3 for further details). This process of analysing conceptual metaphors to identify the inherent narratives within them feeds into Mahlios et al.'s (2010) theory of how metaphors represent overt and covert beliefs teachers hold about themselves and aspects of their teaching practice (Yero, 2010), and hence gives insights into covering the second research question linked to identifying relationships between conceptual metaphors and the narratives of teaching they represent.

The narratives emerging from the conceptual metaphors aimed to draw on the premise of how people naturally tell stories of their experiences (as previously discussed in sections 2.1 and 2.2), and how the conceptual metaphors used to convey a story give insights into narratives inherent within academics' stories and reflections of their teaching experiences. Analysis of the conceptual metaphors to ascertain the narratives contained within them is further discussed below regarding the theories referred to as a lens to interpret and analyse the findings, and establish the narratives underpinning the conceptual metaphors referred to by participants to reflect on their teaching practice.

In locating the use and relationship of the Clean Language model within the conceptual framework, the metaphors expressed by participants were drawn out and explored further via the Clean Language questions (Lawley & Tompkins, 2000; Sullivan & Rees, 2008). The Clean Language questioning model, specifically developed to explore metaphors, initially within the therapeutic setting, and latterly as a qualitative research method used to complete the research interviews, was chosen given the model's novel ability to extract and explore the relevance and significance of metaphors referred to by participants when reflecting on their teaching practice (see section 2.4 for further details).

MH Karolia

### 2.5.3 The theoretical lens

An exploration and analysis of the inference within a metaphor to determine the key conceptual metaphors helps establish the starting point for identifying and analysing metaphors expressed by participants within the interview transcripts. The analysis of the metaphors elicited during the research interviews referred to the work of Kövecses' (2010) source and target domains list to identify the dominant metaphors used in the way academics recall, reflect on and describe their experiences of teaching in line with the first research question of this study. Principally, due to Kövecses' (2010) list presenting a structured process and mechanism through which to code and categorise the metaphors uttered by the participants as part of the content and thematic analysis of the data by isolating the expression of a metaphor (source domain) and the inferences they convey (target domains).

Interpretation and analysis of the relevance and significance of the key conceptual metaphors, to determine the narratives within them, were analysed through the theoretical lens of Tompkins and Lawley's (2006) PRO model, McAdams' (2006; 2013) Narrative Constructs and Schwartz's (2012) theory of Universal Values to identify the frames and narratives representative of and informing academics' conceptualisation of their teaching practice. In acknowledging the diversity existing within different narrative research approaches, as well as the need for narrative research studies to be fit for purpose (Riessman, 2005; Bold, 2012), this study adopted a three-pronged approach to the analysis of conceptual metaphors, which helped to situate the findings and establish generic narrative of academics' conceptualisation of their teaching practice (see section 3.3 and chapter 5 for further details).

Analysis and evaluation of the Clean Language questioning model, to cover the third research question, consists of coding the interviewers' use of Sullivan and Rees' (2008) list of Clean Language questions asked during the research interviews to elicit and probe participants' metaphors and determine the popularity of the Clean Language questions used by the interviewers. Furthermore, Lawley's (2014) *Protocol for Validating the 'Cleanness' of an Interview* was referred to in order to analyse the effectiveness of the Clean Language model within this study (see section 2.4 and chapter 6 for further details).

MH Karolia

### 2.5.4 Integrating elements

Leshem and Trafford's (2007) call to recognise the conceptual framework's role in integrating concepts to scaffold research and inform the research methodology and the broader contextual implications of the study was also considered in formulating the conceptual framework.

The theories described above were referred to and used to analyse the data, interpret the findings and help consolidate the conceptual framework. For example, Sullivan and Rees' (2008) list of Clean Language questions was used as the primary method of questioning participants as part of the research interviews and also coded against to determine the main Clean Language questions asked to assess the third research question probing the extent the Clean Language model allowed participants to reflect on the metaphors of their teaching practice. Kövecses' (2010) list of source and target domains was also referred to as codes to complete content and thematic analysis of the key conceptual metaphors representative and symbolic of participants' teaching practice in keeping with the first research question linked to identifying the dominant metaphors used by academics to recall, reflect and describe their experiences of teaching. Finally, Tompkins and Lawley's (2006) PRO model, McAdams and McLean's (2013) Narrative Constructs and Schwartz's (2012) theory of Universal Values were also used to code and analyse the conceptual metaphors emerging out of the analysis of the interview transcripts, and further referred to identify and interpret the narratives underpinning and informing participants' narratives of their teaching practice.

Within a wider context, this study is located within the role of storytelling as a mode of reflection and examines the possibility and feasibility of exploring conceptual metaphors (Lakoff & Johnson, 1980) to promote and aid approaches to reflective practice, specifically within the context of HE academics' reflections of their teaching experiences. In doing so, this study acknowledges how traditional approaches of reflective practice taught within education lean towards the use of process-driven cyclical models of reflection (e.g., Kolb, 1984; Gibbs, 1988) and have some limitations linked to not appreciating the dynamic way in which individuals learn (Boud et al., 1985), by adopting a one-size-fits-all approach to reflective practice (Johns, 2000), and requiring practitioners to reflect-on-demand (Boud & Walker, 1998). Limitations of traditional models of reflection countered by this study focus on exploring conceptual metaphors via a free-flowing conversational approach to reflecting on experiences. The key conclusions are drawn from the value elicited by probing metaphors within a reflective context and how the findings of the study potentially develop an alternative model of reflection in a move away from traditional process cyclical models of reflection.

Figure 1 below summarises the conceptual framework and how the three varied concepts connect to inform the research questions and give a theoretical lens to research the key concepts of the study.



#### Figure 1 – Summary diagram of Conceptual Framework

In summary, the study's conceptual framework can be understood by drawing on Thomson's (2018) metaphorical wall painting analogy to denote how specific elements of the study combine and link together within the thesis. Within the painting, this study, as an exercise in exploring reflective pedagogy approaches used to reflect on teaching practice, surveys the metaphors uttered by participants as central to painting with the finer details highlighting the nuances of the metaphors' source and target domains (Kövecses, 2010) to identify participants' conceptual metaphors. Furthermore, part of the painting shows how the Clean Language questions was used to elicit and explore metaphors referred to by participants within

the research interviews, and evaluated against Lawley's (2014) protocol. From a distance, the painting communicates and tells the story of the 'narratives' underpinning the painting and the stories told by the metaphors used by participants to reflect on their teaching practice. Specifically, in relation to Lawley and Tompkins (2000), McAdams and McLean (2013), and Schwartz's (2006) theoretical models discussed earlier to identify how HE academics frame aspects of the teaching practice and the narratives the conceptual metaphors convey. Finally, the study's finding identified how reflective pedagogy and academics' self-awareness could be further advanced and gave rise to the development of the model of reflective practice in keeping with the conclusions drawn from the study.

# Chapter 3 – Methodology

The following Methodology chapter details how issues linked to research paradigms, methodology, methods, sample and analysis were considered as part of this study.

The chapter commences by exploring how different research paradigms helped underpin the research methodology and methods adopted by the study (section 3.1). The chapter then progresses to detail the choice and relevance of the research methods chosen, the appropriateness of the sample and details of the pilot survey completed to trial elements of the research methods (section 3.2). The data analysis and coding decisions referred to in scrutinising the data against the research questions and issues linked to the validity and reliability of the findings follow in section 3.3. Finally, the importance of adhering to relevant ethical guidelines and principles to undertake a study responsibly and transparently are discussed within section 3.4.

### 3.1 Research Paradigms – The Starting Point

Miller and Brewer (2003) describe how the word paradigm, within research contexts, refers to how diverse research communities adopt different research methodologies and methods to complete a study in keeping with their preferred lenses for seeing the world. In determining a research paradigm consistent with the research questions, Burrell and Morgan (1992) suggest it is necessary to establish the 'ontological' and 'epistemological' positions wherein a study is situated (Sparkes, 1992, p. 12). Ontology, referring to an exploration of reality, determines explicitly whether the reality investigated exists externally to the individual or is a product of individual consciousness. It is categorised as being either realist - referring to the existence of a single reality/truth; or relativist - whereby multiple subjective realities are acknowledged and impacted by context (Sparkes, 1992; Saunders et al., 2012). On the other hand, epistemology examines reality in terms of how knowledge is understood and categorised as either positivist - leaning towards notions of knowledge being measurable via quantitative deductive approaches; to interpretive - whereby knowledge is subjective and influenced by context and likely to embrace inductive qualitative methods of research (Crotty, 1998). Given that the research questions in this study focus on exploring metaphors, referred to by individuals both consciously and unconsciously, a relativist ontology and interpretive epistemology for this study were adopted. Lakoff & Johnson (1980), in their key text Metaphors We Live By, debate the polar opposite between objective and subjective rationality

represented by metaphors, and reject the objective account of metaphors being solely linked to the literal aspect of language, and in place stress:

"Since the categories of our everyday thought are largely metaphorical and our everyday reasoning involves metaphorical entailments and inferences, ordinary rationality is therefore imaginative by its very nature" (p. 193).

Hence, Lakoff & Johnson (1980) contend how variables linked to everyday individual experiences, such as emotional responses, aesthetic appreciation and personal awareness, cannot easily be understood objectively, but via metaphors that convey and give insights into the subjectivity of the individual articulating the metaphor and understood by the process of creative comprehension by those interpreting the metaphor. Thus, by adopting a relativist ontology and interpretive epistemology, the fluid and dynamic use of metaphors by individuals in explaining concepts and phenomena they experience acknowledges how the notion of reality is subjective, intuitively constructed and shaped by the metaphors chosen, as well as being further impacted by the context in which metaphors are narrated to others. However, the variables inherent within metaphorical utterances may be identified, as indicated by Kövecses (2010), by the vehicle used to convey a metaphor (i.e., source domain) and by the message the metaphor intends to convey (i.e., target domain) (see sections 2.2 and 3.3 for further details).

By embracing a relativist ontology and interpretive epistemology, this study adopts an inductive methodology and a phenomenological research approach to address the research questions and to explore how the metaphors referred to by HE academics, while reflecting on their experiences, inform their teaching narratives. The decision to adopt a phenomenological research approach was taken after considering how other inductive research approaches (e.g., symbolic interaction, realism, etc.) did not lend themselves to studying lived experiences or bring to the fore the interpretations and perceptions individuals attach to their experiences and, potentially, their metaphors of teaching. Furthermore, phenomenological research focuses on the study of direct experiences, in contrast to an external objective analysis of an issue, and Lester's (1999) notion of how individuals can gain knowledge via reflection aided in reinforcing the use of a phenomenological approach to the interpretive research for this study in keeping with the research questions of the thesis. More specifically, adopting a phenomenological research approach tied with the focus and the research questions of the study, whereby academics were required to discuss and explore aspects of their teaching

practice, in keeping with definitions of phenomenological research which centre on the need to understand lived experiences of participants (Creswell, 2009).

### 3.1.1 Embracing a phenomenological approach to the study

In adopting an interpretive phenomenological approach to this study, consideration was given to the range of phenomenological strategies that exist to ensure that the approach chosen was fit for addressing the research questions. While Grbich (2012) initially describes how the work of Edmund Husserl gave rise to the 'classical phenomenological' approach to research, Creswell (2013) further identifies two other approaches to phenomenological studies, namely 'hermeneutic' and 'empirical' phenomenology, also referred to as 'transcendental' or 'psychological' phenomenology (Moustakas, 1994). The classical view of phenomenology advocates an exploration of phenomena with a group of individuals who have experienced a phenomenon via qualitative research methods, with the researcher 'bracketing' themselves out of the study to identify and classify meaningful quotations into themes to develop a 'textual description' of 'what' the participants experienced and a 'structural description' of how they experienced the phenomena (Moustakas, 1994; Creswell, 2013). Van Manen (1997) defined hermeneutical phenomenology as "oriented towards lived experiences and interpreting the 'texts' of life" (p. 4) and promoted a research process whereby the researcher actively interprets the lived experiences of the participants. Finally, empirical phenomenology focuses on 'describing' the lived experiences of the participants, where the researcher, in as far as possible, would 'transcend' the data captured in an attempt to ensure 'everything is perceived freshly, for the first time' (Moustakas, 1994, p. 34). For this study, the combined classical and hermeneutical phenomenological approaches to research are congruent with the research questions, in exploring how academics reflected on aspects of their teaching practice via qualitative research methods, with a specific focus on identifying the metaphors uttered by academics to determine 'what' and 'how' they 'experienced' their teaching practice. The classical view of phenomenology, in advocating an exploration of phenomena with a group of individuals who have experienced a phenomenon, aided in classifying and categorising meaningful metaphors referred to by participants into themes to develop a 'textual description' of 'what' the participants experienced and a 'structural description' of how the textual descriptions translated to convey conceptual metaphors of participants' lived experience of their teaching practice. This research approach also aligns with Creswell's (2013) suggestions of using qualitative research interviews, as adapted by the study, when researching lived experiences within a phenomenological research approach.

The key phenomenological question asked by Moustakas (1994), referring to identifying "what have they (the sample) experienced in terms of the phenomenon?" was addressed within the study and linked to the bracketing process promoted by Moustakas (1994) and Creswell (2013) using the Clean Language questions (Lawley & Tompkins, 2000; Sullivan & Rees, 2008), in keeping with the Clean Language model premise of how 'dirtying' a conversation by questions loaded with interpretations and assumptions (of the interviewer) could negatively influence the discussion and outcomes of conversations with participants (see sections 2.4 and 3.2.2 for further details). The use of the Clean Language questions further helped address the disadvantages associated with the use of phenomenological approaches, whereby the use of an open-ended inductive approach to qualitative interpretive research is cited as an issue whereby an impressionistic and subjective focus on the research may be misplaced due to the inherent dangers of misrepresentation and interpretations resulting from researcher biases impacting the eventual findings reported (Bryman & Bell, 2015).

Further limitations of using the phenomenological approach centre on the articulation and ambiguities existing within the language used to describe and reflect on a phenomenon (Ochieng, 2009; Noon 2018). Specifically, with regard to how participants' ability to identify and describe their experiences, express thoughts and feelings associated with an experience and the vocabulary referred to by participants could impact the findings of phenomenological studies. Within this study, the sample of experienced academics (see Appendix three) who were able to identify and confidently explain, examine, and deliberate over their teaching practice aided in the study's data collection process and assist in countering the limitations of articulating experiences within phenomenological studies.

Noon (2018) further notes how phenomenological studies are prone to tensions searching for representative connections and themes in analysing the data. An issue acknowledged within this study whereby the vast diversity and variety of metaphors participants could refer to while reflecting on their experiences needed to be considered. However, the use of specific codes linked to identifying metaphors and determining the inference of the conceptual metaphors helped to regulate the analysis of the findings in keeping the research questions (see Sections 2.3, 3.3.1 and Appendices seven and eight for further details)

Additionally, difficulties in replicating the research findings conducted via interpretive approaches due to participants' inferences associated with varying responses at any given time, the diversity of sample and context, and a lack of standards and measures to follow and

compare findings against, hinder in making generalisations from phenomenological studies. Therefore, it is important to stress how findings and inferencing emanating from this research cannot be generalised to a broader population and instances. However, the rigour in completing the research will stem from articulating the participants' narrative and their stories and reflections of their teaching practice, as discussed further in sections 3.3.2 and 7.4. In acknowledging how phenomenological studies may not facilitate generalisable finding, this study's value can be determined by the research contribution to knowledge listed in sections 1.6 and 7.2.

### 3.2 The Research Approach and Methods

In determining the methodological approach and use of research methods to complete this study, table 1 summarises the use of different research methods against the research questions, which are subsequently discussed further within this section.

Research Questions	Research Methods	Method of Analysis
What are the dominant metaphors used in the way academics recall, reflect and describe their teaching experiences?	Interviews Focus group	Content analysis of interview and Focus Group transcripts
What are the relationships between academics' use of metaphors and their narratives of teaching?	Focus group	Content and thematic analysis of interview and Focus Group transcripts
To what extent do the Clean Language questions allow participants to reflect on the metaphors representative of their teaching experiences?	Questionnaire Focus group	Content analysis of interview and Focus Group transcripts Quantitative Analysis of questionnaire findings

Table 1 – Use of Research Methods and Analysis against the research questions

Table 1 identifies how the data collection process utilised three research methods to help compile and analyse the findings. Firstly, the research interviews conducted with the participants and subsequently transcribed consisted of the primary data, which were coded according to the coding decisions described in section 3.3.1. The initial findings emerging out the interview transcripts were corroborated via completion of a focus group with the participants and completion of the questionnaire to determine participants' reflections of being interviewed as part of the study and the use of the Clean Language questions to complete the interview.

MH Karolia

In keeping with a phenomenological research methodology, the research methods used within this study consisted primarily of ideographic qualitative research methods, namely interviews and focus group discussions. A vital benefit of using ideographic qualitative research methods is the depth and richness of the data gained via interpretive approaches due to interviews and focus groups' ability to give the findings an emergent quality. Furthermore, interviews allow studies to be steeped within a context due to analysing the phenomena experienced by participants and giving the findings some authenticity by noting 'coherence' (see section 3.2.2 for further details). The use of a questionnaire, distributed to participants on completion of the research interview, was aimed at gaining an insight into participants' reflections of being interviewed as part of this study and corroborating the findings from the analysis of the interview transcripts (further discussed within section 3.2.3).

The study's data analysis stages systemically reviewed the interview transcripts to ascertain metaphors initially, code the meaning behind the significant phrases, and then identify the conceptual metaphors to determine how participants have experienced and reflected on their teaching practice. This is a process Moustakas (1994) termed 'horizontalisation' to stress the researcher's attempts to develop 'clusters of meaning' of significance and to form 'textual descriptions' of participants' experiences and a 'structural description' of the phenomena, leading towards a composite description of the 'essence' (also referred to as essential, invariant structure) of the phenomena (see section 3.3 for further details). Specifically, with regards to the analysis of the data emerging from the use of the research methods against the research questions, open interviews were completed and analysed via content analysis, against Kövecses' (2010) source and target domains (see Appendix seven) of the metaphors referred to by academics to identify the dominant metaphors referred to within the interviews to cover the first research question. The second research question - "What are the relationships between academics' use of metaphors and their narratives of teaching?" - was determined by content and thematic analysis of interview transcripts, against Lawley and Tompkins' (2000) PRO model, McAdams and McLean's (2013) Narrative Constructs and Schwartz's (2012) Universal Values list, and further corroborated via a focus group discussion convened after coding analysis of interview transcripts. Finally, the third research question aimed at identifying "To what extent do the Clean Language questions allow participants to reflect on the metaphors representative of their teaching experiences?" consisted of content analysis of interview transcripts against Sullivan and Rees' (2008) list of Clean Language questions, a questionnaire distributed to participants after the research interviews and further followed up by a focus group discussion with participants to discuss their reflections and evaluate their participation in the interviews.

### 3.2.1 Sampling

In determining the sample for a study, Robinson's (2014) holistic approach to considering a range of variables aided in identifying a suitable sample of participants to contribute to the study. These variables consist of considering the sampling universe (population), sample strategy, size and source of the sample.

Initially, with regards to Robinson's (2014) suggestion to determine the sampling population and strategy, the decision to select a homogenous sample of similar participants (i.e., academics teaching within a HE setting), in preference to a heterogeneous diverse sample, was taken as an inclusion criterion in keeping with Smith et al.'s (2009) assertion that phenomenological research needs to remain contextualised within a defined setting. Hence, regarding this phenomenological research, the sampling strategy consists of adopting a purposive sampling approach to survey participants who had experience and were able to reflect on the phenomena to be examined. Therefore, Yardley's (2000) assertion of the need to ensure the suitability and validity of the sample chosen was considered by only interviewing participants who were employed as lecturers or senior lecturers in a post-92 university across a range of disciplines (education, health and business) and were willing to participate in the study as demonstrated by the completion of the Participant Consent Forms (see Appendix two). Sampling criteria linked to age, gender, location, institution, etc. were not considered in the selection of participants as the research questions and the methods of analysis did not distinguish or require a comparison of these variables.

In determining the sample size for this study, Creswell (2007) proposes a sample size of six participants for phenomenological studies, while Morse (2000) suggests sample sizes of six to ten participants as being practical and credible when completing phenomenological research studies. Therefore, the selection of a sample of 12 participants to complete this study exceeded Creswell (2007) and Morse's (2000) recommendations in order to allow for possible attrition of participants and to ensure sufficiency of data to be collected and analysed without losing the individuality required for phenomenological studies. The 12 lecturers initially approached to participate in this study matched Polkinghorne's (1989) requirement to select individuals who have all experienced the same phenomena when completing phenomenological research studies, were knowledgeable and able to openly and candidly

discuss and reflect on their teaching practice in keeping with the criteria put forward by Yardley (2000). Furthermore, Marshall et al. (2013) recommend identifying a sample size in keeping with the time and resources available to complete a study. Therefore, of the initial sample size of 12 participants, only 11 participants were interviewed and followed up via a focus group discussion as part of this study because of practical limitations. However, concerns linked to interpreting findings from surveying a small sample of 11 participants need to be acknowledged (Mason, 2018). Given the sample size and the research focus on exploring participants' metaphors, the research findings cannot be generalised or deemed to apply to a broader academic population due to the number of participants and the diversity and range of metaphors that might emerge when conveying a phenomenon. This is an issue that needs to be borne in mind regarding the conclusions made from the findings (Marshall et al., 2013) and acknowledged as a caveat and a limitation of the findings, which will be further discussed and addressed as part of the conclusions within chapter 7.

In sourcing the sample to participate in the study, colleagues and associates known to the researcher were initially approached to participate in the research, in line with the purposive sampling strategy used to determine the sampling universe and size. Further participants were located via chain-referrals, whereby existing participants were invited to identify additional colleagues and acquaintances who may welcome the opportunity to participate in Considerations pertaining to ensuring participants were fully aware of the the study. implications of being surveyed as participants in a study and the ethical boundaries governing their involvement are further detailed in section 3.4 of this chapter. Appendix three details the profile of the sample of the academics who agreed to participate in the study regarding their subject specialism, level of teaching experience within HE and those who participated in the focus group meeting. The diversity of the participants' backgrounds, and experience of teaching in HE, as shown in Appendix three, identifies how the purposive sampling approach resulted in recruiting participants who matched Smith et al.'s (2009) and Yardley's (2000) criteria for selecting participants, and Creswell (2007) and Morse (2000) suggestions for the number of participants need to complete phenomenological studies.

It is also important to acknowledge how some of the participants identified as part of the purposive sample to contribute to the study were, on occasion, work colleagues of the researcher due to initially being approached to participate in the study. McEvoy (2002) states that this issue may hinder a study due to established relationships between the researcher and colleagues. However, given the study's focus on allowing HE academics to narrate the

stories of their teaching experiences, Fielding's (1994) view that participants may respond more favourably to researchers who are like themselves reinforced the appropriateness of this decision. Furthermore, using a second interviewer, who was not previously known to the participants to interview six out of the 11 participants, helped counter the criticisms McEvoy (2002) put forward regarding interviewing participants known to the researchers. Section 3.3.2 further details the conduct and relationship between the two interviewers. The share of interviews completed by each of the two interviewers is detailed in Appendix three.

### 3.2.2 The research interviews

In considering the approach to completing the participant interviews, Whiting's (2008) fourstep process of completing research interviews comprising a rapport-building phase, an exploration phase, participative/cooperative and concluding phases helped provide a broad framework. For example, in this study, a rapport-building phase informed the initial introduction to the interviews and the completion of a Participant Consent Form (see Appendix two) by establishing the scope and boundaries of the interviews and allaying participant fears. The main thrust of the interviews focused on the 'exploration' and 'participative' phases via the use of the Clean Language questions to identify participants' metaphors and explore the significance the metaphors had in conceptualising their teaching practice experiences. However, it is important to note that the 'cooperative' phase of the interview would be incongruent with the use of the Clean Language questioning model, the focus of which is on keeping conversation 'clean' by not 'dirtying' the discussions by allowing interviewers' values, beliefs and interpretations of the issues to impact on the questions asked during the interviews. Hence, it is important to stress that Whiting's (2008) four-step process only informed a framework and did not provide an agenda for completing the research interviews. Particular attention was also given to concluding the interview to ensure that participants were emotionally comfortable with the issues discussed and the interview's conduct before its end.

The 'Clean Language' questions were used to probe and explore the metaphors participants referred to in reflecting on their practice and determining the narratives underpinning their practice's conceptualisation (see section 2.4 for a more detailed discussion of the Clean Language model). Three key factors led to employing the Clean Language questions to interview and probe participants. Firstly, Bullough et al. (1992), Pajares (1992) and Mahlios et al.'s (2010) notion of how teachers hold beliefs that influence not only how they teach but also how they interpret their experiences of teaching stress the importance of exploring metaphors representative of the schemas underpinning teachers' thoughts and behaviours

(Cook-Sather, 2003). Therefore, the Clean Language questions' specific emphasis and interest in exploring metaphors was especially relevant for this study's research method and interviewing tool. Secondly, the potential offered by the Clean Language questions focuses on allowing participants to freely discuss their teaching experiences without being confined by the structures of the traditional cyclical models of reflection promoted by Kolb (1984), Gibbs (1988), etc. helped to support the use of the Clean Language questions to complete the research interviews. Furthermore, the use of the Clean Language questions, in allowing researchers to probe participants' metaphors, aided in reinforcing the use of open interviews as a key research method given the concentration and richness of data that could be collected via the use of interpretive qualitative interviews (Leedy & Ormrod, 2015).

As a questioning method, using the Clean Language questions aided the 'bracketing' process referred to within phenomenological studies by acknowledging possible inferences and contamination of the interviewer's subtle wording of questions on participants' responses (Grbich, 2012). The 'bracketing' process is considered key to creating the meaning of everyday phenomena and should be considered when interviewing participants within phenomenological studies. 'Bracketing', the process whereby a conscious attempt is made to avoid subjective interpretation of the phenomena by putting aside values and beliefs that could intrude in understanding the true nuances and structures of a phenomenon, was aided via the use of the Clean Language questions (Lawley & Tompkins, 2000; Sullivan & Rees, 2008). By aiming to minimise the impact of the interviewer's role, which is solely to listen to participants' responses without interpreting or influencing the discussion. This allows participants to relay various metaphors according to their perception and conceptualisation of a phenomenon. It also strove to counteract Loftus' (2005) concerns regarding the overt and covert influences interviewers' questions can have on a discussion.

Two researchers completed the interviews in a bid to manage time and resources in the completion of the research interviews. The synchronicity between the two interviewers was assured via initial discussions regarding Whiting's (2008) interview framework and regular debriefings and discussion of the research interviews. Furthermore, using the Clean Language model as a questioning technique and interviewing tool required the researchers to be skilled in using the model and its focus on what is present and presented (Lawley & Tompkins, 2000; Sullivan & Rees, 2008). Both interviewers developed these skills by

completing a minimum of Modules 1 and 2 of the Clean Language certification courses, thus assuring a level of competence in using the Clean Language questions.

Regarding managing the domestic arrangements of completing the research interviews, participants were given a choice and flexibility regarding the interviews' location and timing. Video conferencing tools (e.g., Skype, Zoom) were used when necessary to complete online interviews. Using a Dictaphone helped free the interviewers from making written notes and provided a record of each interview for transcription purposes. The potential effect an audio-recorder could have on participants' contributions to the interview discussions was acknowledged, with time being taken initially to confirm participants' willingness to be audio-recorded formally with the completion of the Participant Consent Forms (see Appendix two), and again informally before the interviews.

Bryman and Bell (2015) identify the key criticisms of interpretive qualitative interviews with regards to the reliance and impact of the questions asked during the research interviews on the findings, and the impact researcher biases and interpretation may have on the conclusions drawn from the analysis of the interview transcripts. Such criticisms were in part countered by using the Clean Language questions' specific focus of asking minimal questions in order not to 'dirty' the conversations by interviewer inferences and interpretations of the participants' responses (see section 2.4 for further details on the application and use of the Clean Language questions). Researcher bias was further mitigated in the analysis of the interview transcripts whereby specific pre-defined codes consisting of Kövecses' (2010) source and target domain codes (see Appendix seven) were initially used to identify the metaphors mentioned by participants while reflecting on aspects of their teaching experiences (see section 3.3 for further details of the data analysis process employed to review the interview transcripts).

## 3.2.3 Questionnaire evaluation of the interview and Clean Language questions

To identify the extent the Clean Language questions allowed participants to explore metaphors of their teaching experiences (research question two), participants were asked to complete a short questionnaire to reflect on their experiences of completing the research interviews. The use of a questionnaire (see Appendix 11) to complete an initial analysis of participants' reflections of the research interviews stems from the questionnaire's ability to remotely collect participant views about the use of Clean Language questions within the research interview in a time-efficient way (Brace, 2013). However, limitations exist in using such questionnaires since responses depend on a participant's interpretation of the questions asked. This was countered by completing a follow-up focus group (attended by seven participants) to elaborate and air their views on the conduct of the research interview, the use of the Clean Language questions and the metaphors identified as symbolic of their narratives of their teaching practice.

Lietz (2010) argues that consideration needs to be given to the composition of a questionnaire and the wording of questions within a questionnaire. For this study, the ease of completing the structured questionnaire was aided by using questions consisting of a mix of five-point rating scales and open-ended comment boxes, and ensuring the wording of the questions was in keeping with participants' understanding of the issues discussed (see Appendix 11). The use of online questionnaire distribution and data collection software helped produce an easyto-complete questionnaire that allowed participants to respond via point-and-click functions.

Regarding the composition of the questionnaire, an account was taken of Lietz's (2010) assertion to ask general before specific questions to help enhance questionnaire comprehension. This comprised generic thoughts and feelings about the research interviews being probed before more detailed questions and the omission of personal classification questions (age, gender, etc.) to ensure anonymity and/or the non-completion of the questionnaire due to fear of disclosure. Issues regarding the length of the questions, wording and order have been further cited by Lietz (2010) as matters that could negatively impact the significance and relevance of the data collection via questionnaire to be kept short and within the range of 16-20 words a question (Oppenheim, 1992). Appendix 11 shows that ten out of the 11 questions fulfilled the sentence length recommendations put forward by Oppenheim (1992), with only the final question requesting participants' further comments and recommendations about broader issues being slightly longer.

Many questionnaire participants are not likely to want to expend too much energy in completing a questionnaire. Thus, care was taken to ensure all questions were carefully worded and struck a balance between using a range of open and closed questions. Additionally, Foddy (1993), cited in Lietz (2010), coined the term 'question threat' further to emphasise how levels of complexity, poor wording, use of hypothetical questions, negatively worded questions and the need to qualify an answer as factors contributing to creating a
climate of fear that could negatively impact on the completion of the questionnaire. 'Question threats' were considered via a pilot of the questionnaire with willing colleagues who were not part of the final sample, which resulted in minor amendments being made to the wording of some questions, including the use of a five-point rating scale which allowed respondents to give a 'somewhat' answer to questions where they might feel ambivalent in response to the questions asked.

While helpful in allowing the rapid collection of quantifiable data within a condensed time frame, the use of a questionnaire poses challenges in limiting the level and depth of data that the completion of a short questionnaire could collect. Hence, to further corroborate the findings of the questionnaire, a focus group of participants was convened to probe the authenticity of the responses given and assess the use of the Clean Language questions as a research interviewing tool in more depth.

#### 3.2.4 Use of focus group

The focus group, attended by seven out of the 11 participants (see Appendix three) who were available to join the meeting, was primarily guided by the research questions focused on gauging the value of metaphors in contextualising narratives of their teaching practices and evaluating the use of the Clean Language questioning model. It aimed not to infer, generalise or seek explanations, but to allow participants to expand on the questionnaire's initial findings and, in doing so, to gain a greater understanding of participants' perceptions of completing the research interviews via informal group discussions (Parker & Tritter, 2006). Participants unable to attend the focus group meeting were given opportunities to view and comment on the preliminary analysis of the findings.

A key component of fruitful focus group discussions rests on the researcher's ability to facilitate discussions amongst the participants by managing the group dynamics, ensuring all participants fair contribution and accepting all views expressed (Freeman, 2006; Xerri, 2018). Therefore, care was taken to ensure the focus group discussion remained focused, was not leading and flowed from asking general to specific questions about the emotions and thoughts attached to completing the research interviews. Given focus groups' potential to generate unexpected disclosures by participants and resulting confidentiality issues, participants were reminded of the BERA (2018) Code of Ethics (see section 3.4 for further details), and participants' consent was sought before commencing the focus group discussions.

Freeman (2006) asserts the need to carefully consider the size and selection of participants chosen to contribute to a focus group. While variations exist regarding focus groups' size – 6-12 participants (Freeman, 2006) or 3-8 participants (Gill et al., 2008) – there is consensus that a focus group needs to be populated by participants who have experienced a common phenomenon and are willing to contribute and share their experiences via group discussion (Parker & Tritter, 2006). The use of a pre-existing group, consisting of a total of 11 participants, who had initially been interviewed as part of this study, ensured the focus group comprised of participants with shared knowledge and familiarity of being interviewed as part of this study, and they were willing to reflect on and contribute the experiences within the confines of recommended group sizes for informal focus group discussions (Gill et al., 2008).

#### 3.2.5 Pilot surveys

In keeping with Creswell's (2007) suggestion of the need for the research interviews to have inherent validity, two pilot interviews were completed to explore issues related to using the Clean Language questions, and the value and validity of the data collected before the actual research interviews. This consisted of two interviews with two colleagues who were happy to volunteer as respondents to determine the value of using the Clean Language questions within a research context and a pilot survey to determine the potential of exploring academics' metaphors of their teaching experiences.

The two pilot interviews with respondents lasted 30 approx. minutes on average. They were audio-recorded, transcribed by a professional transcription service company (see Appendix four – sample interview transcript) and analysed to determine the application of the Clean Language questions within the research interview context and potential adaptations that needed to be made to use the Clean Language questions before the actual research interview with participants.

Analysis of the pilot interview consisted of determining the depth and level of use of Clean Language questions and attempts to identify metaphors referred to by respondents to reflect on their teaching experiences. A debriefing further followed this analysis of the interview transcripts with respondents to determine their reflections on being interviewed as part of the pilot study.

The debrief discussions with the respondents involved discerned that, while neither of the respondents was familiar with or had previous knowledge of the Clean Language model or

questions associated with the model, it was encouraging to hear respondents state that they "were surprised by the depth of the questions asked" and how they felt "able to develop the conversations and metaphors in their own way". Respondents further confirmed how using the Clean Language questions (see Appendix one) aided in steering the interviews towards identifying and exploring metaphors during both the pilot interviews. Hence, any initial anxieties regarding using Clean Language questions within a research context were allayed before the commencement of the complete study.

However, both the respondents reported they found the questions asked at times to be "*mechanical and repetitive*" and "*disrupt the flow of the interview*". These statements and interviewer observations suggest how the use of the 'Clean Language syntax' (i.e., the mode and manner in which the Clean Language questions are asked) seemed incongruent and, at times, hindered the discussions' flow. This issue of using the Clean Language syntax was further explored via online discussions within the LinkedIn 'Clean Language and Symbolic Research Group', drawing contributions from principal authors from within the Clean Language community, including James Lawley. The combination of the feedback from the pilot interviews and summary of the online discussions with the Clean Language community concluded that using the Clean Language syntax is ideally suited within a therapeutic context but does not necessarily lend itself to use within coaching or a research methodology context. Therefore, as part of the actual research interviews to move the interview towards a more conversational style of interviewing and to enable the researcher to ask questions to verify the value and importance of the metaphors referred to by the participants.

In addition to the pilot interviews, to determine academics' ability to refer to metaphors to describe and conceptualise aspects of their teaching experiences, a small-scale email survey was completed, whereby 20 academics were asked to respond to the questions:

When I'm teaching at my best, it is like [respondent response]?

On an off day, my teaching is like [respondent response]?

These questions were compiled with the research questions in mind to establish the potential relevance and significance of metaphors to convey deeper insights into academics' conceptualisation of their teaching experiences. The survey resulted in 13 academics responding to the email request, (see Appendix five for sample responses) whereby the

answers to the questions were analysed to identify themes inherent within the metaphors used by academics to describe "When they are teaching at their best" and how they perceive their teaching to be like "on an off day".

An analysis of the responses identified metaphors akin to being a 'rock star', 'actor', 'performer' and 'host of a party' used to describe academics' perceptions of when they are teaching at their best and 'performance', 'engaged', 'connection', 'enthuse' and 'inspire' as a way of describing their actual teaching practice. Similarly, with regards to the question "On an off day, my teaching is like?", 'walking through treacle', 'hit and miss' and 'uphill hike' were identified as metaphors to describe the 'disjointed', 'frustration', 'pulling' and 'effort all the way' emotions attached to aspects of their teaching practice. Hence, confirming the ease with which academics could draw on metaphors symbolic of their teaching practice's conceptualisation.

The findings of this small-scale pilot survey were reported as part of a paper delivered at the Association of Neuro-linguistic Programming Research Conference titled *Metaphors We Teach By: An exploration into how HE lecturers reflect on and conceptualise their experiences of teaching via metaphors* (Karolia & Burton, 2012). The presentation of the paper was well attended and received, indicating a good level of interest in the paper and the research study (see Appendix six for a copy of the PowerPoint presentation slides). The range of questions asked by the delegates attending the presentation, including noted Clean Language author Wendy Sullivan (Sullivan & Rees, 2008), and delivery of subsequent poster presentation during the 2014 International Clean Language Conference (Karolia, 2014), further reinforced the significance of exploring the use of the Clean Language questions within a research context.

In summary, the two-pilot survey completed before the actual research interviews suggested that Clean Language questions could be used by researchers interviewing participants to identify and explore the metaphors that they use to describe their teaching experiences within a research context. However, the effectiveness of this process necessitated a modification of the standard Clean Language syntax more commonly used in therapy.

MH Karolia

#### 3.3 Data Analysis

Elo and Kyngäs' (2008) stages of reviewing transcripts were employed to analyse the research interviews as part of this study, given how their procedural approach to analysing qualitative data fitted the research methods adopted for the study. The stages, consisting of 'preparation' (selecting/deciding on a unit of analysis), 'organising' (open coding and analysis of categories) and 'reporting' (write-up and dissemination of findings), aided in establishing the link between the need for a comprehensive approach to select codes representative of metaphors to commence the initial content analysis of the interview transcripts. Elo and Kyngäs' (2008) stages of reviewing interview transcripts further enabled computations of data and categorisation of themes attributing meaning and informing the issues researched. This is a process Charmaz (2014) metaphorically describes as the bones of the analysis (via coding of data) and helps to assemble the working skeleton (the theme), generating an understanding of the phenomena examined. In considering the working skeleton and bones of analysis for this study, a combination of content and thematic approaches to analysing qualitative data was used to examine the transcripts of the research interviews via the use of NVivo data analysis software (Squire et al., 2014). Vaismoradi et al. (2013), in their discussion of the differences between 'content' and 'thematic' methods of analysis, describe content analysis as a "systematic coding and categorising approach used for exploring copious amounts of textual information unobtrusively to determine trends and patterns of words used, their frequency, their relationships and the structures and discourses of communication" (p. 400). In contrast, 'thematic analysis' is defined as "a method of identifying, analysing and reporting patterns (themes) within data" (Braun & Clarke, 2006, p. 79). Hence, within the context of this study, 'content analysis' aided in facilitating the identification of metaphors uttered by academics to describe aspects of their teaching practice and 'thematic analysis' of the metaphor aiding in the analysis of the narratives representative of metaphors appearing out of the initial content analysis of the interview transcripts. However, it is important to acknowledge considerable overlap in employing 'content' and 'thematic' analysis methods.

The decision to employ a joint 'content' and 'thematic' approach to the analysis of the interview transcripts was further reinforced by the critical distinction as to whether the underpinning aim of the research was inductive or deductive. Given the research questions' focus on determining the dominant metaphors academics employ to reflect on their teaching practice, and on evaluating the use of the Clean Language questions within a research context, the use of a deductive method of analysis was employed in keeping with the existence of pre-

76

determined codes to aid the initial analysis of the interview transcripts (see section 3.3.2 below for further details). Where there exists a lack of knowledge of the phenomena being analysed, the use of 'inductive' approaches of analysis, whereby themes emerge from a review and examination of the data, is apt and in keeping with research questions aimed at identifying the dominant metaphors in academics' description of their teaching practice (research question one), and the relationships between academics' use of metaphors and their narratives of teaching (research question two).

The analysis of the interview transcripts to determine the trends and patterns of phrases and metaphors referred to by participants was completed with the assistance of NVivo qualitative data analysis software. NVivo was specifically employed to complete the content analysis of the interview transcripts to identify metaphors referred to by participants, and thematic analysis of the conceptual metaphors emerging from the findings, to determine and give an insight into participants' narratives of their teaching practices via the codes further discussed below in section 3.3.1.

#### 3.3.1 Coding decisions

As part of Elo and Kyngäs' (2008) 'preparation' stage of analysis, consideration was given to selecting codes to analyse the interview transcripts. Coding, also referred to as 'indexing', 'categorising' or 'identifying themes' (Gibbs, 2008), within the context of qualitative research studies, helps to establish the 'critical link' between the captured data and the explanation of its meaning (Charmaz, 2006). However, care needed to be exercised in the selection of the codes as stressed by Saldaña's (2009) phrase: "if it moves, code it" (p. 13), to highlight the endless range of codes that could be referred to analyse data, thus emphasising how the identification of codes to analyse data is not an exact science given the limitless coding possibilities that exist. Furthermore, Sipe and Ghiso (2004) correctly note how "all coding is a judgement call" (p. 482), accentuating how care needs to be exercised in the choice of codes representative of the research questions; in preference to the researcher's perceptions or personal interests, which may function as a lens, and potentially filter and contaminate the interpretations and analysis of the data.

In adopting a content and thematic approach to data analysis for this study, key decisions regarding determining the codes for analysing the interview transcripts acknowledged Mason's (2018) assertion of the need to refer to theories in identifying pre-determined concept-

driven codes in preference to data-driven codes (Gibbs, 2008). More specifically, with regards to the first research question – "What are the dominant metaphors used in the way academics recall, reflect and describe their experiences of teaching?" – the codes referred to in analysing the interview transcripts consisted of Kövecses' (2010) list of metaphor source and target domains. Kövecses' (2010) list was referred to in the first instance as codes to analyse the interview transcripts by academics reflecting on their teaching practice. This is primarily due to most of the 'source' and 'target' variables being uttered by participants linking to Kövecses' (2010) source and target domain list.

The definitions of codes consisting of locating the 'source' and 'target' domains are:

**Source domain:** referring to the expression of a metaphor via variables (predominantly concrete/physical, tangible concept) used to describe a concept consisting of either a word or a phrase – codes denoting possible source domains initially comprised of those identified by Kövecses (2010).

**Target domain:** refers to the inference of a metaphor, where the 'concept' (idea) conveyed by the source domain corresponds to the target domains.

Kövecses' (2010) work on a comprehensive list of source and target metaphor domain categories (listed within Appendix seven) helped to provide a base starting point whereby the interview transcripts were filtered to identify metaphors and the vehicles used to communicate the metaphors (see section 2.2 for further details). Given the unlimited range of codes available to analyse interview transcripts, care was taken to consider how the decision to employ one set of codes over another represents a judgement call on the researcher's part, impacting the findings and conclusions drawn from the analysis of the interview transcripts (Sipe & Ghiso, 2004). Therefore, additional source domain coding categories of colours, communications, faith/religion, food, good/bad, materials/equipment, person/people, place/location, senses, shapes, sizes/measures, text/stories, tools/machines and transport metaphors were coded as part of the analysis of interview transcripts. The identification and addition of these coding categories demonstrate the breadth and depth of metaphors referred to by participants to convey experiences and limitations of the list of source domains originally proposed by Kövecses (2010). The range of 13 target domain categories initially identified by Kövecses (2010) proved restrictive in their application in not acknowledging the varied levels of interaction and engagement between experiences and the metaphors used to convey the experiences, thus not allowing sufficient identification of the patterns and themes from the findings. Consequently, possible additional codes to supplement Kövecses' (2010) list of target domains was explored by reference to Lakoff et al.'s (1991) Master Metaphor List and NLP logical levels of learning (O'Connor & Seymour, 1994; 2002). These lists were referred to as a basis of the mapping process between the source and target domains given the limited range of target domains originally identified by Kövecses (2010). A decision was made to broadly refer to the NLP levels of learning (O'Connor & Seymour, 1994; 2002) as additional coding categories in preference to Lakoff et al.'s (1991) Master Metaphor List, which only groups conceptual metaphors within categories consisting of events, structures, mental events and emotions. The NLP logical levels of learning consist of analysing individuals' experiences via additional categories consisting of:

- 1. Identity
- 2. Values and beliefs
- 3. Capability
- 4. Behaviour

In addition to the mapping of source to target domains offered by NLP logical levels of learning, the decision to refer to them was further reinforced by the simplistic, holistic and identifiable categories of the levels to understand the various interacting elements of experiences, and referred to as thematic codes to allow for more thorough mapping of source to target domains. A further 'emotions' category was added to the list of target domains by identifying how some metaphors expressed by the participants referred to feelings and emotions associated with an experience, thus adding another layer of coding of the interview transcripts to gain deeper insights into the links between the metaphors' source and target domains. Once the conceptual metaphors were identified via the analysis of the interview transcription following Kövecses' (2010) source and target domains, the final stage of analysis consisted of determining the narratives representative of participants' teaching practice.

As with other methods of inquiry, narrative approaches to research need to be fit for purpose and justified as an appropriate methodology and method to investigate phenomena via a consistent and uniformed approach to analysis (Riessman, 1993; Czarniawaska, 2004; Bold, 2012). Bold (2012) suggests these elements should be uppermost in a researcher's mind before starting research to ensure that narrative analysis gives an authentic and genuine account of the lived experience permeating across the stories internalised by participants (Squire et al., 2014). A range of possibilities exists to identify the narratives permeating through stories. For example, narratives could be analysed to identify how individuals think and feel (Bruner, 1984, cited in Moen, 2006); the way narratives are shaped and coconstructed for and by different audiences (Georgakopoulou, 2007); or how narratives are situated within the sphere of performance, context or social identities of meaning (Riessman, 2008; Salmon & Riessman, 2013). As discussed within section 2.3, McAdams and McLean's (2013) Narratives Identity theory was adopted as the primary theoretical lens to analyse the data to determine academics' narratives of their teaching practice against the seven Narrative Constructs helping to establish a coding strategy for the interview transcripts as listed in Appendix eight.

In a bid to gain further and deeper insights and analyse the narratives symbolic of conceptual metaphors expressed by participants as part of this study, and, as discussed within section 2.3 of the literature review, in addition to McAdams and McLean's (2013) Narrative Constructs, the premise underpinning conceptual metaphors was analysed via reference to Tompkins and Lawley's (2006) PRO model and Schwartz's (2012) theory of Universal Values as codes to determine the primary values and beliefs supporting the metaphor. See section 2.3 for further details of the variables attached to Tompkins and Lawley's (2006) PRO model, McAdams and McLean's (2013) Narrative Constructs and Schwartz's (2012) theory of Universal Values used as coding categories and also listed in Appendix eight. The use of the above theories of analysing and interpreting narratives would be congruent with Riessman's (2008) notion of the need for research into narratives to give an insight into the stories people tell of their experiences, or, within the context of this study, academics tell of their teaching practice, as opposed to the linguistic analysis of a story, aspects of personal identity or agents for social change. Furthermore, this study takes on-board Andrews et al.'s (2013) suggestion to move away from the tendency to identify cause and effect patterns in the analysis of narrative by referring to the range of theoretical lenses discussed above to determine academics' use of conceptual metaphors within the stories of their teaching practice.

Regarding the third research question to identify "To what extent do the Clean Language questions allow participants to reflect on the metaphors representative of their teaching experiences?", Sullivan and Rees' (2008) list of Clean Language questions (see Appendix one) was used as codes to identify the interviewer's 'Clean' questions within the interview transcripts. The use of the core Clean Language questions referred to by Sullivan and Rees (2008) was used to complete the interviews and code the interview transcript given how questions listed by Sullivan and Rees (2008) were most commonly recognised as 'Clean'

questions within the Clean Language community. Furthermore, Lawley's (2014) *Protocol for Validating 'Cleanness' of an Interview* was applied to the research interviews' Clean Language questions as previously discussed within section 2.4.

The decision to use Lawley's (2014) criteria came from recognising how the criteria allowed for evaluating the Clean Language model in different contexts and away from the pure form of Clean Language used within a therapeutic setting. The use of these criteria also helped to keep some uniformity in the evaluation of the Clean Language questions and further add to the growing number of empirical research studies into the use of the Clean Language questions against measures referred to in other similar studies, albeit within different contexts (see section 2.4 for further details of the Clean Language model and questions).

The preliminary analysis of the initial interviews identified the use of additional questions asked by the interviewer not matching the protocol definitions supplied by Lawley (2014), but deemed significant in contributing to the interview in allowing participants to discuss and explore their experiences more comprehensively. Therefore, a coding category named 'Reflective response' was added and evaluated as part of the interview analysis by identifying summary and paraphrasing responses by interviewers. Furthermore, on occasion, the interviewer asked questions for clarification to prompt and further explore a response in more depth. Therefore, the addition of code to identify a 'Probing question' aided in more comprehensively evaluating and measuring the complete set of questions asked by the interviewer.

The process of completing the coding analysis of interview transcripts is summarised in figure 2 below and discussed in more depth in the following paragraphs.



Figure 2 – Interview Transcripts Coding Process Summary

Within this study, the key consideration in analysing the interview transcripts was the need to isolate the metaphors symbolic of participants' narratives and the first step completed as part of the interview transcripts' process. The metaphors referred to by academics in reflecting on aspects of their teaching practice form the bedrock of this study and align with Mahlios et al.'s (2010) suggestion of the need to research metaphors representative of teachers' values, beliefs and prior experiences in order to identify how academics conceptualise their experiences of their teaching practice. The focus on discerning the dominant metaphors within the interview transcripts further aided in determining the 'appreciative systems' referred to by Schön (1987) to describe how teachers frame their experiences and Moen's (2006) assertion of how stories provide a lens, which could be explored, and function as a narrative frame influencing the way individuals think and behave (Squire et al., 2014). The metaphors within the interview transcripts were coded, via the use of Nvivo software, against a content analysis of Kövecses' (2010) list of metaphor sources and target domains. Incorporating the additional source and target domains in the analysis of the interview transcripts further aided the

identification of conceptual metaphors (Lakoff & Johnson, 2003) expressed by the participants (see sections 2.2 and 2.5.3 for further details).

The next stage in the analysis of the metaphors consisted of a thematic analysis of the metaphors to determine the key conceptual metaphors emerging from metaphors referred to by participants within the interview transcript. The thematic analysis to identify the conceptual metaphors consisted of searching for patterns and categories appearing out of the initial content-coding analysis of the interview transcripts. Patterns – the search for repetitive, regular or consistent occurrences of action/data that appear "more than twice" (Saldaña, 2013, p. 5) – were viewed via a lens linked to the work of Lakoff and Johnson (1980; 2003) to determine the conceptual metaphors emerging out of the dominant metaphors expressed by participants in reflecting on aspects of their teaching practice.

To solidify and explore the narratives representative of participants' use of conceptual metaphors, as part of the second research question, the next step in the analysis process consisted of a deeper thematic analysis of conceptual metaphors. This further level of thematic analysis involved scrutiny of the conceptual metaphors against Tompkins and Lawley's (2006) PRO model, McAdams and McLean's (2013) Narrative Constructs and Schwartz's (2012) theory of Universal Values (see sections 2.3 and 2.5.3 for further details).

Finally, the third step of the analysis process comprised an analysis of the questions asked by the researchers to determine the depth and level of the use of the Clean Language questions within research interviews (research question three). This process of evaluating the use of the Clean Language questions consisted of coding every question asked by the research interviewers against Sullivan and Rees (2008) list of Clean Language questions. Questions not considered as a Clean Language question were coded and categorised as either reflective prompts or probing questions (see Appendix one and nine). To benchmark the depth, each Clean Language questions used during the interviews were coded against Lawley's (2014) Protocol for Evaluating the 'Cleanness' of an Interview. The listed metaphor and Clean Language coding categories (see Appendix four and five) helped complete the initial content analysis of the interview transcripts and were followed by thematic analysis of the codes to determine the narratives representative of academics' teaching practice. These are reported within the findings and conclusion chapter of the thesis, and examples of the interview transcripts' coding by both the researchers can be found in Appendix 10.

However, the conduct of the research identified limitations in the use of the original codes selected to analyse the interview transcripts, not giving a thorough enough insight into the broader narratives needed to address the research questions. Therefore, using the NLP logical levels of learning helped provide a broad framework against which a range of codes was used to analyse the interview transcripts to give deeper insights into research questions. More specifically, against the NLP learning levels, the research interviews were additionally coded via the theoretical framework listed below:

- 1. Capability and behaviour Lawley and Tompkins' (2006) PRO framework categories
- 2. Narrative Constructs McAdams' (2013; 2016) Narrative Identity theory
- 3. Values and beliefs Schwartz's (2012) Universal Values list

The use of the above codes allowed for a more defined and precise analysis of the data and identifying participants' narrative voices (Squire et al., 2014) and reported in keeping with the research questions in the chapters that follow.

#### 3.3.2 Research validity

Sparkes (1992) notes that a proposition is true if it consistently corresponds with other propositions and reinforces findings from research completed via interpretive approaches. However, addressing the concerns associated with phenomenological research studies necessitates consideration of the validity of the research against the diversity existing in determining the criteria associated with measuring research validity for qualitative studies. To ensure the findings of research studies are meaningful, accurate and credible, Leedy and Ormrod (2015) stress the need to consider the validity of the chosen research methods. With regards to this study, phenomenological research focuses on analysing text sourced from interviews with participants. As such, the truth underpinning reflected account of stories cannot be determined or generalised via definitions linked to validity, reliability and replications referred to within positivist approaches to research. As an alternative, Moreno (2002) asserts the need to develop a deeper understanding of the storied evidence, not to identify what happened, but to determine the narrative attached to a phenomenon experienced, as interpreted by participants. Hence, and as acknowledged by the study, it is impossible to make generalisations or broader inferences from the findings and conclusions drawn from the research, other than implying how the findings may inform relevant discussions.

Reflections and stories of personal experiences do not happen in isolation but in proximity to the context and audience expectations. They are further influenced by how stories are generalised and deleted due to limits of memory, and distorted via levels of creativity and the language used to convey the story. Furthermore, by extension, narratives are construed as being 'fiction' (Moen, 2006) in how they are recalled and narrated, and compounded by inferences and censured self-disclosure by the narrator who may only present aspects of themselves and their story deemed to be in keeping with the audience's expectations. Hence, concepts linked to validity, reliability and replications of research within narrative studies may appear problematic in relation to the standard definition of these terms within research studies. In countering these inferences, Bold (2012) asserts that within narrative research, reliability can be found in exploring how the data represents and re-creates the phenomena experienced by persons, which was borne in mind to stress the validity of the stories told by participants of their teaching experiences.

McDonald et al. (2019), in their discussions of the need to consider inter-rater reliability within qualitative studies cite how reviewing and confirming observations and findings with participants and others involved in the research process, can assist in determining inter-rater reliability of a research project. Therefore, within this study, the use of two researchers assisted in determining the relevance and significance of the research interviews via regular meetings between the two researchers to reflect on their experiences of completing interviews, the use of the Clean Language questions within the interview, initial observations of findings emerging from the interviews, and the metaphors referred to by participants within the research interviews. This process of regularly conferring and reflecting on experiences of completing the research interviews. Furthermore, McDonald et al.'s (2019) recommendation to confirm the findings with participants to raise the inter-rater reliability of the qualitative research project was implemented by sharing initial findings of the analysis of the interview transcripts with participants individually and further corroborated via the focus group discussions.

The subjectivity involved in the interpretation of narrative data has further been cited by Gudmunsdottir (2001) and Moen (2006) as issues linked to the analysis of narrative, which illustrates the need for narrative researchers to reflect on, acknowledge and validate the claims made, and the instability of the truths established via the analysis of interview transcripts. Specifically, regarding the impact of personal biases and judgement on interpretation and

focus of texts by researchers in accurately mirroring the substance and essence of the meaning attached to a phenomenon by participants. Initially, these threats to validity were addressed by reviewing the conclusions drawn from the analysis of the data via a focus group discussion (as discussed within chapters 4-6) with participants following the analysis of the interview transcripts as suggested by Polkinghorne (2007) and Moreno (2002). It is further acknowledged how the process of coding interview transcripts is subjective regarding linking source data (participant utterances within interview transcripts) to the coding analysis of the data. Therefore, it is important to recognise the limits associated with generalising to broader contexts the claims made from analysing the data and these caveats are further acknowledged within the thesis.

#### 3.4 Ethics

Ethics in research, as in most aspects of life, pertain to doing good and avoiding harm, and within the world of research aim to balance the competing demands of studying a phenomenon against protecting the safety, privacy and rights of participants (Orb et al., 2001).

The BERA Code of Ethics (2018) was reviewed to ensure all stages of the research process and participants' safety, rights and privacy were considered before and during the completion of the study. The BERA (2018) guidelines require that "educational research should be conducted within an ethic" (p. 5) of safety for participants contributing to the survey by ensuring they know their rights and responsibilities. Hence, voluntary informed consent, referring to a willing agreement to participate in a survey, was assured by the completion of a Participant Consent Form (see Appendix two) in keeping with the BERA (2018) guidelines regarding minimising harm, avoiding undue intrusion, obtaining informed consent, and showing courtesy and respect to the research participants' willingness to contribute to the study. Furthermore, the Participant Consent Form also informed participants of how they have a choice in agreeing to be involved in the study, including having the right to withdraw at any stage of the research process given the sensitivity of the personal 'stories' participants may recount and reflect on during the research interviews.

BERA (2018) guidelines regarding privacy and data storage were also adhered to in order to ensure participants' rights to confidentiality and anonymity. All data linked with completing the research and analysis of data ensured participants were only identified via pseudonyms with

typed records and referred to directly or indirectly at any stage of the research process. Further care was taken to ensure the secure storage of data aligned with BERA (2018) guidelines via the use of password-protected access to all USB and hard drives used to store confidential participant data.

Each of the above considerations was conveyed to the participants via Mertens' (2020) recommendation to brief research participants before a survey, especially as the study sets out to explore the participants' thoughts and emotions attached to their reflections of their teaching practice, and analyse and report them in the public domain (Brinkmann & Kvale, 2005). The briefing consisted of talking participants through the Participant Consent Form and detailing how their rights and privacy will be maintained before confirming their agreement to participate in the study by signing the form.

The BERA (2018) recommendations to afford participant anonymity and privacy were maintained via the use of aliases in reporting the findings and giving participants open access to the audio-recording and transcription of the interviews they were able to request their withdrawal at any stage of the research process.

#### **Chapter Summary**

In keeping with this study's focus on exploring metaphors to identify the dominant narratives representative of academics' teaching practice, relativist ontology and interpretive epistemology paradigms underpin this study's research methodology and methods. By adopting a relativist ontology and interpretive epistemology, this study adopted an inductive methodology and phenomenological research approach in keeping with research questions intended to explore how the metaphors referred to by HE academics, when they reflect on their experiences, inform their narratives of teaching. The use of the phenomenological research approaches and, more specifically, the combined classical and hermeneutical phenomenological approaches, were employed as being the most appropriate in the context of the study's research questions, which focus on identifying the metaphors uttered by academics to determine 'what' and 'how' academics 'experienced' their teaching practice (Moustakas, 1994), with the researcher 'bracketing' themselves out of the study via the Clean Language model's avoidance of 'dirtying' conversations via inferences of interviewer judgements and assumptions.

In keeping with the use of phenomenological research methodology, the research methods used within this study consisted of ideographic qualitative research methods, namely interviews and subsequent distribution of a questionnaire and focus group discussion aligning with the research questions as listed below:

- What are the dominant metaphors used in the way academics recall, reflect and describe their experiences of teaching? – data was captured via the use of open interviews and analysis of interview transcripts
- To what extent do the Clean Language questions allow participants to reflect on the metaphors representative of their teaching experiences?
  – evaluated via the distribution of a questionnaire after completing research interviews and focus group discussion with participants
- What are the relationships between academics' use of metaphors and their narratives of teaching? – determined by analysis of findings and further corroborated via focus group discussions

More specifically, a sample of 12 participants was chosen for the survey, although only 11 eventually participated, in keeping with Robinson's (2014) suggestions to consider sampling universe, size, source and strategies for choosing participants.

The research methods consisted of open interviews to complete the study, aligning with the phenomenological research methodology, the research questions and the use of the Clean Language questions. A broad-based approach to structure and domestic arrangements of the interview was considered by following Whiting's (2008) approach to completing the research interviews. The third research question aimed at evaluating the use of the Clean Language questions and examining the conceptual metaphors emergent from the interview findings were addressed via a post-interview questionnaire and focus group discussions. Each method links to the appropriate research question. Analysis of the interview transcripts was completed by a mix of content and thematic methods of analysis: content analysis to identify the conceptual metaphors referred to within the research interviews and level of Clean Language questions asked by the interview; and thematic analysis to determine the narratives inherent within the conceptual metaphors identified.

The content analysis of the interviews to identify the conceptual metaphors was completed by the analysis of participants' metaphors against Kövecses' (2010) list of source and target

domains. Content analysis of the Clean Language questions asked was compared to Sullivan and Rees' (2008) list of Clean Language questions and Lawley's (2014) *Protocol for Validating 'Cleanness' of an Interview.* Lastly, the narratives representative of academics' metaphors of the teaching practice were determined by reference to McAdams and McLean's (2013) theory of narrative identities in the first instance.

At all stages of the research process, ethical issues were borne in mind by adhering to the BERA (2018) Code of Ethics guidelines to avoid harm, undue intrusion, safety and anonymity of participants, and completing of the necessary Participant Consent Form to ensure voluntary participation was agreed to in partaking in the study.

## Chapter 4 – Analysis of Findings into Participants' Use of Metaphors to Reflect on Teaching Practice

The findings into the dominant metaphors emerging from the analysis of interview transcripts are reported in this chapter. These findings primarily stem from:

"Collecting examples of linguistic metaphors used to talk about the topic...generalising from them to the conceptual metaphors they exemplify and using the result to suggest understandings or thought patterns which construct or constrain people's beliefs or actions" (Cameron & Low, 1999, p. 88).

The above quotation illustrates the process followed in analysing the interview transcripts in keeping with the three research questions to identify the dominant metaphors used in the way HE lecturers recall, reflect on and describe their teaching experiences. The insights gained from this analysis are detailed below, and the account commences with reporting on the metaphors referred to by participants (i.e., the source domains) and the messages the metaphors intend to convey (i.e., target domains), the identification of the primary conceptual metaphors emerging from the analysis, and the relevance and significance of these conceptual metaphors in informing the study.

The chapter commences by noting the observations made from the content and thematic analysis of the metaphors (section 4.1) and continues by recounting the analysis of identifying the leading 'source' and 'target' domains coded from the interview transcripts (sections 4.2 and 4.3). Finally, section 4.4 comprises a discussion of the implications of the dominant conceptual metaphors expressed by participants.

A deeper thematic analysis of the relationship between academics' use of metaphors and their teaching narratives in relation to the second research question is discussed in chapter 5.

### 4.1 General Observations Emerging from the Analysis of Findings

An initial observation from the review of the interview transcripts identified the use of metaphors over similes in the way participants reflected on aspects of their teaching practice. Saban's (2006) distinction of how similes compare one thing to another in contrast to Lakoff & Johnson's (2003) definition of metaphors as describing "one thing in terms of another" (p. 5) was evident within the coding of the interview transcripts, specifically via phrases conveying

significance, correlations, parallels, similarities and likeness in the metaphors referred to by participants reflecting on their teaching practice. Some participants within the focus group explained how the use of metaphors over similes during the interviews was representative of the greater level of thinking and reflection into aspects of their teaching practice stemming from the extensive encouragement within teacher training courses to become 'reflective practitioners' (Schön, 1983). This was a phenomenon in which participants appeared to be well-versed and familiar with the need to reflect in order to enhance practice within education.

Probing the use of metaphors during the focus group discussions, some surprise was expressed by participants at their level of the use of metaphors via comments:

"didn't appreciate how often we talk in metaphors" [Participant 2] "I would need to read through the interview transcript again carefully, but my general impression was the issues discussed and metaphors explored were significant" [Participant 3]

These comments and subsequent focus group discussion mirror Geary's (2011) claim to the popularity of metaphors to convey phenomena and stories, though not to the extent of Geary's (2011) further claim of how individuals could utter up to six metaphors a minute. This is primarily due to the uncertainty of the context in which the claim is made, since Geary's study likely referred to everyday conversations, in contrast to research interviews and the use of Clean Language questions as a research method. However, participants within the focus group concurred with Tosey et al.'s (2013) review of the 'six metaphors a minute' claim by suggesting it depends on circumstances, since individuals are likely to resort to the use of metaphors to convey concepts that they are unable to express in other ways (Barry et al., 2009).

#### 4.2 Analysis of Source Domains

This section commences with an account of the analysis of the metaphors referred to by participants in relation to the list of Kövecses' (2010) source domains against which the interview transcripts were mapped and analysed. The discussion into the relevance and significance of the primary conceptual metaphors emerging from this analysis of source domains is further debated within section 4.4.

A total of 818 source domains were coded from the research interviews across 29 categories, primarily drawing on Kövecses' (2010) list of source domains and supplemented by some additional source domains not covered by Kövecses' (2010) original list shown within Appendix seven.

The analysis of the source domains (via NVivo qualitative analysis software) used to express a metaphor, interestingly identified a significant pattern whereby metaphors consisting of four dominant source domains were continuously referred to by participants while reflecting on aspects of their teaching practice. Source domains allied to metaphors referring to 'movement and directions', 'place and locations', 'senses' and 'nature and environment' (in order of popularity) together consisted of 52.2% of all source domains coded, with the remaining 25 categories combining to represent only 47.8% of source domains coded (see Appendix 12 for Therefore, these findings show how, despite the extensive range of further details). metaphorical source domains available to participants to reflect on their practice, a narrow bank of source domains was continually and consistently referred to by the participants interviewed. This suggests a pattern whereby participants continually conceptualised their teaching within the realms of orientational and geographical metaphors of movement, locations and self-awareness of their actions. Lakoff & Johnson's (2003) notion of conceptual metaphors, whereby an abstraction is represented via a concrete concept, was evident in matching source and target domains within the interview transcripts. However, the conceptions of container, entity and substance metaphors, referred to by Lakoff & Johnson (2003) as categories of metaphors, were not readily identifiable within mapping the source and target domains of the metaphors used in this study. This reinforced the use of Kövecses' (2010) broad range of source domains categories as relevant coding classifications since they extended beyond the container, entity and substance source domains, and proved more appropriate in coding the interview transcripts by identifying and specifying the metaphors referred to by participants within the research interviews.

A more thorough examination of the four popular metaphors referred to by participants identified source domains implying 'movement and directions' ranked highly, with 19.19% of all source domains quoted being associated with metaphors indicating movement. Examples included:

*"I'm compelled to keep going up the mountain"* [Participant 1] *"That is really what's driving me to reflect"* [Participant 2] "You just run with what you have used" [Participant 3] "I just keep it spinning, and it may not be going in the right direction, but I keep it spinning" [Participant 3]

Closely linked to the widespread use of 'movement and directions' source domains to express a metaphor were domains representative of 'place and locations' (15.53% of all metaphors quoted). Further investigation of the 'place and locations' source domains identified a distinct lack of specific or known locations in preference for abstract non-representational places or locations within the metaphors referred to by participants, for example:

"There are little places all the way along the path" [Participant 1] "I'm stuck in a rut" [Participant 2] "I don't actually think that I will get to that point" [Participant 3] "I've never quite got there" [Participant 6]

Allied to the use of 'movement and directions' and 'place and locations' metaphors were source domains representative of 'senses' (10.15% of all source domains coded) and 'nature and environment' (7.33% of all source domains); examples of which are listed in table 2 below:

'Senses' source domains	'Nature and Environment' source domains	
"When I'm in a rut, I can't see and can only see in a straight line" [Participant 2]	"At the moment, it's quite rocky" [Participant 1]	
"I did have a vision of sliding down the mountain" [Participant 1]	"Survey the whole landscape" [Participant 2]	
"I recognise the piece of music" [Participant 8]	"there's like this big mountain in front of me" [Participant 1]	
"Thinks they're out of tune, in other words, does anybody want some clarification" [Participant 8]	"Well, I suppose still waters throughout" [Participant 10]	
"Feel my way at the moment" [Participant 1]	"you're calm you think clearly, there's no fog" [Participant 10]	
"I think it's that sense of knowing that I'm getting confidence" [Participant 9]	"Like a ray of sunshine really" [Participant 9]	

Table 2 - Examples of Coded Sense and Nature/Environment Domains

In probing the relationships between 'movement and directions' and 'place and locations' metaphors with participants within the focus group, they unanimously identified and agreed on a clear link whereby 'place and locations' source domains were symbolic in representing the presence of outcomes, either in the form of personal aims or formal curriculum objectives. These phenomena were further expanded by discussions within the focus group whereby participants correlated how 'movement and directions' source domains and metaphors symbolised how they constantly strived towards enabling their students to meet set learning objectives and, to a lesser extent, their personal goals.

Secondary to the use of 'movement and directions' and 'place and locations', source domains linked to 'senses' and 'nature and environment', upon further analysis, were summarised as being linked to personal observations or actions, instead of other external factors, and in doing so demonstrate a degree of awareness of how locus of control extends only to themselves with regards to the decisions and actions they take within the classroom while teaching. Sharing this observation with participants within the focus group discussions revealed insights into the level of self-awareness and criticality that participants employed when reflecting on aspects of their work within the classroom while teaching. All the participants within the focus group concurred on how they continually aimed to improve their teaching practice and further linked this to the notion of reflective practice and being a reflective practitioner (Schön, 1983). On probing how participants specifically acted out the role of being a reflective practitioner, the focus group discussions revealed a myriad of strategies primarily linked to cyclical models of reflections espoused by Kolb (1984) and Gibbs (1988). They stressed how these models were used as a broad framework to reflect on practice as opposed to strictly adhered to and followed. Probing the use of the reflective models by Kolb (1984) and Gibbs (1988) by participants further determined how participants concurred with the limitations of these models, discussed within section 2.1, and thus identify with the interest in exploring metaphors and the Clean Language questions, as an alternative approach to reflection proposed by this study.

#### 4.3 Findings Target Domains

To determine and identify the messages being conveyed by a metaphor, the mapping and coding of the source to target domains consisted of matching the metaphors broadly to the NLP logical levels of learning (O'Connor & Seymour, 1994; 2002) consisting of environment,

behaviour, capabilities, values, beliefs, emotions and identity to help establish a starting point towards determining the concepts being conveyed by a metaphor (see section 3.3.1 for further details). The Clean Language PRO frames by Tompkins and Lawley (2006) were further used to determine the participants' perceptions of the frame surrounding metaphors (see sections 2.3 for further details).

A total of 674 target domains were coded within NVivo across the NLP learning levels, the analysis of which identified a trend whereby metaphors referred to by participants during the research interviews were predominantly linked to abstract concepts of 'behaviour', 'emotions' and 'environment', as represented by 71.16% of all coded target domains (see Appendix 13 ). Examples of the use of 'behaviour', 'emotions' and 'environment' target domains referred to by participants are shown in table 3 below:

Behaviour	Emotions	Environment
"I was too accommodating" [Participant 1]	"Feel just a little bit jaded" [Participant 2]	"I'm in an okay place" [Participant 1]
"I've got to get up there to get to it" [Participant 1]	"but I find it mentally draining" [Participant 2]	"It's quite barren, and there's very little vegetation around" [Participant 1]
"Being sharp enough to pick everything up" [Participant 4]	"In a bit of a bouncy mood" [Participant 6]	"Feel I've kind of plateaued a bit here" [Participant 2]
"I'm trying to perform rather than being myself" [Participant 10]	"that put me on edge slightly" [Participant 10]	"that isn't a good place to be" [Participant 2]
"I don't think I ever really got off the starting blocks" [Participant 2]	"I'm feeling a real buzz" [Participant 6]	"room is like a little sanctuary where they can be themselves" [Participant 10]
"That is really what's driving me to reflect" [Participant 2]	"Well that it feels a bit flat perhaps" [Participant 5]	"They might be there somewhere in the background" [Participant 9]

Table 3 - Examples of Coded Behaviour, Emotions and Environment Target Domains

The analysis of the target domains was compared against Tompkins and Lawley's (2006) PRO frames to determine the broader narratives expressed by a metaphor and to balance the way 'problems' and 'remedy' target domains were predominately represented within the metaphors voiced by participants over 'outcome' domains. More specifically, from an analysis of 251 source domains, which were coded against the PRO categories, the dominant use of problem

(40.64%) and remedy (41.04%) target domains overshadowed the use of outcome (18.33%) frames expressed within the metaphors referred to by interview participants.

These findings suggested and were further corroborated by the focus group discussions of participants expressing the importance of achieving their personal aims and formal curriculum learning objectives, as signified by the 'outcome' (PRO frame) when considering their teaching holistically, but within the classroom focus on 'remedying' situations and challenges when working with students.

Furthermore, focus group participants also linked the level of critical self-reflection into their practice to the concept of reflective practice and linked this to how their PGCE or similar studies placed a strong emphasis on reflection and being a reflective practitioner (Schön, 1983), though not wholly in keeping with the framework of models espoused by Kolb (1984) or Gibbs (1988).

#### 4.4 Discussion Points Emerging from the Analysis of Findings

The act of mapping source domains to target domains was aided by focusing on interpreting the generalisations and inferences of the conceptual metaphors identified via an analysis of the interview transcripts. Battino's (2005) suggestion of the need to identify "patterns that connect" was borne in mind to give an insight into the conscious and subconscious conceptual structures through which experiences are reflected and could be understood in the analysis of metaphors and the mapping of the source to the target domain (Botha, 2009). Reddy's (1979) influential work on 'conduit metaphors', mentioned by Lakoff and Johnson (2003), explored the notion of ideas being referred to as 'objects', linguistic terms as 'containers' and communication between these being a 'conduit' to the sender to digest and interpret, also aided in stressing the congruence necessary between the mapping of the source and target domains, and the intended message embedded within a metaphor. Within this study, Kövecses' (2010) source and target domains were employed as initial codes to analyse the metaphors and helped to simplify and identify the conduit of the metaphors expressed and the choice of containers referred to by participants interviewed in expressing a metaphor. Therefore, in the analysis of metaphors, either via discussions or qualitative research study, the importance of having a starting point in mapping and determining inferences of metaphors aids both the process of reflective practice and understanding of how experiences are

perceived and conceptualised. Within this study, the value of Kövecses' (2010) source and target domains, and supplementary coding variables added, was evident in enabling the fluent mapping analysis of metaphors referred to by participants, with the coding strategies adopted corresponding to the 'invariance principle', which Gibbs (2017) refers to in describing the importance of preserving consistency of mapping of source to target domains in identifying and analysing conceptual metaphors.

However, Kövecses (2010) identifies and cautions how the mapping and pairing of source domains to target domains could be problematic given the vast range of ideas and concepts (i.e., target domains and source domains), and the potential to identify arbitrary, as opposed to precise, conceptual metaphors when linking source to target domains. Issues evident in the analysis of metaphors within the interview transcripts were evident, and, therefore, an open-ended and broad approach to referring to a broad range of source domains was used to avoid a restrictive and arbitrary approach to the analysis of metaphors with additional source domains (see section 4.2 for further details). This was included as part of the analysis when the range of source domains identified by Kövecses (2010) was limited in its breadth. Furthermore, mapping source domains to target domains, due to the subjectivity involved, required rigour in the analysis of the metaphors to avoid the finding 'casting shadows' on the conclusions formed and potentially presenting a flawed, distorted or obscure understanding of the conceptual metaphors identified (Cook-Sather, 2003). Establishing rigour in the analysis was also managed by adopting narrow lists of target domains (as identified in section 4.3), allowing for a more precise mapping of source to target domains and identifying conceptual metaphors. For example, Participant 2's utterance "I don't think I ever really got off the starting blocks" could be matched to 'movement', 'location' or 'travel' source domains and further complicated by potentially being linked to 'behaviour', 'thoughts' or 'feelings' target domains. Given the potential subjectivity involved in the identification and interpretation of metaphors uttered by participants, Schmitt (2005) argues that a process of analysing metaphors should give some consideration to the triangulation of findings. This process, Armstrong et al. (2011) stress, can be established by allowing participants to review and input into the analysis of the findings at varying stages of the research process. This was achieved by allowing participants to review and reflect on the interview transcripts, both individually and collectively within a focus group, and to comment on the pattern themes emerging from the analysis of the interview transcripts. Thus, the significance of verifying the inferences drawn from the source and target domains is important in accurately determining the intentions and meaning of the conceptual metaphors with participants.

97

Finally, the mapping of source and target domains was further aided by the researchers' professional and vocational understanding, and experiences of teaching within a range of HE settings. The analysis of the interview transcripts was also undertaken by a researcher, who, as an experienced academic, was able to interpret and map the metaphors being conveyed by the participants by being culturally and socially embedded in the world of academia and able to contextualise the message behind the source and target domains of the metaphors expressed (Schmitt, 2005).

As reported in section 4.2, a closer examination of the metaphors analysed within the interview transcripts revealed a pattern in the prevalent use of metaphors expressing movement and locations, for example:

## "It feels like there's a flow to what I'm doing" [Participant 7] "Your mind is darting around from point to point" [Participant 10]

However, it is important to note that the common understanding of a metaphor within everyday language differed within a teaching context and hence needed to be considered in keeping with the context of the discussion. For example, a participant's metaphor *"that it feels like there's a flow to what I'm doing"* [Participant 7], within the broader discussion of the research interview, referred to a measure of satisfaction in working with students while teaching. Similarly, *"your mind is darting around from point to point"* [Participant 10] related to aspects of the participant's level of productivity and the concentration needed to manage competing demands. Therefore, within the analysis of metaphors as part of the study, a specific focus on detecting and categorising broad conceptual metaphors, which were further confirmed during the focus group discussions, was made in concluding the analysis of interview transcripts.

In analysing the expression of the metaphors uttered by participants, a trend emerged whereby Lakoff and Johnson's (2003) categories of structural (where a concept being conveyed is structured and can be mapped to another concept), orientational (groupings of systems) and ontological (physical objects or substances) classifications were evident. For example, a pattern where concrete aspects of experiences were described more in relation to structural metaphors (e.g., *"being stuck at the top of that mountain"* [Participant 2], *"like you're not in control, you're just riding the wave"* [Participant 10]), and thoughts and feelings via orientational metaphors (e.g., *"that I've got to climb I've got to get up there myself"* [Participant 1], *"at the moment they're all up in the air"* [Participant 4]). The universality of the orientational

metaphors describing positive elements of a phenomenon labelled as being on a 'high' or having an upward trend and harmful elements described via shallow, downwards trending terms was also evident within participants' metaphors. Common elements of metaphors Kövecses (2010) also refers to as 'image schema' metaphors, where broad elements of a structure represent a concept conveyed and contrast with a narrow mapping of source to target domains represented by structural metaphors. Hence, specific and universal patterns were apparent in the expression of metaphors, reflecting aspects of the phenomenon regarding behaviour predominantly expressed via structural metaphors, and cognitions and emotions utilising orientational metaphors. The use of specific categories and classifications referred to within metaphors highlights an important distinction of relevance in the popularity of elements, whereby the common understanding of concepts exists (i.e., positive and negative elements of experiences described as terms of high vs low, or colours representing dark and light). These patterns of reflection on experiences could be utilised by facilitators to aid the reflective process by referring to structural metaphors when discussing aspects of practice within the reflective process and orientational metaphors when exploring thoughts and emotions.

The implications of the popularity of the movement and location source and target domains were reinforced via feedback from the focus group discussion reporting experiences in terms of travelling between two points of interest as symbolic of representing a heavy focus on the need to cover curriculum assessments and to meet students' needs. This prevailing mindset reflected on participants' awareness of the potential of assessment outcomes being used as a metric to evaluate personal performance as part of university appraisal systems and the broader implications of the national *Teaching Excellence Framework* policy as being a policy of relevance at the time of the research interviews.

In summarising the dominant use of movement and location source and target domains as a conceptual metaphor, the focus group was asked to discuss a selection of terms as an initial starting point to identify the conceptual metaphor of teaching in relation to the movement and location domains. These terms consisted of:

Teaching is an adventure, an excursion, exploration, expedition, quest, journey, wandering, trek.

The above terms were gleaned from the findings and aimed at isolating the conceptual metaphors symbolic of participants' teaching practice, generated some discussion and

detailed analysis of the semantics of the terms presented within the focus group. While the focus group put forward no new terms, participants leaned on preferences towards teaching as an adventure and a journey as conceptual metaphors. On being asked to agree on a conceptual metaphor, participants in the group unanimously identified **'teaching is a journey'** as best, symbolising their teaching practice's conceptual metaphor.

In further considering the findings identifying participants' metaphors leaning towards reflecting and critically appraising their teaching, focus group participants suggested reflective practice and encouraging academics to reflect on decisions, actions and behaviours taken while teaching impacted their teaching practice. To identify and isolate a conceptual metaphor symbolic of the participants' focus of reflecting and critically analysing their teaching practice, focus group participants were asked to consider a range of phrases derived from the analysis and mapping of the source and target domains they deemed symbolic of their focus on self-analysis and reflection. These phrases consisted of:

# Teaching is an act, presentation, display, performance, production, theatre, arena.

Discussion within the focus group on the relevance of these terms identified teaching as dance, play and concert, as additional phrases representative of participants' conceptual metaphors of their teaching practice. However, all focus group participants agreed how these terms exemplified the importance they placed on reflecting on their actions and decisions while teaching in preference to other aspects of teaching, such as planning teaching or assessments. After some discussion, the focus group agreed how the conceptual metaphor 'teaching is a performance' was most symbolic of their focus on their teaching practice. The identification of the 'teaching is a journey' and 'teaching is a performance' metaphors being representative and symbolic of academic conceptual systems is significant in representing an attempt to map and identify conceptual metaphors within a teaching context. Mainly, given how previous attempts at identifying and listing conceptual metaphors, for example, Lakoff et al.'s (1991) Master Metaphor List, specifically focused on identifying generic conceptual metaphors in an everyday context. Chapter 5 discusses the relevance and significance of the 'teaching is a journey' and 'teaching is a performance' conceptual metaphors in more depth as part of research question two, which explores the relationships between academics' use of metaphors and their narratives of teaching.

The findings and reflections of participants further support how personal recollections and stories of experiences, in essence, give metaphorical insights into the inner cognitive perceptions and interpretation of the reality and phenomenon experienced. This is a view previously echoed by Saban's (2006) assertions of how:

"metaphors structure our perception, thought, and action. For better or worse, they fundamentally affect our way of conceptualizing the world and reality whether we are aware of this phenomenon or not" (p. 299).

The capacity and ability of conceptual metaphors to convey thought processes and identify conceptual systems are seen as a key cognitive function of metaphors (Lakoff & Johnson, 2003; Saban, 2006) due to the premise that metaphors give preference to specific thoughts and beliefs over others (Cook-Sather, 2003; Mahlios et al., 2010). This was a view identified by participants as being significant via focus group comments, stating:

"It would be useful to know what your analysis of the interviews shows. Which metaphors were referred to and what they convey?" [Participant 4]

"Interesting themes emerging but need to really think about wider implications [of the relevance of metaphors] of this" [Participant 2]

Focus group comments from participants were representative of broader curiosity in interpreting the significance of conceptual metaphors expressed by participants and interest in using metaphors supported by the following statements:

"Didn't appreciate how often we talk in metaphors. They appear to be deeply ingrained in our thoughts" (Participant 2)

"Never really thought before of metaphors as being significant [until now]. Just a way of expressing yourself" (Participant 3)

"The use of metaphors was a really interesting way to think about my relationship with my professional practice. I'm curious to know about the metaphors your other interviewees used. How I relate to them and whether I might perceive my own practice differently through other people's metaphors" [anonymous questionnaire comment]

These findings stressed increased alertness of the relevance and significance of metaphors as representing something more than mere linguistic utterances, with interest in the premise of the power of metaphors to confer deeper meaning and be instrumental in conveying the unconscious conceptual structures, which could also be referred to as a narrative, of the way individuals may think and act, evident in the focus group discussions.

On reviewing their interview transcripts, focus group participants recounted how they believed the metaphors they referred to while reflecting on their experiences were entrenched and representative of their thoughts. They also somehow inferred broader meanings as evidenced by comments stressing:

"Didn't appreciate how often we talk in metaphors. They appear to be deeply ingrained in our thoughts" (Participant 2)

*"Never really thought before of metaphors as being significant. Just a way of expressing yourself" (Participant 3)* 

*"I would be curious to know about what the metaphors I referred to says about me" (Participant 4)* 

The above comments suggest how participants interviewed as part of the research were not familiar with or aware of the notion and theory of conceptual metaphors, predominately believing metaphors to be linguistic terms referred to as a way of expressing what they wish to say with focus group discussion showing some evidence of a shift in perception of the importance of metaphors in everyday discussion amongst participants. These findings further corroborated Lawley and Tompkins' (2000) discussion of the isomorphic qualities of metaphors, with participants expressing how they believed the metaphors referred to within the research interviews matched and conveyed their more profound thoughts and aided in allowing them to make sense of their thought patterns via metaphorical analogies that corresponded to their thoughts. The study results also concurred with Lakoff and Johnson's (2003) observation of how the mapping of concrete source domains to broad target domains aided in identifying the conceptual metaphors referred to by participants, as demonstrated by the coding decisions further discussed in sections 3.3 and 4.2.

The focus group discussions also noted how the relationship between metaphors represents a wider personal story or narratives of experiences; a fundamental premise underpinning the notion of Lakoff and Johnson's (1980; 2003) conceptual metaphors and further supported by Ho (2005) and Saban (2006) in stressing how personal stories of experiences are fundamentally metaphorical and how the choice of metaphors referred to by an individual reflected their reality and conceptualisation of a phenomenon. This premise links to the hypothesis of the value of utilising and focusing on metaphors as part of the reflective process and the broader narrative and stories a metaphor may represent (Clandinin & Connelly, 2000; Bold, 2012).

However, caution needs to be expressed in forming generalisations on the analysis and interpretation of specific conceptual metaphors due to influences of context and inferences intended by the wider population. Lakoff and Johnson's (1980; 2003) original work on conceptual metaphors identified how embracing one metaphorical concept, by default, would hide other aspects of a concept and could give preference to perceptions that are congruent with the idea at the expense of those that are not (Cook-Sather, 2003). This is an aspect of working with metaphors which Armstrong et al. (2011) cautioned against, acknowledging how *"a degree of uncertainty is always present in any post-structural readings of metaphors*" (p. 357). These caveats related to interpreting metaphors were further expressed within the focus group, whereby participants queried:

"Not sure if we would always use the same metaphors. If we were to complete the same interview on another day, would I have used the same metaphor again? I don't know, suppose metaphors are significant if there is a pattern to your use of them" (Participant 1)

"Wouldn't any other aspect of language and the words we say also be important? Maybe anything we pay attention to could be deemed as being important" (Participant 3)

These caveats are further discussed and addressed within chapter 7, where the conclusions drawn from the study in answering the research questions are considered.

#### Chapter Summary

In summary, an analysis of the dominant metaphors used in the way academics recall, reflect and describe their teaching experiences identified trends regarding how academics reflected on aspects of their teaching practice. Specifically, the findings determined the key mapping process between source and target domains was aided by adopting a broad, open-ended approach to the matching of source domains to a narrow band of target domains in keeping with the pattern of concrete source domains matching more abstract target domains (Lakoff & Johnson, 2003; Kövecses, 2010). The findings also tied into the theories and concepts related to the seminal work of Lakoff and Johnson (2003) and Kövecses (2010) into conceptual metaphors with regards to the mapping of concrete source domains to broader abstract target domains. Findings further identified the dominant use of three abstract source domains (movement and directions, place and locations, and senses) in the way academics reflected on aspects of their teaching practice. Source domains, which participants cited as representative of their focus on teaching and curriculum aims, also implied the relevance of metaphors in representing deeper levels of cognition. Similarly, the popularity of target domains denoting behaviour, emotions and environment identified from the analysis of the interview transcripts was confirmed by participants as representative of aspects of their teaching in the classroom.

A sense of conventionality and universality in the use of metaphors was evident within the findings, however, these were dependent on the context of the issues being discussed and predominately tied to structural metaphors regarding aspects of behaviours and orientational metaphors expressing thoughts and emotions in preference to ontological metaphors. Similarly, the use of objects, containers and conduit metaphors (Reddy, 1979; Lakoff & Johnson, 1980; 2003) was identifiable from the analysis of the metaphors and aided by an appreciation of the professional and vocational context of the discussions within the research interview.

The significant conceptual metaphors emerging from the analysis of the data and corroborated by focus group discussions fundamentally agree with Lakoff and Johnson's (1980; 2003) theory of conceptual metaphors. Furthermore, participants concurred with how the metaphors reflected and were representative of a deep level of thinking and broader narratives of how they perceive and interpret their teaching practice.

The identification of 'teaching is a journey' and 'teaching is a performance' conceptual metaphors, emerging out of the analysis of the interview transcripts and mapping of popular source and target domains, was agreed on by the participants as being symbolic in representing their conceptual metaphors and perceptions of their teaching practice. However, caution was expressed and acknowledged in how a focus and analysis of a specific metaphor by default gives significance to a particular pattern or mode of thinking over another.

The use of metaphors to convey experiences was evident in the data analysis via a preference for the use of metaphors over similes or analogies. The often-cited claim of using six metaphors a minute by Geary (2011) was not determined, primarily given the uncertainty of the context in which the claim is made, differences in definitions of a metaphor and the Clean Language questions through which the interviews were conducted.

# Chapter 5 – Analysis of Participants' Use of Metaphors and their Narratives of Teaching

Further to the identification of the dominant metaphors, via analysis of interview transcripts and focus group discussions (see sections 4.2 and 4.3), suggesting academics primarily view their teaching practice via 'teaching is a journey' and, to a lesser extent, 'teaching is a performance' conceptual metaphors, this chapter debates a deeper analysis of these conceptual metaphors and the emerging narratives guiding thoughts and behaviours, and affecting participants' teaching practice is needed to address the second research question: "What are the relationships between academics' use of metaphors and their narratives of teaching?"

To determine the relationships between academics' use of 'teaching is a journey' and 'teaching is a performance' conceptual metaphors and their narratives of teaching, in keeping with the conceptual framework, the following findings will review and discuss the relevance and significance of the conceptual metaphors through three levels of analysis to investigate the narratives symbolic of academics' teaching practice. These three levels of analysis are further discussed within section 3.3 of the methodology and consist of:

- Tompkins and Lawley's (2006) PRO model to identify the broader frames of the narrative.
- McAdams and McLean's (2013) Narrative Identity Constructs to identify the common themes underpinning academics' reflections of their teaching practice.
- Schwartz's (2012) Universal Values list to identify the deeper values and beliefs underpinning the narrative.

The coding analysis of the conceptual metaphors against these levels of analysis was measured in terms of matching the conceptual metaphors within the interview transcripts, with these findings subsequently further explored and corroborated by analysis of questionnaire comments and focus group discussions, and then by subsequent scrutiny and reflection of the analysis of findings from a theoretical perspective (see section 2.3 for further details of the use of the three levels of narrative analysis referred to within this study and section 3.2 for further details of the use of the use of research methods).

This chapter commences by discussing the findings of the narrative analysis of the conceptual metaphors in section 5.1, before the broader significance of the narratives emerging from the findings are discussed within section 5.2.

#### 5.1 Findings of Narrative Coding Analysis of Conceptual Metaphors

5.1.1 Reporting on focus group reflections of conceptual metaphors identified

The significance of the 'teaching is a journey' conceptual metaphor was reflected on individually by participants and collectively within the focus group discussions. Participants within the focus group concurred that they identified with and saw the 'teaching is a journey' conceptual metaphor as representing how they perceived their teaching and gave insights into the strategies they employed within their teaching practice to plan, prepare and teach lectures and seminars. For example, participants recounted how the 'journey' metaphor was apt in mirroring the start and end-points of the teaching process as illustrated by the following focus group comments:

"By the end of it, this is what we're going to have done or where we're going to be" (focus group comment) "I agree. I am always thinking about where I am going with my sessions and with my students" (focus group comment) "Something to do with covering learning outcomes" (focus group comment)

The above comments, and subsequent focus group discussions, reflected how participants continually had in mind the learning outcomes they were required to achieve while teaching and concurred that the 'teaching is a journey' conceptual metaphor was apt in symbolically identifying how the starting point of their teaching journey linked to their focus on meeting module learning outcomes.

Participants, both individually and within the focus group, in identifying with the 'teaching is a journey' conceptual metaphor, further probed the conceptual metaphor by considering who were the passengers on the metaphorical journey and how the metaphor may give insights into their approach to their teaching with regards to how they planned lectures and seminars and taught within the classroom. Regarding passengers, participants unanimously identified 'students' as the passengers on the journey as characterised by the following comments:

"...about thinking at the end of this session where do I want the students to be" (focus group comment)

"Just as I watch my students flourish" (focus group comment)

"In my lesson planning, I do often think about how I take students on a learning journey" (focus group comment)

A sense of enthusiasm and expectation in taking the journey was also evident, with participants within the focus group reporting:

"Because there's a learning experience for you in there as well, I don't see teaching for me as filling empty vessels" (focus group comment)

"...that is a real joy for a teacher, and that's what keeps us going" (focus group comment

The completion and end-point of the journey were emphasised via links to course assessments through focus group comments, stating:

"Understand how a metaphor of a journey is apt. We are always going somewhere to meet targets, doing something to students to complete assessments and assignments" (focus group comment)

"At the end of the day, it all boils down to assessments and helping students graduate" (focus group comment)

These, and other similar focus group participants' comments, suggest a sense of agreement and acknowledgement into the nuances of 'teaching is a journey' conceptual metaphors as being symbolic of how they contextualise and conceptualise their teaching practice in keeping with findings discussed within chapter 4 and chapter 6 into the use of the Clean Language questioning model to examine metaphors. Differences in what specifically denoted the journey were apparent, with some participants extolling a preference towards enhancing the student experience, while others focused on the importance of working towards assessment and learning outcomes. This is a distinction discussed later in this chapter regarding values leaning towards either a process versus the product of teaching (see section 5.2).

Participants within the focus group further wished to explore and compare how alternative conceptual metaphors could have led to a different narrative and conceptual understanding of their teaching practice, as suggested by the following comments:

"Wonder if we adopted the metaphor of teaching as being a sport or an orchestra. Would we have a different view of how we teach?" (focus group comment)
"I'm a sports fan, and if I thought of teaching as a game, I think there would be some mileage in exploring how the game is played out in the classroom?" (focus group comment)

"I teach counselling and always considered my teaching as a conversation between the students and me. There is always much discussion within the sessions I teach" (focus group comment)

Together with further focus group discussions, these comments helped to concur with how some relevance existed in how the choice of conceptual metaphors could potentially impact teaching practice, specifically with regards to how the choice of conceptual metaphors aligned with Lakoff and Johnson's (1980; 2003) theory of conceptual metaphors being representative of a person's conceptual systems, and how McAdams' (2001) notions of how individuals make choices to tell stories that are meaningful to them and integrate with their actions and behaviours. The focus group participants comment on how alternative metaphors could yield different insights and approaches to teaching, help to demonstrate how reference to an alternative metaphor and telling of a different story could potentially offer alternative interpretations of their teaching practice, and direct their perceptions of how they approach their teaching practice in keeping with the conceptual metaphor.

Participants were also able to digest and dissect the secondary conceptual metaphor of teaching practice akin to a 'performance'. Namely, in terms of performance linked to good teaching as denoted by participants within the interview transcripts stating:

"You can see the learning happening" [Participant 2] "See people growing" [Participant 2] "I'm waving my baton, and the orchestra is in tune" [Participant 8]

These sample participants' comments suggest differences existed in how participants perceived aspects of performance while teaching, with measures alluding to sensory confirmation, control and congruence being referred to within the above examples and potentially construed as positive or negative aspects of teaching. For example, the metaphoric phrase *"I'm waving my baton, and the orchestra is in tune"* [Participant 8] could be interpreted as guiding students or as controlling or correcting students to get in tune. However, further probing of participants' interpretations of the 'teaching as a performance' conceptual metaphors showed how participants perceived the metaphor to be akin to and symbolic of good teaching by employing a range of measures consisting of their thoughts and feelings, judging participation levels, and mood of the group and external sources of feedback

(students' evaluations, peer review comments, etc.), with congruence between these three elements being taken as signs of good teaching.

Other interpretations of 'teaching as a performance' conceptual metaphors related to 'not performing well' as alluded to by participants within the interview transcripts, stressing:

"Musical instruments or the musicians are out of tune" [Participant 8] "I'm trying to perform rather than being myself" [Participant 10] "it's when you see those cartoon characters with clouds over their head" [Participant 9]

These sample comments of participants 'not performing well' suggest incongruence and low mood attached to occasions when student disposition, participation and energy levels are depleted. However, these examples of believing that they were not teaching well were countered by utterances denoting the need to teach and perform better, as signified by the following participant comments:

"How do I grab students' attention?" [Participant 5] "Because you tend to be concentrating on your own performance when you first start teaching [referring to a career in teaching]" [Participant 6]

The act of considering storytelling and exploration of conceptual metaphors as an approach to reflective practice was commented on within the questionnaire via the following responses:

"I found it all (storytelling and exploring metaphors) useful and it aided my powers of reflection" [Participant 6]

"I really found the interviewing interesting, and it made me think about my own practice in several different ways" [Participant 7]

Participants further commented on the value of reflecting on and exploring their conceptual metaphors for teaching in their questionnaire responses, for example:

"Teaching and learning is complex, and that thinking through the metaphors enabled me to re-consider the pedagogical processes we (I) use with students" [Participant 10]

"It helped me think quite deeply about my teaching practice and approach. Good technique – consistently brought me back to reexamine metaphors and descriptions I had used to try to drill down what I may have meant/understood in describing my practice" [Participant 5] "The use of metaphors was a really interesting way to think about my relationship with my professional practice. I'm curious to know about the metaphors your other interviewees used. How I relate to them and whether I might perceive my own practice differently through other people's metaphors" [Participant 3]

The above questionnaire comments were further elaborated during the focus group discussions, with participants stating:

"It seems obvious now but not previously considered how telling stories of our experiences are actually part of the reflective process"

"I always thought of reflective practice as something more abstract and broader. Homing in on and exploring metaphors we use to explore our experiences was novel and very interesting"

These participant observations suggest how probing the relevance of a conceptual metaphor could yield a greater level of reflection into the complexities of teaching, potentially not possible by examining teaching experiences via traditional models of reflection espoused by Kolb (1984) and Gibbs (1988). Primarily, this may be explained by how these cyclical models of reflection emphasise and promote a procedural approach to analysing experiences, instead of a more open approach to analysing all aspects of experiences via metaphors. Furthermore, the focus on referring to and exploring metaphors while reflecting could avoid employing mechanical cyclical approaches of reflection requiring practitioners to reflect on demand and often out of context of their teaching practice (Boud & Walker, 1998; Surgenor, 2011). Additionally, a move towards a focus on exploring metaphors and the use of Clean Language questions as a method of reflection, whereby academics can freely reflect on their practice as part of a conversation as promoted by Hoffman-Kipp et al. (2003) and Guo (2021), acknowledges the value of reflection via discussions and counters criticism of Schön's (1987; 1988) writing on reflection predominantly promoting reflective practice as a solitary activity. Hence, the findings suggest an opening exists, where a focus on exploring conceptual metaphors to encourage reflective practice is evident, and helped to inform the development of the proposed model of reflection originating from the findings of this study (see section 7.3 for further details).

#### 5.1.2 Findings of analysis of PRO categories

The process of determining links between metaphors and participants' narratives of their teaching practices was undertaken by analysis of the two identified conceptual metaphors (i.e., teaching is a journey and teaching is a performance) against Tompkins and Lawley's

(2006) PRO model. This initial level of analysis aimed to establish an understanding of participants' narratives of their teaching practice, prior to the deeper level of analysis of the link between participants' conceptual metaphors and narratives of teaching through McAdams and McLean's (2013) Narrative Constructs and Schwartz's (2012) Universal Values list.

Discussions of the relevance and significance of the analysis of interviews against Tompkins and Lawley's (2006) PRO frames were positive and welcomed via participant focus group comments stating:

"Wasn't aware of the concept of framing and that my reflections had a particular slant to them" (focus group comment)

"In terms of insight, you could look way too deep into these things and end up navel-gazing, but I like the idea of how a simple frame (PRO model) can help to structure your reflections" (focus group comment)

"Not heard of the Problem, Outcomes, Remedy model before but think this could be something both use to reflect and use in my teaching" (focus group comment)

Of the 251 coding references attached to the PRO categories, an overwhelming 81.67% of coding matched the 'problem' or 'remedy' frame category, with both the frame categories being represented equally (problem – 41.04% and remedy – 40.64%). Reflections of these findings with the focus group identified a consensus whereby participants recognised how they consciously and unconsciously monitored their teaching when working with a group for potential problems (as denoted by the Tompkins and Lawley 'problem' frame) that may hinder learning and making adaptions (characterised by the 'remedy' frame) to their teaching practice when there appears to be a lack of engagement by students. Participants within the focus group further reported a sense of satisfaction of how their proactive approach to teaching was recognised within the conceptual metaphors referred to during the interviews and findings, and PRO framing categories, via comments stating:

"I think this PRO model, which I have not come across before, sums up neatly how I watch out for problems while teaching and making changes to the planned sessions where necessary"

"I think the PRO model highlights my level of self-awareness when teaching and working with a group, in that I look out for when teaching is not going well and always strive to mix things up to get the students motivated to engage with the session again" On reviewing the relevance of the PRO model to participants' teaching practice within the focus group, participants expressed mixed views of the model's value in informing their reflections of their teaching. On a personal level, participants discussed how they believed the PRO model offered a practical but somewhat simplistic way of understanding how reflections can be framed within categories which they could identify and relate to, and also noted how the emphasis on navigating challenges is mirrored within Tompkins and Lawley's (2006) 'problem' and 'remedy' framing categories. From a teaching perspective, participants also discussed how the PRO model had wider potential to be used within a teaching context and working with students in helping them identify how their students framed and conceptualised aspects of their learning. However, as part of the focus group discussion, some participants identified the PRO model as lacking in not giving a more detailed account of their narratives for teaching, through comments suggesting:

"I like this concept of working with frames and can see how the PRO model helps towards that, but surely within academic circles, we can think and identify frames that suggest a higher, deeper level of thinking and reflections" (focus group comment)

"There must be other framing categories and models we can work with to inform our reflections and help students understand theirs? But I do like the PRO model as a way of summing up our reflections of experiences in a way that is easy to understand and follow" (focus group comment)

These comments and observations were echoed by participants within the focus group in suggesting they wanted to know more about the wider implications of how the conceptual metaphors might link to their narratives of teaching and therefore welcomed the more profound analysis of the metaphor, which was forthcoming via the analysis of the conceptual metaphors against McAdams and McLean's (2013) Narrative Constructs.

#### 5.1.3 Findings of analysis of McAdams and McLean's Narrative Constructs

A total of 758 coding references were matched against McAdams and McLean's (2013) Narrative Identity Constructs from analysis of the metaphors identified within the interview transcripts to procure a deeper level of understanding of how participants' metaphors of their teaching experiences were representative of and informed their teaching practice. Of the seven Narrative Construct categories listed by and detailed in Appendix eight, a key pattern emerged whereby three key Narrative Constructs dominated 75.98% of all metaphors matched to McAdams and McLean's (2013) Narrative Constructs. Namely, these were:

- Exploratory narrative processing construct (31.12% of all coding references) Described as the extent that the narrator engages in self-exploration while telling a story
- Agency (27.18% of all coding references)

Refers to the extent to which the narrator is autonomous and has the power to affect their own life/decisions

Redemption stories (17.68% dominance of coding references)
The narrator transitions from a generally bad/negative state to a generally good/positive state

Discussing the relevance of these findings with the focus group, after explaining the definitions underpinning the dominant Narrative Constructs to participants, some consensus into McAdams and McLean's (2013) Narrative Constructs existed with participants identifying with the significance of the findings through example comments indicating:

"This is really interesting; it makes me think we could make use of this as a way of making sense of our reflections"

"This reminds me of Booker's book on seven story plots, but more relevant than the seven plots, I think, not sure how, but I'd like to find out more about McAdams' theory of narratives"

McAdams and McLean (2013) define the exploratory narrative processing construct as the "extent of self-exploration as expressed in a story" (p. 234), concurred with focus group reflections regarding the depth participants believed they are required to reflect on their teaching practice. This emphasis on reflective practice was reported earlier (within chapter 4), correlating with participants' observations of their quest to continually enhance their teaching practice stemming from the encouragement to reflect on their teaching and be a reflective practitioner (Schön, 1983). Furthermore, McAdams and McLean's (2013) 'agency' construct, defined as the extent individuals perceive they have autonomy and the power to affect their actions, focus group participants linked to corresponding to their ability to facilitate and manage their teaching.

Similarly, findings identifying the use of 'redemptive' Narrative Constructs – the transition within a story indicating a move between a negative to a positive state – was seen by focus group participants to be in keeping with the optimistic and constructive tone of reflecting on experiences to enhance practice. This was particularly evident within the focus group, with participants' comments stating:

"The definition you give of redemptive narrative stories fits my thinking very well. Striving to move from negative to positive teaching, I always try to do, and think is something we should all strive for" (focus group comment)

"Thinking about this [redemptive narrative stories] and our earlier discussions on reflection. Isn't the reason we reflect on our teaching all about identifying what isn't working and making changes to our teaching practice towards what does work?" (focus group comment)

These participant reflections and seeming popularity of the reference to the use of 'redemptive' narrative stories within the focus group discussions, suggest a positive slant towards participants' optimistic solutions-focused approach to their teaching practice aligning with Tompkins and Lawley's (2006) remedy frames of the conceptual metaphors referred to by participants; the wider significance of these findings is further discussed in section 5.2. Furthermore, the positivity of participants' approaches to redemptive stories aligns with McAdams and McLean's (2013) observations of individuals who tell redemptive stories tending to be more generative and have higher levels of well-being than individuals who prefer 'contamination' stories of their experiences; a phenomena McAdams and McLean (2013) specifically discuss in relation to individuals' personal stories of their experiences, but could also be translated within a teaching context of how a problem-solving optimistic approach to teaching could yield a greater sense of well-being of academics.

Habermas and Bluck (2000) and Daiute (2013) identify the need to determine the coherence of the analysis into narratives. Hence, the dominant Narrative Constructs ('exploratory narrative processing', 'agency' and 'redemption') emerging out of the analysis of the conceptual metaphors were put to the focus group participants to determine their significance. After sharing McAdams and McLean's (2013) definitions of the three dominant constructs with participants, focus group participants were able to identify a thread linking the three dominant narratives' identity constructs as suggested by focus group comments as follows:

"Isn't exploratory narrative processing a form of reflection? You [researcher] mention the word self-exploration. I would then link this to the reflection that we are encouraged to practise while teaching" (focus group comment)

"I would say the same, exploratory narrative processing is about reflection and self-awareness, and the other construct [redemptive stories] is about changing practice from reflection. Actually, maybe the stuff about 'agency' is about the responsibility we take in changing our practice as a result of reflection" (focus group comment) "It seems to me we are going around in circles here, not in a bad way, but that what you have found in your research links to reflection very closely, not sure if that was what you intended or what you have found from your analysis, but there does seem to be a clear link between your narrative constructs and reflective practice" (focus group comment)

These and other similar participant observations suggest and identify a link between the three dominant Narrative Constructs and reflective practice. More specifically, this link could be denoted as a narrative strand whereby participants' focus on reflective practice and self-awareness is akin to and exemplified by McAdams and McLean's (2013) 'exploratory narrative processing' and 'redemptive' narrative agency constructs related to the actions taken to improve aspects of their teaching practice. Thus, Habermas and Bluck (2000) and Daiute's (2013) call for determining coherence in the analysis of narrative could relate to how aspects of reflective practice are mirrored and align with the three dominant Narrative Constructs arising out of the analysis of the conceptual metaphors determined from the interview transcripts.

In summary, analysis of the metaphors emerging out of the interview transcripts against McAdams and McLean's (2013) Narrative Constructs identified a significant narrative strand implying a focus on reflective practice (exploratory narrative processing), autonomy and self-sufficiency in actions (agency).

#### 5.1.4 Findings of analysis of Schwartz's Universal Values list

A further deeper layer of analysis of the interview transcripts was completed by coding the interview transcripts against Schwartz's (2012) Universal Values list (as detailed in section 2.3 and section 3.3) to explore the narratives underpinning participants' use of conceptual metaphors. This analysis layer resulted in 335 coding references of the metaphors identified in the interview transcripts matched to Schwartz's (2012) Universal Values list. Analysis of the coding of the interview transcripts identified a pattern whereby a strong preference towards the following values linked to Schwartz's (2012) emerged from the findings:

- Self-direction (71.4% of all coded references) Emphasis on placing importance towards independent thought, choosing actions, creating and exploring
- Stimulation (15.82% of all coded references) Placing importance on seeking excitement, novelty and challenge in life

Achievement (13.13% of all coded references)
Personal success through demonstrating competence according to social standards

The above values emerging from the findings indicate the importance of reflecting on teaching practice by the participants, and, more specifically, the beliefs that function as internalised guides that define behaviours to desired goals (Schwartz, 2012). As such, participants within the focus group discussion both identified and subscribed to the three fundamental values emerging from the findings via the following comments:

"Erm, interesting how I can relate to the findings of values. Not given this much thought before but can relate to the three values identified" (focus group comment)

"I'm surprised how we have gone from talking about metaphors to now values. Initially, I thought this was a big leap, but I can see the link between metaphors, narratives and values" (focus group comment)

"I'm particularly interested in how this notion of values acts as guides directing our motivations and actions. I kind of knew about this in the back of my mind but never given this much thought before" (focus group comment)

Further exploring the relevance of these values within a personal and wider teaching context, participants expressed some surprise to discover how these findings link to earlier discussions around the use of Tompkins and Lawley's (2006) PRO model, and McAdams and McLean's (2013) Narrative Constructs via the following comments:

"Not sure how this links to the PRO model, but I think there are some similarities between the values we are talking about here and the narratives ideas stuff [McAdams and McLean's Narrative Constructs] we discussed earlier. I mean, isn't self-direction all about agency?" (focus group comment)

"I can identify and relate to 'self-direction' being the top value, as I like the autonomy of being in control and deciding what I want to do, both in my teaching and personal life" (focus group comment)

"I am not sure how representative the percentages are, but I can identify with the value of stimulation as I constantly challenge myself to do and try new things" (focus group comment)

"Not really surprised achievement featured lower than self-direction and stimulation values. For myself, and maybe this relates to others too, I'm more bothered about being stimulated and motivated about teaching than achievement on a personal level. But that is not to say I don't value achievement in terms of students' success in assessments" (focus group comment)

The above comments and similar discussions within the focus group help to show how participants interpreted and identified the findings of values embedded within the metaphors aligning with the earlier Narrative Constructs on 'agency' and the need to be able to exercise and have autonomy in how they teach. The finding of 'stimulation' as the second most popular value acknowledged the importance participants placed on excitement and being challenged in how they teach, and, to a lesser extent, the value of 'achievement' demonstrative of the personal competence, success and recognition associated with good teaching practices.

Schwartz (2012) identified how the expression and pursuit of some values could conflict with other values and therefore create dissonance and incongruence between individuals' motivations, decisions and actions. More specifically, figure 2 below illustrates how all the values of Schwartz's (2012) Universal Values list can be grouped within four categories of openness to change, self-transcendence, self-enhancement and conservation.



Schwartz (2012) stresses how a need for and importance of congruence and alignment between the values can have psychological and practical consequences. For example, a person's conceptual system based on values seeking security and stimulation would conflict due to these values directing thoughts and behaviours at polar opposite categories of 'openness to change' vs 'conservation'.

Within Schwartz's (2012) values categories, an important observation emerging from the findings into the values embedded into the dominant conceptual metaphors relates to the

alignment and congruence between the three key values determined by the findings. More specifically, by expressing and following values of self-direction, stimulation and achievement within their teaching practice, a pattern emerges where Schwartz's (2012) values categories of 'self-enhancement' and 'openness to change' align, and would not conflict in impacting participants' decisions and actions towards their goals. Furthermore, these values align with McAdams and McLean's (2013) dominant Narrative Constructs whereby 'exploratory narrative processing', 'agency' and redemptive stories reflect and mirror the values of 'self-enhancement' and 'openness to change', reinforcing the coherence between Narrative Construct strand and values.

Dauite (2013) identifies how individuals consciously and unconsciously negotiate values, and implicitly and explicitly enact them in discussing values within narrative contexts. Furthermore, Dauite (2013) suggests values echo diverse personal, situational and cultural factors giving preference and importance to specific beliefs, thoughts and behaviours. Hence, participants' broader relevance of values within the conceptual metaphors and the narrative they convey need to be considered and further discussed in section 5.2 below.

# 5.2 Discussion into the Significance of Narratives Emerging from the Findings

Bleakley (2005) identifies narratives as a process of endowing reflections of experiences with meaning; a process going beyond merely recollecting experiences towards determining the preferred frames, mindset and schema surrounding reflections to answer what Dauite (2013) refers to as the 'now what?' question in drawing narrative inferences from the findings. Within the context of this study, this refers to exploring the relevance of conceptual metaphors identified as being symbolic in endowing narrative meaning on experiences to answer further the second research question relating to determining the relationships between academics' use of metaphors and their narratives of teaching.

The following account discusses the inferences and implications of the narratives underpinning the conceptual metaphors identified via the findings. Subsequently, the broader understanding of the narratives emerging from the findings is further discussed to explore the relationship between participants' teaching practices' conceptual metaphors and academics' narratives.

An interesting finding emerging from the analysis of the narratives embedded within the conceptual metaphors and discussions within the focus group was the appreciation of the relevance and significance of the concept of storytelling by participants and stories told. Squire et al. (2014), in citing how "your stories about your life are not the same as the life you live" (p. 110), stress the tensions between 'thinking about a story' (reflections of experiences) and 'thinking with a story', with regards to stories having a particular impact on practice (Bleakley, 2005). These distinctions between thinking about and with a story were borne out via focus group comments as follows:

"Aren't the stories we tell, our interpretation of an event, they not reality, are they?" (focus group comment)

"I heard you [the researcher] use the phrase, there is what happened, then there is the story of what happened, during our previous discussions. This is so true; we attach our interpretation to our stories which is probably so different from the reality of what happened" (focus group comment)

"I really like this concept of exploring the meaning of stories we tell compared to what happened, both in terms of our self-awareness and reflective practice. I think there is a lot of mileage in this" (focus group comment)

The comments above tap into the notion and help to signify participants' awareness of how stories and narratives by their nature are multi-dimensional and impacted by perception, interpretation and context of experiences, as are conceptual metaphors. Hence, they present difficulties in gaining a precise insight into the lives of individuals. Moen (2006), in citing Jerome Bruner, recognised this but sought to distinguish between narratives of 'life lived', 'life experienced' and 'life told', or the 'facts', 'fallacies' and the 'fiction' of a story (Moen, 2006). These aspects linked to the analysis of narratives require closer scrutiny of the multi-layered meaning and interpretation of narratives, specifically within the context of this study the relevance of metaphors referred to within the stories (Riessman, 1993, 2008; Squire et al., 2014).

Participants acknowledged the multi-layered meaning their reflections and stories had on their narratives of their teaching practice, and acknowledged the potential impact of conceptual metaphors used to reflect on and frame their experiences of teaching via focus group comments, stating:

"Didn't appreciate how often we talk in metaphors. They appear to be deeply ingrained in our thoughts" (focus group comment)

## *"I never really thought before of metaphors as being significant. Just a way of expressing yourself" (focus group comment)*

The above comments and further discussions within the focus group, suggest some consensus amongst participants of the significance in the pervasive qualities of conceptual metaphors mirroring and framing their reflections, narratives and approaches to teaching, and thus suggest how the two key conceptual metaphors identified framed narratives and impact actions. This is in keeping with Lakoff and Johnson's (1980; 2003) theory of conceptual metaphors being symbolic and representative of encoding a person's conceptualisation of experiences. However, metaphors and the meaning they convey have not traditionally been linked to narratives, with Dauite (2013) explicitly not listing or considering metaphors as an evaluative device or phrase imbuing meaning, or referring to a way of knowing or being. However, the findings of this study and exploration into the relevance and significance of conceptual metaphors, suggest metaphors could be considered evaluative language in determining both the symbolic meaning of the metaphor and offering insights into how experiences are perceived and conceptualised by individuals. Thus, concerning the findings of this study, the conceptual metaphors of 'teaching is a journey' and 'teaching is a performance' can be explored to identify the schemas and narratives participants may consciously or unconsciously associate with their teaching practice.

In interpreting the meaning, wider implications and significance of the two primary conceptual metaphors emerging from the findings, the concept of exploring 'frames' surrounding and linked to the conceptual metaphor helped establish a starting point for interpreting the conceptual metaphors and narratives they represent. As previously discussed within the Literature Review chapter, Goffman (1974), an early pioneer of the theory of framing, described how the act of placing frames around concepts and arguments consciously and unconsciously layers a schema over the way experiences are understood, interpreted and communicated (Borah, 2011). The participants within the focus group further acknowledged how reference to specific conceptual metaphors and narratives could positively frame an experience and further articulated this through the following focus group comments:

"I get it; thinking about my teaching as a factory will cause me to interpret and frame my teaching experiences differently instead of thinking of my teaching as a playground?" (focus group comment)

"Until now, I never really appreciate the power of metaphors. This whole thing about metaphors mirroring our thoughts and framing how we perceive things, I'm going to have to go away and think about more" (focus group comment)

Therefore, an exploration of metaphors whereby the notion of conceptual metaphors describing 'one thing in terms of another' (Lakoff & Johnson, 2003, p. 5) and giving preferences to one schema over another, with an emphasis on how the chosen frame and schema having an impact on academics' teaching practice, resonated with focus group participants. This notion of conceptual metaphors giving preference of one schema over other ties with Dauite's (2013) theory of dynamic narrating with regards to how a story's expression can give significant clues to the narrative underpinning the story, specifically by focusing on evaluative devices (phrases, expression, etc.) used to convey a story. Therefore, the research findings, in identifying the conceptual metaphors of 'teaching is a journey' and 'teaching is a performance' as being representative of how participants made sense of their teaching, give clues regarding inherent narratives expressing the values and beliefs participants hold about aspects of their teaching practice (Connelly & Clandinin, 1988). Furthermore, Robins and Mayer (2000) note how the framing process is ingrained into participants' thoughts and regulate behaviour (Moser, 2000), yet often out of conscious awareness of the individual (Borah, 2011), as shown by focus group comments identifying a lack of appreciation of the value and impact of exploring metaphors within the comments above. Thus, by adopting a specific conceptual metaphor, certain aspects of teaching are given preference over others. For example, by adopting and embracing working to the 'teaching is a journey' conceptual metaphor, academics may emphasise lesson preparation and student experience in preference over other parts of teaching, such as curriculum planning and meeting quality assurance procedures. This concept of conceptual metaphors and the narratives that underpin them, framing thoughts and actions, has been alluded to by Thibodeau and Boroditsky (2011, 2013, 2015) and Steen et al. (2014). Their studies suggest metaphors can frame perception and subsequent thoughts of a phenomenon, which concurs with how adopting 'teaching is a journey' and 'teaching is a performance' conceptual metaphors can frame academics' narratives of their teaching practice and influence their approaches to their teaching. Therefore, the use of a specific metaphor and the source target domains used to convey a metaphor, identify both how individual thoughts and behaviours could be influenced by the way the metaphor is framed and, by default, highlights specific elements of a concept or argument over others (Robins & Mayer, 2000; Borah, 2011).

Furthermore, in considering stories as reflections of how an academic conceptualises their experiences of teaching, Yero (2010), citing the words of Carl Jung, explains: "if teachers don't

know who they are – if they are unaware of their beliefs about learning, teaching ... then they are also unaware of whom they are teaching" (p. 14). Within the context of this study, identifying the frames surrounding the conceptual metaphors helps to raise awareness of the subtle ways the conceptual metaphors aid in contextualising and conceptualising experiences (Allbritton et al., 1995) and, thus, suggesting that there is a tendency for individuals to make decisions and take actions consistent with the metaphors used to frame their experiences (Robins & Mayer, 2000; Thibodeau & Boroditsky, 2011, 2013, 2015).

Digging deeper, these observations are in keeping with Chong and Druckman's (2007) suggestion of how individuals are likely to favour frames consistent with their values and beliefs. Thus, recognising how 'teaching is a journey' and 'teaching is a performance' conceptual metaphors are symbolic and representative of one's teaching practice reinforces Gillis and Johnson's (2002) argument that:

"Because they [metaphors] reveal our educational values, beliefs, and principles, they contain information essential to our growth as professionals" (p. 37).

Values, as expressions of personal beliefs, can be defined as "judgements and evaluations that we make about ourselves, about others and the world around us" (Yero, 2010, p. 28) and help give a person a sense of stability. As such, Mahlios et al. (2010) and Yero (2010) specifically identify how a person's beliefs impact the metaphors they choose and use to describe their experiences, which aligns with Lakoff and Johnson's (1980; 2003) theory of conceptual metaphors. However, as beliefs are chosen from a range of differing judgements and evaluations, the findings relating to values espousing 'self-direction', 'stimulation' and 'achievement', identify values underpinning practice, which Dilts (2006) defines as "have intrinsic worth [to a person]" (Yero, 2010 p. 34) and reflect a person's beliefs. Hence, within this study's context, the detailed narrative choice was made whereby participants veered towards using metaphors congruent with their values and beliefs, and further reinforced the significance of the narrative strand emerging and identified from the earlier finding.

Despite the yields to be gained from analysing conceptual metaphors to raise awareness of the narratives that underpin them, and inform thoughts and behaviours, caution needs to be taken in interpreting the stories told by individuals. This is especially regarding how the phenomenon of telling a story is more akin to the way in which individual reflections are a 'representative construction' of an actual event with regards to how they have consciously and/or unconsciously emphasised, generalised and distorted aspects of their story (Bold, 2012). Squire et al. (2014) further emphasise the uncertainty presented by the subjectivity inherent in the telling and interpretation of a story, and within the context of this study, metaphors, by stating how "there are blurred lines between what is said and unsaid, what is heard and not heard, what is analysed and not analysed" (p. 99); blurred lines requiring the need to corroborate findings of the interpretation of metaphors where possible. Hence, as part of this study, a reflexive approach to analysing the interview transcripts was taken by forwarding copies of the interview transcripts to each of the participants interviewed and inviting them to discuss the initial findings of the conceptual metaphors emerging out of the interview transcripts to a focus group to corroborate the findings into conceptual metaphors and the extent to which they give insight into the storied lives of their teaching experiences. This combined analysis of the findings revealed a degree of consensus amongst participants about the two primary conceptual metaphors identified and their subsequent narratives as being symbolic in giving a perceptive and accurate insight into the storied lives of their teaching experiences. This is evident via the concluding focus group comments, emphasising:

"I must admit to being surprised how we move from discussing stories, reflection, metaphors and narratives, but I can see the link between them now" (focus group comment)

"I like the two main metaphors we have talked about (i.e., teaching is a journey, teaching is performing). They sum up my approach to teaching, I think. But I especially like the concept of how they link to personal narratives and values. I'd like to read your completed thesis after it's finished" (focus group comment)

"The fact that I could follow the logic between all the different elements of what you researched (i.e., stories, reflections, metaphors, narratives, etc.) suggests to me there is something of real value here, and I'd like to continue this conversation with you further on another day" (focus group comment)

"A lot to take in here. I need to go away and think about this further. Like the others, I would agree with what your research shows and could identify with a lot of what we discussed today" (focus group comment)

Within a broader context, the patterns emerging from the analysis of the conceptual metaphors of 'teaching is a journey' and 'teaching is a performance' identified findings suggesting participants shared a common mindset regarding the narrative they adopted when reflecting on their teaching experiences. Dauite (2013), in suggesting how individuals "become good at narrating within cultural norms" (p. 6) to tell their stories and situate and conceptualise their experiences, within the context of this study, alludes to a commonly shared narrative amongst participants whereby the conscious and unconscious focus on reflective practice was both a product of, and process arising out of, a culture valuing and embracing the concept of continuing professional development in a bid to enhance teaching practice. Hence, it can be surmised how the coding analysis of the interview transcripts, of participant observations within the focus group and analysis of conceptual metaphors, against three levels of narrative frames, combined to identify a narrative steeped and embedded in a culture of reflective practice, adopting a solutions-focused approach to teaching, and underpinned by a keen sense of values espousing an attitude of 'self-enhancement' and 'openness to change' (Schwartz, 2012).

#### Chapter Summary

In summary, this chapter identifies how a degree of consensus was evident in acknowledging how the choice of conceptual metaphor referred to in reflecting on teaching experiences could impact teaching practice, with participants identifying how the two primary conceptual metaphors (i.e., teaching is a journey and teaching is a performance) identified within the study were symbolic of how participants contextualised and conceptualised their teaching practice and were representative of their conceptual teaching systems. These findings suggest how a greater level of reflection was possible by probing the significance of conceptual metaphors as an alternative to traditional cyclical models of reflection (e.g., Kolb, 1984; Gibbs, 1988).

From a theoretical perspective, the findings support notions of how conceptual metaphors can frame reflections and give insights into patterns of how a tendency exists for individuals to make decisions and take actions aligning with and consistent with their conceptual metaphors. Furthermore, the findings also aligned with Lakoff and Johnson's (1980; 2003) theory of conceptual metaphors, and McAdams and McLean's (2013) Narrative Identity theory regarding how the stories individuals tell of their experiences have relevance and inform their actions and behaviours. Regarding McAdams and McLean's (2013) Narrative Identity theory, constructs categories referring to exploratory narrative processing, agency and redemptive stories, emerged from the findings as being prevalent in being representative of participants' reflective stories told of their teaching experiences. These findings signify the emergence of a narrative strand whereby participants adopted an optimistic problem-solving approach to their teaching practice, in keeping with the further analysis of the interview transcripts against Tompkins and Lawley's (2006) PRO model.

The deeper level of analysis of the conceptual metaphors referred to by participants against Schwartz's (2012) Universal Values list identified values espousing self-direction, stimulation and achievements as being dominant and representative of participants' underpinning values and beliefs of their teaching practice acting as internalised guides that define and inform behaviours.

The combined analysis of the dominant conceptual metaphors against McAdams and McLean's (2013) Narrative Constructs and Schwartz's (2012) Universal Values list suggest how the exploration of metaphors, whereby the notion of conceptual metaphors with an emphasis on how the chosen frame can give insights to inform and impact on academics' teaching practice, emerged as key findings of this study.

Finally, this chapter identified how an appreciation existed in valuing the relevance and significance as an approach to reflective practice in a move away from the traditional approach to reflection encouraged by cyclical models of reflection.

### Chapter 6 – Analysis of Use of the Clean Language Questions Within the Research Interviews

This chapter reports on the findings into the use of the Clean Language questions to allow participants to reflect on their teaching practice in keeping with the third research question: "To what extent do the Clean Language questions allow participants to reflect on the metaphors representative of their teaching experiences?"

As previously discussed within the Methodology chapter (sections 3.1.2 and 3.3), three approaches of analysis were employed by this study to determine the level at which the Clean Language questions within the research interviews were facilitated and allowed participants to reflect on their teaching practice. These three approaches consisted of:

- A coded content analysis of the Clean Language questions asked by the interviewers within the research interviews
- Quantitative analysis of questionnaires distributed to participants after the interviews to seek participants' views of the use of Clean Language questions as a method of exploring participants' reflections and metaphors of their teaching practice
- Qualitative reflections of the use of the Clean Language questions within the questionnaire and focus group discussion to corroborate the participants' observations and reflections of the value of exploring metaphors and use of the Clean Language questions

This section reports on the analysis of the use of the Clean Language questions via coding decisions reported in section 3.3; namely, the categories of the type of questions asked, the level of Clean Language questions asked by the research interviewers, against Sullivan and Rees' (2008) list of Clean Language questions, and the identification of the 'cleanness' of Clean Language questions in keeping with Lawley's (2014) *Protocol for Validating 'Cleanness' of an Interview* classifications. The coding of the interviews was completed via NVivo data analysis software, whereby a total of 336 Clean Language questions were identified across 11 research interviews, against the Sullivan and Rees (2008) list of Clean Language questions included in Appendix one.

The findings reported within the chapter are based on the return of 10 out of 11 questionnaires and seven participants contributing to the focus group discussions.

The chapter initially details the findings of the content analysis of the use of the Clean Language questions within section 6.1 and proceeds to report on participants' reflections of the use of the Clean Language questions within the research interviews (section 6.2). Finally, section 6.3 summarises the key findings into using the Clean Language questions as part of this study.

#### 6.1 Analysis into the Use of Clean Language Questions

Initially, the value and relevance of the questions asked during the research interview were probed within the focus group, where participants commented:

"The questions seem neutral in their response, and the manner in which they were asked gave me scope to respond to them as I wish" (focus group comment)

"Questions asked by [the interviewer] appear vague, sometimes just repeating what I had said, but now understand how this is part of the model and in a way gave me the freedom to direct my answers to the question" (focus group comment)

"A strange interview in that I did not know what was expected of me or the answers I was required to give as in traditional research interviews. In a way, this was good ... I could reflect and say what I wanted to at will" (focus group comment)

These comments were further substantiated by questionnaire findings whereby eight out of ten participants reported they found the questions asked during the research interviews 'considerably' or 'to most extents' useful. Furthermore, figure 3 reports on the findings of participants being invited to describe the experiences of being interviewed and details how the combined findings from the questionnaire and focus group discussions suggest that the interview and interviewers' use of the questions were positively welcomed by participants in enabling them to reflect on and discuss aspects of their teaching practice.

A detailed analysis of the specific questions asked during the research interviews identified that a over two thirds of questions asked by the interviewers being either a Clean Language question, or reflective prompts question, and to a lesser degree probing questions inviting participants to volunteer further information (see appendix 14 for further details). These findings suggest the presence of a mix of 'clean' and other 'response' questions identified a divergence away from the pure form of Clean Language interviewing advocated by David

Grove, whose concept of the clean interview comprised only of solitary use of Clean Language questions and the use of Clean Language 'syntax'. This move towards using a more varied range of questions could be explained by recognising that David Grove's background in using the Clean Language model within a psychotherapy setting differs from the aims of the reflective or research-based context of the interviews completed as part of this study. This importance in acknowledging the context and aims of the interview is further recognised by Meese (2014), who identified how the aim of an interview could have a bearing on the use of the Clean Language questions to promote a more pragmatic approach to employing the use of Clean Language questions within research contexts and within this study.

An exploration of the interviewers' specific use of the Clean Language questions was made against the 12 Clean Language questions (see Appendix one) identified by Sullivan and Rees (2008), and coded to determine the frequency of their use within the research interviews and identify the extent to which interviewers asked specific Clean Language questions. Of the three categories of Clean Language questions listed by Sullivan and Rees (2008), 'developing', 'intention' and 'source and sequence' questions from the total of 134 Clean Language questions were asked across all research interviews. These findings signify a clear pattern in the use of the 'developing' category of Clean Language questions. Therefore, a clear trend in the prevalent use of the 'developing' category of Clean Language questions is evident from the findings. The relevance of the findings in identifying the frequency and popularity of the use of four key 'developing' Clean Language questions is significant regarding the level of competence needed to employ and ask these questions effectively to reflect on practice, due to the ease with which the interviewers can recall and ask the developing questions, within a research interview context or facilitate reflection.

A more detailed analysis of the use of the 'developing' Clean Language questions shows a pattern whereby four key 'developing' Clean Language questions were predominantly referred to within the research interviews, namely:

- What kind of [x]?
- Anything else about [x]?
- Where is/whereabouts is [x]?
- That is like what?

The above listed developing questions represented 92.24% of all 'developing' Clean Language questions asked within the research interviews. These findings concur with Rees (2016) and Hartley's (2020) observations as sufficient in employing the use of the Clean Language question, with 92.24% of all questions asked originating from the 'developing' category of Clean Language questions. Thus, the notion of 'less is more' could be applied to stress how using a minimum range of clean questions outside of a therapeutic context could be employed effectively to encourage individuals to reflect on their practice.

38.69% of the questions asked across all the research interviews were 'reflective prompts' of participants' utterances, whereby the interviewer responded by mirroring participants' words/phrases back to them to either promote further discussions or clarify/explore the issues being discussed. This was a manner of responding which participants found helpful via focus group comments, stating:

"Enjoyed the way in which my thoughts were repeated back to me. Made me realise what I was thinking"

"Interesting how some of the questions asked were not questions, just a reflection of my own words. Don't get me wrong, this did not seem out of place. Actually helpful"

The act of the interviewer responding to participants' comments by mirroring back participants' own words and phrases can also be found within the humanistic and Rogerian approaches to helping, and thus suggests how the use of the Clean Language questions can be complemented and enhanced by reference to other models of helping. Regarding the use of the humanistic and Rogerian approaches to helping specifically, Nelson-Jones (2014) identifies how the act of reflecting keywords and phrases back to the client can help pull together the key contents and feelings expressed by a participant, and can help to show understanding on the part of the interviewer, as well as clarifying the most important points and acting as a bridge in allowing the participants to expand on the issues being discussed, or move on to another. Mearns et al. (2013) further stress how reflecting on participants' sentiments, particularly feelings, aids in building empathy with the client and allows for a deeper and more meaningful exploration of the issues by a participant. However, the act and importance of reflecting and summarising as a mode of responding to participants is underplayed within the Clean Language model, in preference to using the Clean Language 'syntax' to explore metaphors, yet is considered by participants to be of value.

The second level of coding undertaken in the analysis of the interview questions consisted of examining the levels of 'cleanness' of each of the Clean Language questions asked by interviewers against Lawley's (2014) protocol to further evaluate the use of the Clean Language questions (see Appendix 11 for further details. Table 5 details the coding percentage analysis of the Clean Language questions asked by the research interviewers. In scrutinising the use of the Clean Language questions, Lawley's (2014) contention of 90% of a range of questions asked within an interview needing to be 'classically clean' or 'contextually clean' as a measure of the validity of the cleanness of the interview was adhered to, with 92.75% of the questions asked being within the measure stipulated by Lawley (2014). However, the 90% measure of cleanness of interview questions identified by Lawley (2014) illustrates a personal preference for measuring cleanness of Clean Language questions asked, as opposed to one corroborated by research. Though, the high degree of clean questions asked within the 'classically clean' and 'contextually clean' categories aids in stressing the degree of skill and competence in the use of the Clean Language questions that was evident when exercised by the two research interviewers.

The *Protocol for Validating a 'Cleanness' of an Interview* represents an attempt by Lawley (2014) to establish a range of criteria to measure the cleanness of the interviews and validate the Clean Language model's use via academic research studies. The use of the categories proposed by Lawley (2014) allows for clear identification and distinction between the 'cleanness' of Clean Language questions and corroboration in supporting the validity of the protocol, while further consolidating the legitimacy of using the Clean Language model. However, limits exist in the use of Lawley's (2014) protocol criteria since it does not acknowledge the value of questions asked outside of the Clean Language model. In this study, value was placed on how Clean Language questions were complemented by the use of reflective prompts and probing questions during the research interviews, as discussed earlier.

#### 6.2 Analysis of Interviewers' Conduct in the Use of the Clean Language Questions

Analysis of the questionnaire findings suggests participants positively welcomed the interviewers' conduct and enjoyed the experience of being interviewed, as shown by the words chosen to describe the experience of the interview in figure 3 within section 6.1.

These questionnaire findings into the conduct of the research interview were further supported by questionnaire and focus group comments stressing how:

"[interviewer] stuck to the pointing and picked up the metaphors to explore... appeared detached, but still able to precisely pick up on what I was saying especially in relation to metaphors" (focus group comment)

"More conversational and cathartic in his use of questions. Almost a self-help approach to reflections" (focus group comment)

"It was a really well-conducted interview. The interviewer is clearly skilled in the Clean Language method and was also very good at quickly building rapport" [Participant 2]

"It took a little bit of time for the interviewer to coax out specific metaphors. However, she spent time re-questioning and coaxing ... then after this I found the experience useful for reflecting on my practice" [Participant 6]

The interviewers' skills and experience in using the Clean Language questions can influence the number and the 'cleanness' of Clean Language questions asked and engagement from participants. Efficient use of the Clean Language model requires a degree of skills and competence as demonstrated by the range of online and offline courses existing to teach the Clean Language model, and local practice groups to assist Clean Language practitioners and those with interest in Clean Language to fine-tune and hone their Clean Language questioning skills. Both the interviewers involved in interviewing participants had previously completed a minimum of Modules 1 and 2 of the Clean Language model.

Nevertheless, these positive findings into the use of the Clean Langauge questions need to be tempered by acknowledging how some participants expressed mild to moderate unease in the interviewers' mode of questions and questioning approach; an issue reported by a couple of the participants who, on reflecting on their experiences of being interviewed, stated:

*"Unsure of direction, but questions required real thought, which was hard work at the time" (focus group comment)* 

"It was a deep interview, sometimes not easy, but ultimately productive in getting me to think about my practice" (focus group comment)

"Found myself working hard to think my responses through, not in a negative way but the questions asked focused my attention in a way that I am not used to" (focus group comment)

Possible explanations of the contrasting views expressed by participants on being interviewed could be linked to a lack of preparation and clarification of the Clean Language model on behalf of the participants. Each participant completed a Participant Consent Form (see Appendix two) before the interview, detailing how the focus of the interview required them to reflect on their practice, and some participants were only given a brief explanation of the Clean Language model or approach to questioning as advocated by the model. This issue was evident whereby a couple of participants reported how they were "unsure of expectations [of the approach to interview,", and of how the interview dynamic functioned especially since, where the onus was on them to lead the conversation, was at times "uncomfortable", thus denoting how a lack of understanding of the Clean Language questions could hinder the effective use of the model and might have been addressed by a more thorough debrief during the Participant Consent Form of the Clean Language model and an approach to questioning employed within the interviews. However, it is important to acknowledge how the role of the interviewer and the level of rapport they had with participants were found to be key variables aiding the success of a Clean Language interview, as does the perception of the level of the difficulty some participants reported in responding to the Clean Language questions; issues that will need to be considered and debated in the broader use of the range of Clean Language questions in the discussion that follows.

The controversy over the 'mechanical' use of the Clean Language syntax, previously discussed within section 2.4, was analysed to determine the use of the syntax within the research interviews. While a lack of consensus exists into the use of the 'syntax' in the Clean Language community, the debates within online professional media forums lean towards a more flexible and pragmatic approach to the use of the syntax being encouraged to match the aims of the interview (Lawley, 2014; Meese, 2014). This decision to move away from employing the use of the Clean Language syntax within the research interview, while remaining mindful of the need to be 'clean' in the way questions are asked to maintain rapport with the participant, was reinforced via focus group discussions which suggested:

"I found the interviewer's approach to asking questions relaxed and casual. I don't know the Clean Language questions, but my chat with the interviewer appeared very informal" (focus group comment)

"Following my interview, I was interested in finding out more about Clean Language, and I have read some online articles about it also watched some YouTube videos. From what I know, the interviewers didn't precisely follow how Clean Language is advocated, but this is a positive. I think I would have found the format I have read about somewhat intimidating" (focus group comment)

"[Interviewer 1] was more conversational and cathartic in his use of questions. Almost a self-help approach to reflections" (focus group comment)

*"[Interviewer 2] came across as a warm person and professional in her approach" (focus group comment)* 

The above focus group comments, echoed by other participants, about the interviewers' approach, identify how the combination of the move away from Clean Language syntax and use of additional reflective summaries and probing questions, aided a more flexible approach to employing the Clean Language questions, and encouraged participants to engage productively with the research interview and the Clean Language questions asked of the participants. Furthermore, the more conversational style of questioning and the Clean Language model adopted by interviewers and welcomed by participants reinforced the need to promote conversational reflection methods advocated by Hoffman-Kipp et al. (2003) and Guo (2021).

#### 6.3 Analysis into Clean Language Questions' Ability to Promote Reflective Practice

In further probing how participants found the range of questions asked allowed them to reflect on their teaching practice, questions two, eight, nine and ten within the questionnaire (see Appendix 11) aimed to determine participants' views of the range of questions asked by the interviewers. The majority of the participants within the questionnaire reported positively on the range of questions asked during the interview as either 'considerably' or 'to some extent' helpful in encouraging and enabling them to reflect on their teaching practice. Findings correlating with qualitative comments within the questionnaire and focus group discussions suggest:

"The questions were sufficiently open to give me plenty of scope to explore my practice; follow-up questions pointed me in the right direction to explore further" (focus group comment)

"[the questions] Helped me think quite deeply about my teaching practice and approach" (focus group comment)

"The style of interviewing was open and non-judgemental. I was therefore enabled by the open nature of the questions to reflect on my practice" (focus group comment) As discussed within section 6.1 earlier, the questionnaire identified a strong leaning towards identifying positive elements of the interview as acknowledged via question eight, whereby from a list of 18 supportive and undesirable words aimed at gauging participants' broad reflections of the interview, the words 'informative', 'revealing', 'useful' and 'helpful' proved popular in describing the participants' experience of being interviewed as part of this study; a point further reinforced by question ten, where participants appear to be generally 'satisfied' or 'very satisfied' with the range of Clean Language questions asked, and further supported by the findings of question nine, where participants reported they found the range of questions during the research interview helpful via comments such as:

"The questions provided an opportunity to reflect on aspects of my practice, I do not often think about in-depth though being too involved in the everyday minutiae of my role" (focus group comment)

"Gave me time to think about my practice and spent some time reflecting after the interview" (focus group comment)

These positive reflections of the interview identify a common theme whereby a degree of agreement exists regarding the value of Clean Language questions in enabling participants to reflect on their teaching practice was further reinforced by the findings of questions one, three and five of the post-interview questionnaires completed by participants. The findings from question one of the questionnaire showed participants identified how the interview questions allowed them to 'considerably' or 'to most extents' explore aspects of their teaching practice, and 70% reported how the interviews allowed participants to explore their thoughts about their teaching practice (question three). This is a significant finding supported and congruent with focus group comments whereby participants recounted:

"Yes, a very deep level of reflections, which I was unprepared for or expected as part of the interview" (focus group comment) "A very different form of reflection" (focus group comment) "Required me to reflect at a level I had not done before" (focus group comment)

In further analysis of the use of the Clean Language questions, participants, responding to question five, determined how a sense of development was evident in their reflections of the interview, with 80% of participants reporting how the research interviews helped raise their self-awareness of their teaching practice. These findings suggest a pattern of acceptance and relevance of the approach to the interview, and the use of the Clean Language questions as a reflection method. The combined findings from both questionnaire and focus group

discussion identified how, as a mode of reflection, the Clean Language questions allowed participants to explore, analyse and challenge their appreciative systems (Schön, 1987; Zeichner & Liston, 2014), and via an exploration of the frames surrounding the metaphors referred to when reflecting on their teaching practice, identify a potential alternative approach to reflecting on practice.

These findings further suggest how conversing and interaction (with the interviewee) helped address some of the criticisms levelled at contemporary approaches to reflective practice, for example, the value of dialogue as a mode of reflection advocated by Hoffman-Kipp et al. (2003) allowed participants to move away from the use of checklist approaches to reflection as identified by Boud and Walker (1998), and criticism of the dominant use of process and cyclical models of reflection used within different aspects of education today. Additionally, Surgenor's (2011) criticisms of how a premise to 'reflect-on-demand' prompted by cyclical approaches to reflection is at odds with and could prove counterproductive in allowing individuals to reflect on their practice effectively are countered by the informal conversational approach to asking questions utilised as a part of this study.

The conversational approach to reflective practice advocated by the Clean Language questions also ties into Black and Halliwell's (2000) and Bullman and Schutz's (2008) notions of how reflective practice approaches need to take into account the different ways individuals learn by giving participants autonomy in steering discussions to give them greater scope to think through and explore the metaphors symbolic of their teaching practice. Hence, it is important to consider how the Clean Language questions' productive use is aided by the rapport between the interviewer and interviewee, and the practical and pragmatic approach of asking Clean Language questions in an informal and conversational way.

Participants' observations of experiencing the Clean Language questioning model as a method of reflection suggest a degree of acceptance in using the model as a reflective approach which allowed them to explore how their thoughts were impacted by the act of placing frames around concepts and arguments consciously and unconsciously, layers a schema over the way experiences are perceived. The frames and schemas analysed and identified as part of this study are discussed further in sections 5.1 and 5.2.

The extent to which Clean Language questions allowed participants to explore metaphors associated with their teaching practice and emotions was queried as part of questions four, six and seven on the questionnaire, and further probed as part of the focus group discussions.

More specifically, the value of reflecting on metaphors allied to participants' teaching practice was evident as part of the analysis of question six, with analysis of the questionnaire responses showing 80% of the participants asserting they found the focus of reflecting on metaphors to be 'considerably' or 'to some extent' useful as a way of reflecting on their practice. These findings are corroborated via focus group comments stressing:

"Never before explored metaphors and viewed them as important. But do so now. Not sure how but feel that they are significant and useful" (focus group comment)

"Made me think about metaphors which were not really metaphors" (focus group comment)

The significance of exploring metaphors was also revealed by the findings of question seven, with 80% of the participants reporting they found the specific focus on analysing metaphors to be either 'considerably' or 'to some extent', helpful despite the element of surprise reported by the focus group:

"Didn't appreciate how often we talk in metaphors" (focus group comment)

"Never really thought before of metaphors as being significant. Just a way of expressing yourself" (focus group comment)

"I would need to listen and read through the interview transcript again carefully, but my general impression was the issues discussed and metaphors explored were significant, but just don't know or able to summarise how specifically" (focus group comment)

Interestingly, the consensus of the value of exploring metaphors to reflect on practice did not extend to allowing participants to explore their emotions. Responses to question four aimed at establishing to what extent the interview allowed participants to explore their feelings resulted in 70% of the participants identifying how the Clean Language questions either 'considerably' or to 'some extent' allowed participants to explore their emotions. Findings, however, suggesting a lesser degree of certainty as to the value of the Clean Language questions in tapping into participants' emotions, were demonstrated by focus group comments which reflected:

"I would be curious to know about what the metaphors I referred to say about me" (focus group comment)

"Felt the interview was deep, and so were the metaphors which [interviewer] picked up to explore, but not sure what all this means,

## what is the significance. Maybe a follow-up interview would be helpful?" (focus group comment)

The combined analysis of the questions aimed at determining the significance of the Clean Language questions in allowing participants to explore metaphors representative of their teaching practice identified a degree of consensus exists in the positive value of exploring metaphors as a way of reflecting on experiences and via the use of the Clean Language questions.

In summary, common themes and threads emerging from the analysis of the post-interview questionnaires and focus groups, with regards to determining "how the Clean Language model and range of questions allowed participants to reflect on their teaching practice", suggest a strong correlation whereby participants acknowledged and appreciated the value of the use of Clean Language questions as helpful prompts that aid reflection and the specific focus on exploration of metaphors to be significant. The role of the interviewer and mode of questioning also appeared significant and were issues of note in the analysis of participants' experiences of completing the research interviews.

#### Chapter Summary

As part of the findings into the level to which the Clean Language questions facilitated and allowed participants to reflect on their teaching practice, three approaches of analysis were employed by this study: a coded analysis of the Clean Language questions asked by the interviewers; quantitative analysis of questionnaires distributed to participants after the interviewes; and qualitative reflections of the use of the Clean Language questions within the focus group discussion. The combined findings identified a favourable response towards the use of the Clean Language questions in order to allow participants to reflect on and raise their self-awareness of their teaching practice. A mode of reflection, when combined with interviewers' use of reflective prompts and probing questions, was specifically aided by the use of the conversational mode of questioning to encourage reflections over traditional self-reflect activities more routinely utilised and used within various aspects of education today (Guo, 2021), and this was further impacted by the level of rapport between the participant and interviewer. Participants also welcomed the interview's focus on exploring metaphors as a potentially valuable approach to reflecting on practice.

A measure of the use of the Clean Language questions was shown via an analysis of the questions asked against Lawley's (2014) *Protocol for Validating a 'Cleanness' of an Interview*,

which, in addition to proving to be influential in determining the level of 'cleanness' of interviews, aided in identifying the high degree of Clean Language questions asked and exceeded the 90% measure stipulated by Lawley (2014) of cleanness of interviews. Variations in the specific use of the Clean Language model, with the addition of reflective prompts and probing questions, was evident in the analysis of the finding showing how the proper form use of the Clean Language model, as advocated by its founder, David Grove, was at odds with the depth the model used by a research interview, which consisted of a mix of reflective and clean questions and, therefore, moved away from the use of the 'clean syntax'. The mode and method of use of the Clean Language questions were welcomed by participants in identifying how a more relaxed and flexible use of the Clean Language questions and enhanced depth of rapport with the interviewer allowed for a more thorough and insightful level of reflections.

An analysis of the interviewers' use of Clean Language questions identified a leaning towards using a limited range of four 'key developing questions' which utilised a range of reflective prompts. This suggests how the effective use of the Clean Language model could be achieved by focusing on a narrow set of Clean Language questions that individuals could be trained to employ effectively.

However, these findings are hinged with some uncertainty expressed about the style of questioning employed by interviewers and how the contrasting style of questioning could influence the level of engagement of the participant and their ability to reflect on their practice. This was a finding, however, which could possibly be accommodated by a more thorough briefing and the introduction to the Clean Language model and questions before the interview.

### **Chapter 7 – Conclusion**

In acknowledging the importance of reflection within education, this study set out to determine the significance of conceptual metaphors in aiding reflective practice, the relationships between conceptual metaphors and academics' narratives for teaching, and to evaluate the Clean Language questioning model as a research interviewing method. The following chapter summarises the findings in light of the research questions (section 7.1) before proceeding to detail the study's contribution to knowledge (section 7.2) and then progressing to discuss the emergence of a working model of reflection based on the empirical research findings, and a discussion of how the model may work in practice (section 7.3). The chapter ends by acknowledging limitations and caveats associated with the study (section 7.4) before ending with a concluding summary.

#### 7.1 Review of the Research Questions

The section below reviews and discusses the research questions posed within the introduction chapter. The specific research questions were:

- 1. What are the dominant metaphors used in the way academics recall, reflect and describe their teaching experiences?
- 2. What are the relationships between academics' use of metaphors and their narratives of teaching?
- 3. To what extent do the Clean Language questions allow participants to reflect on the metaphors representative of their teaching experiences?

Regarding the first research question, which focused on the dominant metaphors referred to by academics to recall, reflect on and inform their teaching practice, this study identified three key findings from the analysis of the research interviews and focus group discussions. Firstly, the analysis of the metaphors identified the dominant use of three source domains from the range of source domains coded and analysed within the interview transcripts (see chapter 4). Namely, source domains referring to 'movement and directions', 'place and locations', and 'senses' were confirmed by participants as being representative of their teaching practice and indicative of the outcomes (primarily curriculum and local/national standards) that participants are required to maintain and achieve. Secondly, the mapping of source to target domain codes in relation to NLP logical learning levels identified target domains conveying characteristics of 'behaviour', 'environment' and 'emotions' was central to the participants' metaphors and were mentioned in preference to deeper learning levels linked to values, beliefs and identity. Thirdly, a sense of conventionality and universality in the use of the metaphors expressed, especially regarding the use of 'structural metaphors' to communicate aspects of behaviours and 'orientational metaphors' to express thoughts and emotions, was evident within the findings. The analysis and the mapping of the source to target domains were aided by and confirmed the relevance of Lakoff and Johnson's (1980; 2003) work on conceptual metaphors and Kövecses' (2010) list of source and target domains in analysing the findings, and offered potential to draw inferences with regards to the relevance of the conceptual metaphors referred to by participants and identified via the study. In summary, the findings into the dominant metaphors referred to by academics to reflect on aspects of their teaching practice identified an emerging trend towards 'movement and directions', 'place and locations', and 'senses' source domains to convey thoughts associated with 'behavioural', 'environmental' and 'emotional' aspects of their teaching practice.

The research findings determined the applicability of focusing on metaphors as meaningful variables to explore as part of the reflective process. They can raise awareness and give insights into how academics contextualise and conceptualise their teaching experiences by identifying and recognising how the dominant conceptual metaphors referred to by academics to reflect on teaching practice are symbolic and representative of the narratives underpinning their teaching practice. This suggests how, within a reflective discussion, there is potential to focus on and explore metaphors in ways that could be of value in enhancing the reflective process.

These findings also yielded some interesting observations regarding an emphasis on the role of metaphors and, more widely, storytelling within the context of reflective practice. Academics within HE are not obliged to reflect on their teaching practice for CPD purposes but are encouraged to ponder on their experiences to enhance their teaching practice. Mälkki and Lindblom-Ylänne (2012) cite how negative perceptions of what constitutes reflective practice and a lack of knowledge about teaching pedagogy (McAlpine & Weston, 2000) can hinder reflective practice within HE. However, adopting a conversational approach to reflecting on experiences, as utilised within this study, can help to facilitate a more detailed analysis of thoughts and feelings regarding aspects of teaching practice, and adopt a more amenable and flexible approach to reflective practice. This move from a descriptive to interactive approach to reflection fosters the opportunity to explore and critique practice in a way traditional cyclical models of reflections via ruminations may not (Ashwin & Boud, 2020; Guo, 2021).

Furthermore, regarding the focus on exploring metaphors within the context of reflective practice, Thibodeau and Boroditsky (2011; 2013; 2015) – as previously discussed within the literature review – suggest how framing and explaining experiences via specific metaphors impacts how phenomena are perceived. This was evident within the findings of this study and through the identification of academics' conceptual metaphors of their teaching. Therefore, adopting a storytelling approach to encourage reflective practice in order to explore how the stories and metaphors referred to within a story are framed, can help to raise awareness and give insights into the effectiveness of approaches to teaching, and may also help to address barriers to reflective practice identified by Mälkki and Lindblom-Ylänne (2012) and McAlpine and Weston (2000).

The findings, which indicate a preference towards appreciating the value of exploring metaphors by participants, could be widened and extended to a teaching context. The value of referring to metaphors as a strategy for teaching students has previously been acknowledged and espoused by Wormeli (2009), Saban (2006) and Karolia (2015). However, within a teaching context, considering and exploring the metaphors referred to by students to demonstrate their understanding of knowledge and of concepts covered during lectures and seminars appears to have not been given due regard. Attending to and questioning the metaphors referred to by students when explaining their understanding of content and concepts covered during sessions via the Clean Language questions (further discussed below) could help provide an additional avenue to clarify and verify students' understanding of concepts discussed during teaching sessions and facilitate reflection.

The second research question – "What are the relationships between academics' use of metaphors and their narratives of teaching?" – saw the emergence of two key conceptual metaphors, primarily, perceiving 'teaching is a journey' and, to a lesser extent, 'teaching is a performance'. Participants were able to recognise and concur with these conceptual metaphors as representative of their teaching experiences and yield broader insight into the problem-solving and decision-making strategies they adopt as part of their teaching practice. Further analysis of 'teaching is a journey' equated the conceptual metaphor as an excursion between two points to meet the students' needs, and address curriculum and assessment needs. Similarly, the 'teaching is a performance' conceptual metaphor related to 'not teaching well' and 'attempts to teach better' as themes within participants' perspectives of their teaching narratives. Participants acknowledged how these conceptual metaphors were symbolic and representative of how they contextualised and conceptualised their teaching practice. They

also identified how the choice of conceptual metaphors could potentially impact teaching practice, specifically regarding how individuals make choices to tell meaningful stories and potentially inform decisions and actions.

Exploring the relationship between conceptual metaphors and wider narratives was established by exploring the implications and significance of how the theory of frames can allow participants to understand the ways in which their conceptual metaphors for teaching can inform practice. The implications of the conceptual metaphors were understood by exploring those referred to by participants within the interview transcripts in relation to the use of the PRO framing categories (Tompkins & Lawley, 2006), and the wider frames and narrative of McAdams and McLean's (2013) Narrative Constructs and Schwartz's (2012) Universal Values Theory. Against McAdams and McLean's (2013) Narrative Constructs, the study identified a pattern where constructs linked to 'exploratory narrative processing', 'agency' and 'redemptive' stories were dominant within the conceptual metaphors mentioned by participants while reflecting on aspects of their teaching practice. This, thus, suggests that participants' narratives of their teaching consist of a quest to enhance their teaching practice and to seek the autonomy to adopt a solution-focused approach to teaching positively. These findings were further supported via analysis of the conceptual metaphors against Schwartz's (2012) Universal Values Theory, where a strong focus on values espousing self-direction, stimulation and achievement was evident. These values appear to be aligned with and be congruent with the findings of McAdams and McLean's (2013) Narrative Constructs in being consistent with expressing a narrative whereby participants' schemas of their teaching consist of a positive wish to reflect on their pedagogy and be open to change in adjusting their teaching practice where necessary. These findings also shed light on how theory could underpin and give insights into the reflective process and model of reflection proposed. Within the context of this study, the relevance of Lakoff and Johnson's (1980; 2003) theory of conceptual metaphors in the telling of stories (of teaching experiences) within the realm of narrative studies aided in reinforcing the exploration of metaphors within the reflective stories people tell.

The findings into interpreting and determining the narratives underpinning conceptual metaphors referred to by participants emphasised the value of linking conceptual metaphors and narratives to raising awareness of participants' agency and teaching identities. Ashwin and Boud (2020) specifically identify how telling stories and reflecting via discussions helps us understand how emotions, thoughts, and perceptions of the past and present, can enhance

teaching practice and awareness of who we are as teachers. This study helped raise awareness and give a deeper insight into the values and beliefs underpinning participants' teaching practice through the variables discussed earlier to code and analyse the interview transcripts. The importance of exploring academics' values and beliefs has been alluded to by MacFarlane (2004), who suggests how many aspects of education are imbued with values and the need for academics to strive to teach through and demonstrate the values they espouse. In referring to Schwartz's (2012) Universal Values list to analyse participants' conceptual metaphors, this study helps to stress an avenue of how values of beliefs underpinning academics' teaching practice could be explored.

However, caveats in analysing the links between conceptual metaphors and wider narratives need to be acknowledged. For example, the generic, simplified phrasing of a conceptual metaphor (e.g., teaching is a journey) may not highlight the comprehensive understanding and complexities of experiences or may change over time and not accurately convey the participants' intentions. Furthermore, individuals may consciously and unconsciously censor aspects of their stories, while the metaphors used to convey their experiences may not fully define the broader narrative they purport to represent. Therefore, any analysis and interpretation of a metaphor need to consider participants' specific understanding of the multi-layered meaning of the metaphors. Within the context of the study, using the Clean Language questioning model approach encouraged a more open participant-centred and driven dialogue in exploring the relevance and significance of a metaphor, and helped to focus the research interview discussions against the simplified analysis and interpretation of a metaphor.

The retrospective analysis of metaphors and qualitative data needs to be recognised regarding the relevance of past reflections of experiences and the findings to current practice; an aspect of reflection whereby participants are likely to make causal connections between events and variables in keeping with their need to make sense of phenomena; a need further extending to researchers who themselves are prone to creating narrative fallacies when analysing retrospective data and identifying causal connections in the analysis of raw data; a matter countered to an extent by allowing participants to review their interview transcripts and an invitation to discuss the significance and implications of the findings, individually and via the focus group, where participants expressed a link between their retrospective reflections and the metaphors referred to while discussing aspects of their teaching practice.
In recognising the possibilities of exploring conceptual metaphors and frames surrounding the relevance and significance of the metaphors, adopting an alternative approach to interpreting narrative may be helpful in further determining a more robust link between conceptual metaphors and the broader underplaying premise conveyed. Dauite's (2014) concept of narrative inquiry and model of 'significance analysis' process particularly appeal given the model's emphasis on determining the evaluative devices and exploring narrative from holistic perspectives, which may allow for an original and novel approach to interpreting the relevance of the conceptual metaphors.

The findings into the third research question into the use of the Clean Language model showed a favourable response towards the use of the Clean Language questions to explore metaphors into aspects of participants' teaching practice (research question three). Findings, confirmed by both the questionnaire and focus group comments, indicated that participants specifically noted that the Clean Language model allowed them to explore, analyse and challenge elements of their practice, and also raised their self-awareness by framing and reframing reflections of their teaching practice. A key finding emerging out of the coding analysis of the Clean Language questions was the prominent use of four specific 'developing' clean questions over 'intention' and 'sort and sequence' category questions. Namely, questions stating:

- What kind of (x)?
- Anything else about (x)?
- That is like what?
- Whereabouts/where is (x)?

In keeping with the findings above, the use of a limited number of Clean Language questions links to Rees' (2016) and Hartley's (2020) notions of the possibility of utilising a limited range of Clean Language questions to employ the Clean Language model effectively. The research implications of this have potential in the broader methodological use of the Clean Language model given the limited time needed to learn its fundamental principles and the four questions of the Clean Language model as a tool for encouraging reflection.

The study's evaluation into using the Clean Language questioning model to complete qualitative research interviews identified the potential value of using the model as a research method. Specifically, it identified how using a limited range of four Clean Language questions, together with reflective and probing questions, was especially productive in promoting a conversational approach to reflection. This further supports the extended use of the Clean

Language model in alternative contexts. Of particular note is the role of the interviewer and the mode of questioning adopted by the interviewers, whereby participants noted a level of rapport with the interviewer and observed how effectively the Clean Language questions avoided the interviewers appearing distant. The questioning approach employed by the interviewers, alongside the absence of the 'Clean Language syntax', adopted a more pragmatic approach to asking a mix of 'clean', and reflective and probing questions. The act of conversing and the use of dialogue also stimulated the reflective process. This was in contrast with reflecting on paper in isolation and the more dominant process and cyclical models of reflection typically used within different aspects of education today (Hoffman-Kipp et al., 2003; Zeichner & Liston, 2013).

The 'cleanness' of these research interviews, with regards to the range of questions asked by the interviewers, was confirmed by analysis of the interview transcripts against Lawley's (2014) *Protocol for Validating 'Cleanness' of an Interview* classifications and the list of Clean Language questions identified by Sullivan and Rees (2008). The research interviews completed as part of this study exceeded Lawley's (2014) premise of 90% of the range of questions asked in an interview needing to be 'classically clean' or 'contextually clean', thus, demonstrating the two researchers' confidence and competence in using the Clean Language model when completing the research interviews and acknowledging the value of Lawley's (2014) protocol as a measure of determining the depth of the Clean Language interview, as discussed in section 6.1.

A key secondary finding in using the questions asked during the research interviews centred on reflective summaries and probing questions. The use of reflective summary and probing questions within the research interviews was in keeping with the ethos of the Clean Language model, whereby the zero inference embedded within a reflective summary/paraphrase could be considered as a 'classically clean' question within Lawley's (2014) protocol criteria. This encouraged a balance between using four key clean questions (identified earlier) with a series of reflective paraphrases and probing questions to aid reflections and promote a conversational approach to the interviews in preference to the Clean Language syntax.

Despite acknowledging the value of the Clean Language questions in aiding reflections, a small number of participants expressed a degree of uncertainty about the questioning approach employed by interviewers, specifically, regarding the lack of direction from the interviewers and how the model requires a 'reflect-on-demand' approach to isolating and

exploring specific metaphors. On consideration, this issue may be linked to inadequate preparation and clarification of the use of the Clean Language model before the interview and was possibly impacted by some participants' learning and reflective preferences (Black Halliwell, 2000; Bulman & Schutz, 2006; Surgenor, 2011). These anxieties could have been potentially addressed and overcome by debriefing participants before the interview about the scope and manner of the questioning approach adopted using the Clean Language questions and model.

Given the above conclusions, in keeping with the finding welcoming the use of the Clean Language questioning model to aid reflections, a study trialling the emerging model of reflection would be beneficial to determine the value of the effectiveness of the model in practice and in order to expand the appeal and application of the Clean Language model. The findings into the effectiveness and popularity of using four key Clean Language questions, in conjunction with reflective paraphrase prompts, would benefit from being promoted and debated within the wider Clean Language community and research to determine the approaches' value within the context of completing 'clean' research/questioning interviews.

The development of a new model of reflection based on the research findings gives rise to a practical outcome of the study. This model of reflection, developed in line with the study's key findings, acknowledges the complexity and individual nature of the reflective process, adopting a model of reflection that encourages a conversational storytelling approach to reflection and allows for an exploration of metaphors and the use of the Clean Language questioning model.

#### 7.2 Contribution to Knowledge

In concluding this thesis, the key elements of this study – storytelling within reflective practice, metaphors, narratives and application of the Clean Language questions – discussed initially within the introduction chapter, are revisited to summarise the value of the study within various contexts.

The study investigated an approach to reflective practice previously not studied within the context of how HE academics could reflect on aspects of their teaching practice or how reflective stories and the metaphors referred to by academics link to how they might conceptualise their teaching experiences. The findings of this study illustrated how

conversational storytelling was valued and could be used as a method of reflection and a welcome alternative to the traditional cyclical process method of reflection.

The study's findings further noted the relevance of exploring metaphors as part of the reflective process, specifically, focusing on identifying the dominant conceptual metaphors referred to by academics to reflect on their teaching practice which had significance in terms of the narratives the metaphors represent within broader teaching and reflective practice contexts.

From a methodological viewpoint, the study's use of the Clean Language questioning model to complete qualitative research interviews was effective and productive when used as a research method in facilitating the reflective process and exploration of the conceptual metaphors, thus, recognising the positive potential use of the Clean Language questioning model as a research method.

Theoretically, the study provided insights into the telling of stories (of personal experiences) and the role of conceptual metaphors in conveying a greater understanding of the narrative embedded within them. This was specifically regarding how the theory of frames could be used to determine an individual's narrative against the work of Tompkins and Lawley's (2006) PRO model, McAdams and McLean (2013) on Narrative Identity Constructs, and represents values and beliefs in relation to Schwartz's (2012) theory of Universal Values.

Practically, the study sees the development of a new model of reflection based on the research findings. This is based on empirical evidence from the research and acknowledges the complexity and individual nature of the reflective process. It amalgamates the three strands of the thesis (exploration of conceptual metaphors, interpretation of narratives and use of the Clean Language model) to promote a pragmatic model of reflection as an alternative to traditional cyclical process-driven approaches to reflection, which are described in more detail below within section 7.3.

#### 7.3 An Emerging Model of Reflection

The study resulted in the development of a model of reflection amalgamating the findings of the three research questions, combining elements of the Clean Language questions to elicit metaphors, identify conceptual metaphors and refer to frames to promote reflective practice amongst academics and educators. The model, shown as a diagram below, is subsequently explained via links to findings from the study further within this section.



Figure 4 - Proposed Model of Reflection

To demonstrate how the proposed model of reflection links to theory and an analysis of the findings, it is recommended that reflective practice should be encouraged via discussions with others, as shown in the centre of the diagram stating, 'Reflections via dialogue and discussions'. This aspect of the model, stemming from a conversational approach to reflecting on experiences, was welcomed by participants.

Adopting a conversational approach to reflection would allow participants to air their teaching stories and enable facilitators to determine the dominant metaphors and frames permeating across the reflections (Bold, 2012).

#### 7.3.1 Eliciting metaphors

The model proposes a flexible approach to eliciting metaphors via conversation in keeping with Black and Halliwell's (2000), Bullman and Schutz's (2006) and Guo's (2021) notions of how reflective practice approaches need to consider the different ways individuals learn. The approach encouraged within this model would also take into account the findings allied to the effectiveness of reflecting aspects and words of the participant as responsive paraphrases and probing questions, and the use of a limited range of four Clean Language questions (as

previously discussed) which would require only nominal training to build competence and confidence in the use of clean interviewing techniques.

The use of reflective summaries and probing questions is consistent with the findings, showing how participants reflect on their practice while maintaining a flow of discussions. Reflective summaries and probing questions promote empathy and deeper and meaningful exploration of the key issues (Nelson-Jones, 2014; Mearns et al., 2013). In keeping with the key principle of the Clean Language model of the need for the interviewer to have a detached facilitator role in influencing and interpreting the participants' responses (Lawley & Tompkins, 2000; Sullivan & Rees, 2008), it is important, therefore, to acknowledge how the use of reflective paraphrase prompts, within the context of the Clean Language model, represents a 'clean' reactionary stimulus to participants' utterances. This is a way of responding which recognises how the move away from the use of the Clean Language 'syntax' is more appropriate to the research interview context and a more pragmatic approach to the Clean Language model and questions (Meese, 2014).

In eliciting metaphors, some awareness of the concept and differences between the source and target domains, and structural, orientational and ontological metaphor types could aid in exploring and identifying conceptual metaphors (Lakoff & Johnson, 1980; Kövecses, 2010). Therefore, some understanding, recognition and training with respect to these concepts of working with metaphors would be helpful alongside the training necessary to employ a limited range of Clean Language questions appropriately.

#### 7.3.2 Identify conceptual metaphors

The aim of identifying conceptual metaphors within the model is to help the participants gain an understanding of the subtle impact metaphors have in making sense of complex thought patterns via metaphorical analogies, particularly with regard to their cognitive and decisionmaking processes, their wider practice and the isomorphic qualities metaphors have on conveying more profound thoughts (Robins & Mayer, 2010). This process is aided by exploring the entailments of a metaphor (transfer of information from source to a target domain) and the inferences resulting from these entailments. Identifying an agreed-upon conceptual metaphor further can act as a 'tethering post' and as a focal point to explore the frames, both limiting and enhancing, which surround the conceptual metaphors and may potentially impact practice (Hartley, 2020). The process of identifying conceptual metaphors could take account of and consider Lawley and Tompkins' (2000) suggestions of the need to observe repeated or corresponding patterns of metaphors within conversations. Specifically, Lawley and Tompkins (2000) encourage the determination of three characteristics of a pattern that could act as a catalyst; namely the parts (i.e., source and target domain of metaphors) of a pattern, the arrangement (mapping of source and target domains) of a pattern and how individuals repeat these patterns. As part of the reflective process within the proposed model, the facilitator could take account of using Clean Language questions, a reflective summary and probing questions to note the patterns and analyse them with regard to the similarities and differences, and in relation to the compare and contrast analysis of the metaphor referred to in order to establish the primary conceptual metaphors emerging out of the reflective process.

The process of determining conceptual metaphors emerging out of the discussions could further be aided by reference to examples of conceptual metaphors from existing publications. For example, Lakoff et al.'s (1991) Master Metaphor List could provide an avenue whereby conceptual metaphors could be categories consisting of events, structures, mental events and emotions, which might be a starting point for exploring patterns of metaphors arising from reflective discussions with individuals.

#### 7.3.3 Explore frames

The exploring frames stage of the model of reflection would focus on summarising and confirming the presence of the conceptual metaphor with participants before progressing to discuss and identify the impact the frames surrounding a conceptual metaphor would have on a person's thoughts and behaviours. This stage of the proposed model of reflection would allow for a deeper level of awareness, reflection and exploration of the broader impact of the conceptual metaphors in expressing a preferred schema of teaching over others.

Within this study, the PRO model (Tompkins & Lawley, 2006), McAdams and McLean's (2013) Narrative Constructs and Schwartz's (2012) Universal Values Theory were used as a framework to explore and examine the frames and narratives evident and expressed via metaphors. However, another alternative range of frames could be used to explore conceptual metaphors within different contexts and give flexibility to the participant to examine metaphors in line with the context, remit and aim of the reflective practice being pursued.

A vulnerability in analysing and working with metaphors is emphasised by Cook-Sather (2003) and Carpenter (2008), who stress that adopting a specific metaphor in the analysis of experiences, in preference to other metaphors, could limit rather than enhance self-awareness and practice. Hence, careful consideration needs to be given to the use of metaphors. By default, consciously or unconsciously, using a specific metaphor and frame to convey and interpret an experience could exclude the substance and importance of alternative interpretations of experiences. Therefore, as part of the 'identifying frames' stage of the proposed reflective model, it would be important to adopt a questioning approach to determining patterns of relevance within the conceptual metaphors identified as the significance, and a probing approach to explore frames surrounding the conceptual metaphors.

#### 7.3.4 Links and transferability to practice

The value of the proposed model emerging from the findings can be further determined by exploring the potential links to practice and the transferability of the model within different contexts. As well as higher education, the generic approach and process in which this model of reflection could be applied and utilised would enable and appeal to a range of educational contexts (schools, FE) where reflective practice is encouraged and required to enhance pedagogy and practice and continuing professional development. The model might also be promoted and used within social care and healthcare settings, where reflective practice is also valued and could further be utilised within professional activity where supervision, monitoring, and evaluation activities (e.g., management, performance appraisals, etc) are encouraged and practised.

Appreciating the significance of the value of conversational storytelling could be further considered within the teaching context. Green (2004) identifies how the role of storytelling in teaching has been superseded by a focus on imparting knowledge by the sharing of facts (e.g., bullet points on PowerPoint slides). However, utilising stories and narratives within a teaching context may aid learning by personalising content, generating interest, making learning more relevant, and overcoming anxiety by allowing learners to reflect on how their own stories and experiences relate to their learning and apply to practice. Similarly, the value of exploring metaphors and reflections via specific and chosen frames of references linked to a relevant theoretical lens (e.g., professional identity, personality/behavioural traits, pedagogy, etc.), specific context, or profession may engender a bespoke approach to reflective practice not always possible by following traditional cyclical methods of reflection.

There might also be further scope for the Clean Language questioning model's transferability within different contexts in addition to its use as a qualitative research method as utilised within this study. The findings demonstrating the value of a limited number of four key Clean Language questions, which could be assimilated without the need for specialist training could have wider applications to encourage reflection. For example, employing Clean Language questions, personal development reviews, group work, brainstorming, etc. indicates the wider relevance and significance of the findings of the study and the possible application of the proposed model of reflection amongst different contexts. This aspect of the model is potentially further enhanced by adopting a conversational approach to reflection, as mentioned previously, to exploring metaphors via the proposed model with the aid of a facilitator who appreciates the vocational context of the participants.

#### 7.4 Limitations and Caveats of the study

In addition to the challenges of working with retrospective data and addressing phenomenological perspectives, as discussed within sections 2.3, 3.2 and 3.3.2. The necessity of grounding the findings in the data collected and of reconciling any findings that might initially appear contradictory were accommodated as part of the analysis of the data within this study (Neusar, 2014). This was achieved by allowing and involving the participants in reviewing the findings from the initial analysis both individually and via focus group discussion, which confirmed the relevance of the preliminary findings from the analysis of the data. Furthermore, the critical nature of the relationship and regular meetings between the two researchers encouraged an evaluative approach to the relevance of the findings and informed the initial conclusions of the data analysis.

The findings from this study are based on a small sample of participants which undermines the potential for generalisation. This research story, it must be acknowledged, depends upon subjective dispositions of metaphors, liable to changes in both moods and circumstances, as conveyed by participants from the same institution, and interpreted by a researcher, at a particular point in time. Clearly, this limited sample of 11 participants, although diverse in age, experience, and vocational discipline (see Appendix three), cannot be deemed representative of the wider academic population. However, they do constitute a rich and recognisable research story. Furthermore, the study's research questions aimed to identify the narratives

represented within participants' conceptual metaphors and to evaluate the use of the Clean Language questioning model within the specific context of a sample of HE academics instead of inferences signifying broader generalisations of aspects of practice across other contexts (Neusar, 2014).

#### **Concluding Summary**

Overall, this study helped to determine how the limitations of traditional methods of reflection encouraged and promoted within education and, within the context of this study, HE, could be addressed by fostering a storytelling approach to reflection, exploration of conceptual metaphors and the use of the Clean Language questioning model as part of the storytelling process. The findings and conclusions of the study link to reflective practice within HE, help to attend to notions of how reflective practice has yet to be fully integrated into HE to enhance teaching practice and to promote professional development (Chan & Lee, 2021), specifically, regarding the use of a conversational approach to reflection and use of alternative methods of reflection practice (Guo, 2021). Further exploration of the deeper narratives embedded within the conceptual metaphors could also help to inform and give insights into how academics contextualise, conceptualise and adapt their teaching practice. The study contributes to our understanding of the value of exploring conceptual metaphors with reflective and teaching contexts, the potential of referring to the Clean Language questioning model as qualitative interviewing methods and to an awareness gained from analysing the narratives as part of the reflective process. Although set within the backdrop of academics working within a HE setting, the insights emerging from this study would be of interest and relevance to educators within a range of contexts who are required or appreciate the value of reflection to enhance teaching practice.

Finally, the metaphorical journey of completing this thesis has ultimately been one where peaks and many valleys have had to be negotiated and managed, and where the passengers on the journey (i.e., support from others) have ultimately aided and led to the completion of the journey and this thesis. The insights gained from completing the journey have helped to form a narrative, best summarised by the quote: "you can't write the next chapter, without turning the page on the last".

MH Karolia

## References

Allbritton, D. W., McKoon, G., & Gerrig, R. J. (1995). Metaphor-based schemas and text representations: Making connections through conceptual metaphors. *Journal of Experimental Psychology: Learning, Memory, and Cognition,* 21 (3), pp. 612-625.

Andrews, M., Squire, C., & Tamboukou, M. (2013). *Doing Narrative Research* (2<sup>nd</sup> ed.). Sage.

Armstrong, P. (2020). *Stories and the Brain: The Neuroscience of Narrative*, Johns Hopkins University Press

Armstrong, S., Davis, H., & Paulson, E. (2011). The Subjectivity Problem: Improving Triangulation Approaches in Metaphor Analysis Studies in *International journal of qualitative methods*, 06/2011, Volume 10, Issue 2, DOI https://doi.org/10.1177/160940691101000204

Ashwin, P., & Boud, D. (2020). *Reflective Teaching in Higher Education* (2nd ed.). Bloomsbury Academic

Barry, C. L., Brescoll, V. L., Brownell, K. D., & Schlesinger, M. (2009). Obesity metaphors: How beliefs about the causes of obesity affect support for public policy. *Milbank Quarterly*, *87*(1), 7–47. doi:10.1111/j.1468-0009.2009.00546.x

Battino, R. (2005). *Metaphoria: Metaphor and guided metaphor for psychotherapy and healing*. Crown House Publishing Ltd

Billot, J. (2010). The imagined and the real: identifying the tensions for academic identity *Higher Education Research & Development*, 29:6, 709-721, DOI: 10.1080/07294360.2010.487201

Black, A. L., & Halliwell, G. (2000). Accessing practical knowledge: How? Why? *Teaching and Teacher Education*, *16*(1), 103–115. doi:10.1016/s0742-051x(99)00045-1

Bleakley, A. (2005). Stories as data, data as stories: making sense of narrative inquiry in clinical education. *Medical Education*, *39*(5), 534-40. doi: 10.1111/j.1365-2929.2005.02126.

Bold, C. (2012). Using Narratives in Research. Sage Publications.

Bolton, G. (2005). *Reflective practice: Writing and professional development*. 2nd edn. SAGE Publications.

Booker, C. (2004). *The seven basic plots: Why we tell stories*. Continuum.

Borah, P. (2011). Conceptual Issues in Framing Theory: A Systematic Examination of a Decade's. *Journal of Communication*, *61*, 246–263. doi:10.1111/j.1460-2466.2011.01539.x

Botha, E. (2009). Why Metaphors in Education Matter in *South African Journal of Education*, Vol 29:43, pp.1-444. DOI:10.15700/saje.v29n4a287

Boud, D., & Walker, D. (1998). "Promoting reflection in professional courses: The challenge of context," *Studies in Higher Education*, 23(2), pp. 191–206. doi: 10.1080/03075079812331380384

Boud, D., Keogh, R., & Walker, D. (eds.) (1985). *Reflection. Turning experience into learning*, Kogan Page.

Brace, I. (2013). *Questionnaire design: How to plan, structure and write survey material for effective market research* (2nd ed.). Kogan Page Ltd

Braun, V., & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative Research in Psychology*, 3(2), 77-101. https://doi.org/10.1191/1478088706qp063oa

Brinkmann, S., & Kvale, S. (2005). Confronting the Ethics of Qualitative Research. *Journal of Constructivist Psychology*, *18*(2), 157–181. doi:10.1080/10720530590914789

British Educational Research Association (BERA) (2018). *Ethical guidelines for educational research*. 4th ed. Available at: https://www.bera.ac.uk/researchers-resources/publications/ethical-guidelines-for-educational-research-2018

Brookfield, S. (1987). Developing Critical Thinkers: Challenging Adults to Explore Alternative Ways of Thinking and Acting: Jossey-Bass.

Brookfield, S. (1995). Becoming a Critically Reflective Teacher., Jossey Bass

Bruner, J. S. (1990). Acts of meaning. Harvard University Press.

Bryman, A., & Bell, E. (2015). Business research methods (Fourth ed.). Oxford University Press

Bullough, R.V., Knowles, J.G., & Crow, N.A. (1992). Emerging as a teacher. Routledge

Bulman, C. & Schutz, S. (2008). Reflective Practice in Nursing, 4, Chichester: Blackwell Publishing

Bulman, C., & Schutz, S. (2013). Reflective practice in Nursing (4th ed.). Blackwell.

Burchell, H., & Dyson, J. (2005). Action research in higher education: exploring ways of creating and holding the space for reflection, *Educational Action Research*, 13:2, 291-300, DOI: 10.1080/09650790500200280

Burchell, H., & Dyson, J. (2000). Just a little story: The use of stories to aid reflection on teaching in higher education. *Educational Action Research*, *8*(3), 435–450. doi:10.1080/0965079000020012

Burrell, G., & Morgan, G. (1992). Sociological paradigms and organisational analysis: Elements of the sociology of corporate life. Ashgate

Cameron, L. (n.d.). Identifying and describing metaphor in spoken discourse. In L. Cameron & G. Low (Eds.), *Researching and Applying Metaphor* (pp. 105–132). Cambridge University Press.

Cameron, L., & Low, G. (1999). metaphor. *Language Teaching*, 32(2), 77-96. https://doi.org/10.1017/S0261444800013781

Carpenter, J. (2008). Metaphors in qualitative research: Shedding light or casting shadows. *Research in Nursing and Health*, *31*(3), 274-282. doi.org/10.1002/nur.20253

Chan, C. K. Y., & Lee, K. K. W. (2021). Reflection literacy: A multilevel perspective on the challenges of using reflections in higher education through a comprehensive literature review. *Educational Research Review*, 32, 100376. https://doi.org/10.1016/j.edurev.2020.100376

Charmaz, K. (2006). *Constructing grounded theory: A practical guide through qualitative analysis*. SAGE.

Charmaz, K. (2014). Constructing grounded theory (2nd ed.). SAGE

Chong, D., & Druckman, J. N. (2007). A theory of framing and opinion formation in competitive elite environments. *Journal of Communication*. https://doi.org/10.1111/j.1460-2466.2006.00331.x

Clandinin, D. & Connelly, M. (2000). *Narrative Inquiry: Experience and Story in Qualitative Research*. Jossey-Bass Publishers

Clegg, S. (2008). Academic identities under threat?, *British Educational Research Journal*, 34:3, 329-345, DOI: 10.1080/01411920701532269

Cook-Sather, A. (2003). Movements of mind: The matrix, metaphors, and re-imagining education. *Teachers College Record*, 105(6), pp 946–977

Cooperrider, D. L., Whitney, D., Stavros, J. M., & Fry, R. (2007). *Appreciative inquiry handbook, 2nd edition (book & CD)* (2nd ed.). Crown Custom Publishing.

Creswell, J. W. (2007). *Qualitative inquiry and research design: Choosing among five traditions*. Sage Publication

Creswell, J. W. (2009). *Research design: Qualitative, quantitative, and mixed methods approaches* (3rd ed.). Sage Publications, Inc

Creswell, J. W. (2013). *Qualitative inquiry and research design: Choosing among five approaches*. 3rd ed. SAGE Publications.

Crotty, M. (1998). The foundations of social research.: Sage Publications.

Czarniawska, B. (2004). *Narratives in Social Science Research*. Sage Publications. https://doi.org/10.4135/9781849209502

Daiute, C. (2013). Narrative inquiry: A dynamic approach. Sage Publications.

Dewey, J. (2008). How we think. Book Jungle.

DiCicco-Bloom, B. & Crabtree, B. (2006). The Qualitative Research Interview, *Medical Education*, 40, 4 pp. 314-321

Dilts, R. V. (2006). Sleight of mouth: The magic of conversational belief change. Meta Publications

Dunbar, A. (2017). Clean Coaching: The insider Guide to Making Change Happen. Routledge.

Elliott, J. (2005). Using narrative in social research: Qualitative and quantitative approaches. SAGE

Elo, S., & Kyngäs, H. (2008). The qualitative content analysis process. *Journal of Advanced Nursing*, 62(1), 107-115. https://doi.org/10.1111/j.1365-2648.2007.04569.x

Fechter, J. (2021). *What is Dan Harmon's Story Circle?* https://www.squibler.io/. https://www.squibler.io/blog/dan-harmons-story-circle/

Fielding, N. (1994). Varieties of research interviews. *Nurse Researcher*, *1*(3), 4–11. doi:10.7748/nr1994.04.1.3.4.c6292

Fivush, R. (2008). Remembering and reminiscing: How individual lives are constructed in family narratives. *Memory Studies*, 1(1), 49-58. https://doi.org/10.1177/1750698007083888

Freeman, T. (2006). "Best practice" in focus group research: Making sense of different views. *Journal of Advanced Nursing*. https://doi.org/10.1111/j.1365-2648.2006.04043.x

Garnet, D., Sinner, A., Walker, C., Esmat, R., & Yi, S. (2018). Learning in the third age: drawing wisdom from reflective stories in community art education. *International Journal of Lifelong Education*, *1370*, 1–14. https://doi.org/10.1080/02601370.2018.1450304

Geary, J. (2011). I is an Other, Harper Collins Publishers

Gendlin, E. T. (1982). Focusing (2nd ed.). Bantam Doubleday Dell Publishing Group

Georgakopoulou, A. (2007). Small stories, interaction and identities. John Benjamins Publishing Co.

Gibbs, G. (1988). *Learning by Doing, A Guide to Teaching and Learning Methods*, Oxford Polytechnic FEU

Gibbs, R. W. (2017). *Metaphor wars: Conceptual metaphors in human life*. Cambridge University Press. https://doi.org/10.1017/9781107762350

Gill, P., Stewart, K., Treasure, E., & Chadwick, B. (2008). Methods of data collection in qualitative research: interviews and focus groups. *British Dental Journal*. https://doi.org/10.1038/bdj.2008.192

Gillis, C., & Johnson, C. L. (2002). Metaphor as renewal: Re-imagining our professional selves. English *Journal*, 91(6), 37-43. https://doi.org/10.2307/821814

Goffman, E. (1974). *Frame analysis: An essay on the organization of experience.* Harvard University Press

Grbich, C. (2012). *Qualitative data analysis: An introduction* (2nd ed.). Sage Publications – update to 2012

Green, M. (2004). *Storytelling in Teaching*. Association of Psychological Sciences. https://www.psychologicalscience.org/observer/storytelling-in-teaching

Grove, D. (1996). And ...What kind of a Man is David Grove? an interview by Penny Tompkins and James Lawley, *Rapport*, 33

Gudmundsdottir, S. (2001). *Narrative Research in School Practice*. In V. Richardson (Ed.), Fourth Handbook for Research on Teaching (pp. 226-240). Macmillan

Guo, L. (2021). How should reflection be supported in higher education? A meta-analysis of reflection interventions. *Reflective Practice*, 23(1), 118 146. https://doi.org/10.1080/14623943.2021.1995856

Habermas, T., & Bluck, S. (2000). Getting a life: The emergence of the life story in adolescence. *Psychological Bulletin*, 126(5), 748-769. https://doi.org/10.1037/0033-2909.126.5.748

Hartley, T. (2020). The Listening Space (2nd ed.). Routledge. Listening Space Publications,

Ho, J. W. Y. (2005). Metaphorical construction of self in teachers' narratives. *Language and Education*, *19*(5), 359–379. doi:10.1080/09500780508668691

Hoffman-Kipp, P., Artiles, A. J., & Lopez-Torres, L. (2003). Beyond reflection: Teacher learning as Praxis. *Theory into Practice*, 42(3), 248–254. doi:10.1353/tip.2003.0030

Jääskelä, P., Poikkeus, A., Vasalampi, K., Valleala, U. M., & Rasku-Puttonen, H. (2017). Assessing agency of university students: Validation of the AUS scale. *Studies in Higher Education* (Dorchester-on-Thames), 42(11), 2061-2079. https://doi.org/10.1080/03075079.2015.1130693

Johns, C. (2000). Becoming a Reflective Practitioner, A Reflective and Holistic Approach to Clinical Nursing, Practice Development and Clinical Supervision, Blackwell Science

Kahneman, D. (2012). Thinking, fast and slow. Penguin.

Karolia, M. (2014, June 6-7). *Metaphors we teach by*. Poster presented at the 2014 International Clean Language Conference, London.

Karolia, M. (2015). Using Metaphors in Teaching, Training, and Learning. In L. Ashmore, & D. Robinson (Eds.), *Learning, Teaching and Development: Strategies for Action* (pp. 99-116). Sage Publishing.

Karolia, M., & Burton, R. (2012, July 6-7). *Metaphors we teach by: An exploration into how HE lecturers reflect on and conceptualise their experiences of teaching via the use of clean language interviewing model of exploring metaphors*. Presentation presented at the 3rd International NLP Research Conference, Hertfordshire.

Kolb, D. (1984). Experiential Learning, Prentice-Hall Inc.

Kövecses, Z. (2010). Metaphor: A Practical Introduction (2nd ed.). Oxford University Press.

Lakoff, G., & Johnson, M. (1980). Metaphors We Live By, London: The University of Chicago Press

Lakoff, G., & Johnson, M. (2003). Metaphors We Live By (2nd ed.). The University of Chicago Press

Lakoff, G., & Turner, M. (1989). *More than cool reason: A field guide to poetic metaphor*. University of Chicago Press.

Lakoff, G., Espenson, J., & Schwartz, A. (1991). *Master Metaphor List.* http://araw.mede.uic.edu/~alansz/metaphor/METAPHORLIST.pdf

Lawley, J. (2014). Protocol for Validating 'Cleanness' of an Interview. http://www.cleanlanguage.co.uk/articles/attachments/5-Protocol\_for\_Validating\_Cleanness\_of\_Interviews\_v3.pdf

Lawley, J. (2014). Using the CL syntax in research interviews? https://www.linkedin.com/groups/4622774/4622774-5834679703016845316

Lawley, J. (2017). Clean Language Interviewing: Making qualitative research interviews verifiable. In J. Nehyba, & P. Svojanovsky (Eds.), *Becoming a Teacher: The Dance Between Tacit and Explicit Knowledge* (pp. 115-122). Masarykova univerzita.

Lawley, J. (2021). *Publications that make use of David Grove's ideas*. The Clean Collection. https://www.cleanlanguage.co.uk/articles/articles/264/1/Citations-and-Research/Page1.html

Lawley, J., & Tompkins, P. (2000). *Metaphors of the Mind: Transformation Through Symbolic Modelling*, UK: The Developing Company Press

Lawley, J., & Way, M. (2017). *Insights in Space: How to Use Clean Space to Solve Problems, Generate Ideas and Spark Creativity*. Clean Publishing.

Lawley, J., & Small, S. (2018). *Clean Language Interviewing in 2018 with James Lawley*. Retrieved from http://cleanlanguagetraining.com/clean-interviewing/clean-language-interviewing-sharon-small-james-lawley-2018/.

Leedy, P. D., & Ormrod, J. E. (2015). *Practical research: Planning and design* (Global; Eleventh; ed.). Pearson.

Leshem, S., & Trafford, V. (2007). Overlooking the conceptual framework. Innovations in *Education* and *Teaching International*, 44(1), 93-105. https://doi.org/10.1080/14703290601081407

Lester, S. (1999). An introduction to phenomenological Research. Retrieved May 22, 2013, from http://www.sld.demon.co.uk/resmethy.pdf

Lieblich, A., Tuval-Mashiach, R., & Zilber, T. (1998). *Narrative Research: Reading, Analysis, and Interpretation, Vol. 47* (1st ed.). Sage Publications (CA).

Lietz, P. (2010). Research into questionnaire design: A summary of the literature. *International Journal of Market Research*, *52*(2), 249. doi:10.2501/s147078530920120x

Loftus, E. F. (2005). *Planting misinformation in the human mind: A 30-year investigation of the malleability of memory*. Learning & Memory, 12, 361–366.

Lynch, M. (1991). Tell Me a Story. Isleworth: BFSS National RE Centre Brunel University.

Macfarlane, B. (2004; 2003). *Teaching with integrity: The ethics of higher education practice*. RoutledgeFalmer. https://doi.org/10.4324/9780203416501

Mahlios, M., Massengill-Shaw, D., & Barry, A. (2010). Making sense of teaching through metaphors: a review across three studies, *Teachers and Teaching*, 16:1, pp 49-71

Mälkki, K., & Lindblom-Ylänne, S. (2012). From reflection to action? Barriers and bridges between higher education teachers' thoughts and actions. *Studies in Higher Education* (Dorchester-on-Thames), 37(1), 33-50. https://doi.org/10.1080/03075079.2010.492500

Marshall, B., Cardon, P., Poddar, A., & Fontenot, R. (2013). Does sample size matter in qualitative research?: A review of qualitative interviews in is research. *The Journal of Computer Information Systems*, 54(1), 11-22. https://doi.org/10.1080/08874417.2013.11645667

Martínez, M. A., Sauleda, N., & Huber, G. L. (2001). Metaphors as blueprints of thinking about teaching and learning. *Teaching and Teacher Education*, *17*(8), 965–977. doi:10.1016/s0742-051x(01)00043-9

Mason, J. (2018). Qualitative researching (Third ed.). SAGE

McAdams, D. (2009). *The Person, An Introduction to the Science of Personality Psychology.* 5, John Wiley & Sons Inc

McAdams, D. P. (2001). The psychology of life stories. *Review of General Psychology*, 5(2), 100-122. https://doi.org/10.1037/1089-2680.5.2.100

McAdams, D. P., & McLean, K. C. (2013). Narrative identity. Current Directions in Psychological Science: *A Journal of the American Psychological Society*, 22(3), 233-238. https://doi.org/10.1177/0963721413475622

McAdams, D. P., & Pals, J. L. (2006). A new Big Five: Fundamental principles for an integrative science of personality. *American Psychologist*, 61, 204–21

McAdams, D. P., McAdams, D. P. (Ed.), Josselson, R. (Ed.), & Lieblich, A. (Ed.) (2006). *Identity and story: Creating self in narrative*. American Psychological Association Press.

McAlpine, L., & Weston, C. (2000). Reflection: Issues related to improving professors' teaching and students' learning. *Instructional Science*, 28(5/6), 363-385. https://doi.org/10.1023/A:1026583208230

McCracken, J. (2016). Clean Language in the Classroom. Crown House Publishing

McDonald, N., Schoenebeck, S., & Forte, A. (2019). Reliability and inter-rater reliability in qualitative research: Norms and guidelines for CSCW and HCI practice. *Proceedings of the ACM on Human-Computer Interaction*, 3(CSCW), 1-23. https://doi.org/10.1145/3359174

McEvoy, P. (2002). Interviewing colleagues: Addressing the issues of perspective, inquiry and representation. *Nurse Researcher*, *9*(2), 49–59. doi:10.7748/nr2002.01.9.2.49.c6182

Mearns, D., Thorne, B., & McLeod, J. (2013). Person-centred counselling in action (Fourth ed.). SAGE.

Meese, R. (2014). Using the CL syntax in research interviews? https://www.linkedin.com/groups/4622774/4622774-5834679703016845316 Mertens, D. M. (2020). Research and evaluation in education and psychology: Integrating diversity with quantitative, qualitative, and mixed methods (Fifth ed.). Sage.

Miles, M. B. & Huberman, A. M. (1984) *Qualitative data analysis: a sourcebook of new methods*, Sage

Miles, M. B., & Saldaą, J. (2014). Qualitative data analysis: A methods sourcebook (3.th ed.). Sage

Miller, R. L., & Brewer, J. D. (2003). *The A-Z of social research a dictionary of key social science research concepts*. SAGE.

Moen, T. (2006). Reflections on the Narrative Approach. *International Journal of Qualitative Methods*, *5*(4), 56–59.

Moreno, V. (2002). Validity Issues in Phenomenological Research: AMCIS 2002 Proceedings. 241. http://aisel.aisnet.org/amcis2002/241

Morse, J. M. (2000). Determining sample size. *Qualitative Health Research*, 10(1), 3-5. https://doi.org/10.1177/104973200129118183

Morson, G. S. (1994). Narrative and freedom: The shadows of time. Yale University Press.

Moser, K. S. (2000). Metaphor Analysis in Psychology—Method, Theory, and Fields of Application. *Forum Qualitative Sozialforschung*, 1(2), Art 21. http://nbn-resolving.de/urn:nbn:de:0114-fqs0002212

Moustakas, C. E. (1994). Phenomenological research methods (6th ed.). SAGE Publications.

Noon, E. J. (2018). Interpretive phenomenological analysis: An appropriate methodology for educational research? *Journal of Perspectives in Applied Academic Practice*, 6(1), 75-83. https://doi.org/10.14297/jpaap.v6i1.304

Nelson-Jones, R. (2014). Practical counselling and helping skills (1st ed.). SAGE.

Neusar, A. (2014). To trust or not to trust? interpretations in qualitative research. *Human Affairs* (Bratislava, Slovakia), 24(2), 178-188. https://doi.org/10.2478/s13374-014-0218-9

Ochieng, P. A. (2009). An analysis of the strengths and limitation of qualitative and quantitative research paradigms. *Problems of Education in the 21st Century*, 13, 13.

O'Connor, J., & Seymour, J. (1994). *Training with NLP: Skills for managers, trainers and communicators*. Thorsons.

O'Connor, J., & Seymour, J. (2002). *Introducing NLP: Psychological skills for understanding and influencing people* (Rev. ed.). Element.

Ollerenshaw , J.A. and Creswell, J. (2002). "Narrative Research: A Comparison of Two Restorying Data Analysis Approaches," *Qualitative Inquiry*, 8(3), pp. 329–347.

Oppenheim, A. N. (1992). *Questionnaire design, interviewing, and attitude measurement*. St. Martin's Press.

Orb, A., Eisenhauer, L., & Wynaden, D. (2001). Ethics in qualitative research. *Journal of Nursing Scholarship*, 33(1), 93-96. https://doi.org/10.1111/j.1547-5069.2001.00093.x

Ortony, A. (Ed.) (1979). Metaphor and Thought. Cambridge University Press

Ortony, A. (Ed.) (2008). Metaphor and Thought: 2. Cambridge University Press

Pajares, M. F. (1992). Teachers' beliefs and educational research: Cleaning up a messy construct. *Review of Educational Research*, 62, pp 307–332

Parker, A., & Tritter, J. (2006). Focus group method and methodology: Current practice and recent debate. In *International Journal of Research and Method in Education*. https://doi.org/10.1080/01406720500537304

Perrin, S. (1987). Metaphorical Revelations: A description of metaphor as the reciprocal engagement of abstract perspectives and concrete phenomena in experience in *Metaphor and Symbolic Activity*, 2, pp. 251-280

Polkinghorne, D. (1989). *Phenomenological research methods*. In R. Valle & S. Halling (Eds.), Existential phenomenological perspectives in psychology: Exploring the breadth of human experience (pp. 41- 60). New York: Plenum. doi:10.1007/978-1-4615-6989-3\_3

Polkinghorne, D. E. (2007) "Validity Issues in Narrative Research." *Qualitative Inquiry 13*, no. 4 (May 2007): 471–86. doi:10.1177/1077800406297670.

Pollard, A. (2008). *Reflective Teaching: Evidence-informed Professional Practice* (3rd ed.). Bloomsbury Publishing.

Ravitch, S. M., & Riggan, M. (2017). *Reason & rigor: how conceptual frameworks guide research* (Second edition). SAGE Publications.

Reddy, M. J. (1979). The Conduit Metaphor: A case of Frame Conflict in our Language about Language. In *Metaphor and Thought* (pp. 284–310).

Rees, J. (2016). Your Clean Language Questions answered (ebook). http://judyrees.co.uk/your-clean-language-questions-answered/

Revans, R. (2017). *ABC of action learning*. Taylor and Francis. https://doi.org/10.4324/9781315263533

Riessman, C. K. (1993). Narrative analysis. Sage Publications.

Riessman, C. K. (2008). Narrative methods for the human sciences. SAGE Publications.

Robins, S., & Mayer, R. E. (2000). The metaphor framing effect: Metaphorical reasoning about textbased dilemmas. *Discourse Processes*, 30(1), 57-86. https://doi.org/10.1207/S15326950dp3001\_03 Robinson, O. (2014). Sampling in Interview-Based Qualitative Research: A Theoretical and Practical Guide. *Qualitative Research In Psychology*, 11(1), 25-41.

Saban, A. (2006). Functions of metaphor in teaching and teacher education: A review essay. *Teaching Education*, 17(4), 299–315. doi:10.1080/10476210601017386

Saldana, J. (2013). The coding manual for qualitative researchers (2<sup>nd</sup> ed). SAGE Publications

Saldana, J. (2015). The coding manual for qualitative researchers. SAGE Publications.

Salmon, P., & Riessman, C. K. (2013). Looking back on narrative research: An exchange. In M. Andrews, C. Squire, & M. Tamboukou (Eds.), *Doing narrative research* (2nd ed., pp. 197–204). Sage

Saunders, M., Lewis, P., & Thornhill, A. (2012). *Research Methods for Business Students* (6th ed.). Pearson Education Ltd.

Schmitt, R. (2005). The Qualitative Report Systematic Metaphor Analysis as a Method of Qualitative Research Systematic Metaphor Analysis as a Method of Qualitative Research. *The Qualitative Report*, *10*(2), 358–394. http://nsuworks.nova.edu/tqr

Schön, D. (1983). The Reflective Practitioner: How Professionals Think in Action. Temple Smith

Schön, D. (1987). Educating the Reflective Practitioner. Jossey-Bass

Schön, D. (1988). Coaching Reflective Thinking. In P. P. Grimmett & G. L. Erickson (Eds.), *Reflection in teacher education* (pp. 19–29). New York: Pacific Education Press.

Schwartz, S. H. (2012). An Overview of the Schwartz Theory of Basic Values. *Online Readings in Psychology and Culture*, 2(1). https://doi.org/10.9707/2307-0919.1116

Sipe, L., & Ghiso, M. (2004). Developing Conceptual Categories in Classroom Descriptive Research: Some Problems and Possibilities. *Anthropology & Education Quarterly*, 35: 472-485. https://doi.org/10.1525/aeq.2004.35.4.472

Smith, J. A., Flowers, P., & Larkin, M. (2009). *Interpretative phenomenological analysis: Theory, method and research*. SAGE.

Sparkes, A.C. (1992). The paradigms debate: An extended review and a celebration of difference. In: Sparkes, A.C. Ed., *Research in Physical Education and Sport. Exploring Alternative Visions*, The Falmer Press, 9-60.

Squire, C., Davis, M., Erin, C., Hayden, L.-C., & Hayden, M. (2014). *What is Narrative Research?* London: Bloomsbury.

Steen, G. J., Reijnierse, W. G., & Burgers, C. (2014). When Do Natural Language Metaphors Influence Reasoning? A Follow-Up Study to Thibodeau and Boroditsky (2013). *Plos* One 9 (12: ): e113536. doi: 10.1371/ journal.pone.0113536

Storr, W. (2020). *The Science of Storytelling: Why Stories Make Us Human, and How to Tell Them Better*. William Collins.

Sullivan, W., & Rees, J. (2008). *Clean Language: Revealing Metaphors and Opening Minds*. Crown House Publishing

Surgenor, P. (2011). *Reflective practice tutor, Demonstrator & coordinator development at UCD*. Available at: http://www.ucd.ie/t4cms/Reflective%20Practice.pdf (Accessed: 23 July 2016).

Thibodeau, P. H., & Boroditsky, L. (2011). Metaphors we think with: The role of metaphor in reasoning. *PLoS* One 6: e16782. doi: 10.1371/journal.pone.0016782

Thibodeau, P. H., & Boroditsky, L. (2013). Natural Language Metaphors Covertly Influence Reasoning. *PLoS ONE*. https://doi.org/10.1371/journal.pone.0052961

Thibodeau, P. H., & Boroditsky, L. (2015). Measuring effects of metaphor in a dynamic opinion landscape. *PLoS ONE*. https://doi.org/10.1371/journal.pone.0133939

Thibodeau, P. H., Crow, L., & Flusberg, S. J. (2016;2017;). The metaphor police: A case study of the role of metaphor in explanation. *Psychonomic Bulletin & Review*, 24(5), 1375-1386. https://doi.org/10.3758/s13423-016-1192-5

Tompkins, P., & Lawley, J. (2006). *Coaching FOR P.R.O.'s*. http://www.cleanlanguage.co.uk/articles/articles/31/1/Coaching-for-PROs/Page1.html

Thomson, P. (2018). *what's a framework? – as in, conceptual or theoretical framework*. Patter. https://patthomson.net/2018/11/26/whats-a-framework-conceptual-or-theoretical/

Tosey, P. (2011). "Symbolic Modelling" as an innovative phenomenological method in HRD research: the work-life balance project, paper presented at the 12th International HRD conference, University of Gloucestershire, 25th – 27th May 2011.

Tosey, P., Sullivan, W., & Meyer, M. (2013). Retrieved August 5, 2016, from Clean sources: *Six metaphors a minute?* http://www.cleanchange.co.uk/cleanlanguage/wp-content/uploads/2014/02/Six-metaphors-a-minute-final.pdf

Trick, L., & Katz, A. (1986). The Domain interaction approach to metaphor processing: relating individual differences and metaphor characteristics in *Metaphor and Symbolic Activity*, 1, pp. 185-213

Vaismoradi, M., Turunen, H., & Bondas, T. (2013). Content analysis and thematic analysis: Implications for conducting a qualitative descriptive study. *Nursing & Health Sciences*, *15*(3), 398–405. doi:10.1111/nhs.12048

Van Manen, M. (1997). *Researching lived experience: human science for an action sensitive pedagogy*. The Althouse Press.

Wald, N., & Daniel, B. K. (2020). Enhancing students' engagement with abstract ideas through conceptual and theoretical frameworks. *Innovations in Education and Teaching International*, 57(4), 496-505. https://doi.org/10.1080/14703297.2019.1692055

Walker, C. (2014). From Contempt to Curiosity: Creating the Conditions for Groups to Collaborate Using Clean Language and System Modelling. Clean Publishing.

Way, M. (2013). Clean Approaches for Coaches. Clean Publishing.

Weade, R., & Ernst, G. (1990). Pictures of life in classrooms, and the search for metaphors to frame them in *Theory into Practice*, 29(2), pp. 133–140.

Whiting, L. S. (2008). Semi-structured interviews: Guidance for novice researchers. *Nursing Standard*, *22*(23), 35–40. doi:10.7748/ns2008.02.22.23.35.c6420

Wormeli, R. (2009). *Metaphor Analogies: Power Tools for Teaching any subject*, Stenhouse Publishing

Xerri, D. (2018). *The use of interviews and focus groups in teacher research*. The Clearing House, 91(3), 140-146. https://doi.org/10.1080/00098655.2018.1436820

Yardley, L. (2000). Dilemmas in qualitative health research. *Psychology & Health*, 15(2), 215-228. https://doi.org/10.1080/08870440008400302

Yero, J. L. (2010). *Teaching in Mind: How Teacher Thinking Shapes Education*, 2, MindFlight Publishing

Yob, I. M. (2003). Thinking constructively with metaphors. *Studies in Philosophy and Education, 22*, 127–138.

+Zeichner, K. M., & Liston, D. P. (2014). Reflective teaching: An introduction. Routledge

## Appendix 1 – Clean Language Questions

#### **Developing Questions**

- (and) what kind of 'x' (is that 'x')?
- (and) is there anything else about 'x'?
- (and) where is 'x'? (and) whereabouts is 'x'?
- (and) is there a relationship between 'x' and 'y'?
- (and) when 'x', what happens to 'y'
- (and) 'x' is like what?

#### **Sequence and Source Questions**

- (and) then what happens ? or (and) what happens next?
- (and) what happens just before 'x'?
- (and) where could 'x' come from?

#### **Intentions Questions**

- (and) what would 'x' like to have happen?
- (and) what needs to happen to 'x'?
- (and) can 'x' happen?

Sullivan, W., & Rees, J. (2008). *Clean Language: Revealing Metaphors and Opening Minds*. United Kingdom: Crown House Publishing.

## **Appendix 2 - Participant Consent Form**

Thank you for agreeing to participate in my EdD Doctoral thesis research which aims to explore the effectiveness of using metaphors as method of reflection within Higher Education setting.

This form gives a brief overview of the aims of the research, your commitment to the research, the ethical considerations that you need to be aware of, your rights as a participant in the study and your consent agreeing to participate in the research.

## **Purpose of the Study**

The proposed study, on recognising the value and importance of reflecting on practice within education, aims to counter some of the criticisms levelled at current methods and models of reflective practice by trailing and evaluating a new method of reflection (Clean Language model) that is more akin to the way in individuals think and conceptualise their experiences via metaphors. A study whereby a range of academics working in various HE institutions will be interviewed at various stages through the academic year to determine the effectiveness of the new model of reflection.

## Your Commitment to the Research

The research will require the following from you:

- To complete a minimum of four 60-minutes interview over an academic year.
- A complete a short online questionnaire shortly after each interview aimed at evaluating and gauging your reactions to interview.

## **Ethical Considerations**

You will need to be aware of the following:

- 1. The interview will be recorded.
- 2. The interview will be transcribed and aliases will be used to ensure your identity is protected both within the thesis itself and any notes taken as part of the interviews.
- 3. All transcripts of the interviews are available for you to view should you wish to see them.
- 4. Your employing organisation or the department in which you work will identified within the thesis itself and any notes taken as part of the interviews.
- 5. There is a possibility that the findings from the research may be submitted for publication purposes to the relevant journal(s).
- 6. All data will be stored electronically within a computer folder that is only accessible by the researcher and protected by a password

## Consent

Please read the tick the relevant boxes below:

•	I have been fully informed of the nature and aims of this research	
•	I understand that I have the right to withdraw from the research at any time without giving any reason	
•	I give permission for my words to be quoted (by use of aliases)	
•	I understand that the information collected will be kept in the secure place by the researcher	
•	I understand that no other person other than the researcher and the research supervisor will have access to the information and that all information with be held in secure location by the researcher.	
•	I understand that my identity will be protected by use of a aliases in the report and that no written information that could lead to my being identified will be used in the report	
•	My contribution to this research is entirely voluntary and I am able to withdraw my involvement in the research at any time	

Please do not hesitate to contact the researcher should have any questions about the consent for the research study.

If you are satisfied that you understand the your commitment and involvement with the research study, please sign below.

Signature of the Participant:	Print Name:	Date:
		//
Signature of the Researcher:	Print Name:	Date:
		//

One signed copy to be retained by the Participant and one by the Researcher

## Appendix 3 – Participants Profile

Participant Number	Subject Specialism	Level of HE experience	Interview by	Participant in Focus Group
1	Healthcare	21+ years	Researcher 1	
2	Education	11-15 years	Researcher 1	Yes
3	Science	11-15 years	Researcher 1	Yes
4	Psychology	6-10 years	Researcher 1	Yes
5	Psychology	6-10 years	Researcher 1	
6	Languages	6-10 years	Researcher 2	Yes
7	Business	11-15 years	Researcher 2	Yes
8	Education	11-15 years	Researcher 2	
9	Education	6-10 years	Researcher 2	Yes
10	History	21+ years	Researcher 2	
11	Business	0-5 years	Researcher 2	Yes

# Appendix 4 – Pilot interview sample transcript

## **<u>Pilot Interview 1</u>**

#### S1: Interviewer

#### S2: Responder (referred to as Jane to maintain anonymity)

Timecode	Speaker	Transcript
00:00:001	S1	This is a test just to see if the recording is working. You want to say something, Jane?
00:00:05	S2	Okay. Hi, I am Jane. I am very interested in your study.
00:00:08	S1	Thank you. Now, we will begin. So basically, I would explain a little bit what we're doing. One point that we basically find out is a bit more about your teaching experiences over the last academic year. So feel free to kind of describe your teaching experiences over the last academic year? What would you say it was like?
00:00:30	S2	Okay. I suppose I choose just the way it's like <b>0:35</b> and what would I say it's like? I often regard my teaching has been in the youth group and it has been particularly like that with one or two groups. Busy. What's it like? That's where I am with it.
00:01:04	S1	It's okay. No problem. So when you say youth group, what kind of youth group would that be?

	1r	
00:01:12	52	What youth group it would be? Well, I suppose when I think of when I am saying with a lot of the student's I teach are these community students. So it's <b>lively</b> , it's quite an open conversational kind of process really. So a <b>lively</b> open youth club with lots of conversation going on.
00:01:40	S1	Okay. So I feel when the conversations going on, what kind of conversations are going on?
00:01:50	52	Well, there is lots of debate. There is lots of opinion. There is lots of points for development really, no routes to go down sort of chewing things over and drinking coffee and putting the way it's right those kinds of debates and conversations. But more focus on that obviously.
00:02:21	S1	More focus on that. Okay. So it does a kind of point of development and routes. Is there anything else about the new study, you think?
00:02:33	52	Well, obviously, they are focused on a particular curriculum and I think that the subjects that I teach provides opportunity for those kinds of development and I think about also things that I'm teaching and other modules. I wouldn't say that's quite the same really and that ends up feeling different. What do I describe the other ones as – sometimes it just <b>feels like you are in the school</b> .
00:03:17	S1	Okay. As I did appoint a development on the routes, is there a relationship between the points of development and the routes that you take?
00:03:28	S2	Yeah. I mean what happens is that you see the opportunity to you know all the students would say, he had a particular route and it seems that with the discussion of the topic. And then what you have got to do is to try a spot the points of development for the students and put in the points that you want

00:04:04	S1	at that particular point in time. So to draw them back to the theory and to draw them back to the key concepts that we're trying to discuss to get them to look at a different view things or whatever. Okay.
00:04:08	S2	That's what I mean by the points of development and if it then to be think more critically.
00:04:14	S1	Okay. So if there is a point of development speaking more and thinking more critically putting the points of development thing, what exactly are the points of development?
00:04:35	52	I was to say about thatusually what you find is that the students will grasp the concepts or won't grasp the concept. So for those who grasp the concept in some way, then they will start their conversation, they will start to tell you what they think. They will give you their opinions. At some point in that discussion will be an opportunity to actually challenge of feeding some different information really.
00:05:10	S1	Okay. So does it gasping the concept or not gasping the concept? What has to happen before the gasp the concept?
00:05:22	S2	Well, they have to have the input of some kind and that's happening in the lectures at the moment.
00:05:29	S1	Okay. What kind of an input would that be?

<b></b>	Ĩ	ر ــــــــــــــــــــــــــــــــــــ
00:05:36	52	They got a basic overview and the key concepts that set out to them as far as I am thinking really – it's a bit of a <b>big picture kind of frame way</b> and then when they come to the seminars then what we are trying to do is get them to look at the picture in a bit more detail. So they got to think a bit in terms of pictures and do some of <b>your own</b> .
00:06:08	S1	Okay. What kind of a picture would that be?
00:06:17	S2	Again, limiting we have rather than actually be able to describe it. It's sort of – maybe it's a bit disconnected but it's full of bits, a bit like a country side really here. Yeah. Like a horizon in a particular view.
00:06:46	S1	So it is a picture bits country side, horizons on a particular view. What kind of a view would this be?
00:06:55	52	So it's a You know it might be an outlook coming from a particular perspective really and I think the reason that's in bits are probably contrary more than one perspective. You're trying to get and sew the layouts of the land in the lecture and then the bits are trying to hone in on particular places as those. That would be the way that I would say and to understand the shape of those places and maybe the influences on those places.
00:07:40	S1	So there are places lay out on the <b>07:41</b> to its places and things in particular places. What about <b>site</b> in these places?
00:08:04	S2	Well, something08:05 valleys and hills really. I suppose it depends on the concept of what you are acting08:18 you know for certain equate that the group with module again will come back to may be saying a particular setting where a group would be weaving together so it lines up images of the

		where about is the route? I don't know. It probably goes over the hills far away if probably weaves around the landscape really I suppose.
00:10:29 00:10:34	S1 S2	Okay. And where about is the route?
00:10:19	S2	Oh that's where they are with they're grasping of the concept I suppose, yeah, and it's also why trying to get them to follow the route I suppose.
00:10:13	S1	Stuck in the valleys up in the hills. Is there a relationship between stuck in the valleys up in the hills?
00:09:53	S2	I cannot think of right now really it's just sort of $-1$ suppose it's sort of that relates to what they grasp and grasp for things or not and then whether there a bit stuck in the valley or whether the $-$ up the hill a bit or whatever 1 suppose. Yeah, that's really where I am going.
00:09:31	S1	So your <b>09:32</b> cities and towns and all kind of theory. Is there anything else about the valleys and the hills?
		particular group. If they are talking about gender and society. You might have a place that's a home or whatsoever you know and then a woman would be or something in that kind of own way. We're talking about something that affects society then you might see that as a city or as a part of the city or part of the town or – yeah. And then we used to do a role theory, I used to not get them to look at <b>09:15</b> so you need to get a system and you get things like transport in that way and stuff like that. Maybe sometimes its networks that blocking out certain things they ask them to join and link together.

00:10:52	S1	Okay. And what would need to happen for you to kind of follow the route?
00:11:03	S2	It's probably when I am at the11:04 Well they've got to go with the11:07 you know or so suppose and so much better and trying to leave them in some way or in a quite portion. The source of – I will keep up with them that's where I'm looking for. So marked of them yeah I suppose that this sort of come and follow and that's what usually really like if it's also – or you would like you know, I cannot wait to make because you've heard ahead of you and you try to put those as same sense, to slow down and make them think of it more and be a bit more critical. Some of them you know, almost can drag in into the frame way and some of them you know it's a bit of pushing them up the hill and I cannot think a sort of — that's made mess for us I suppose and I suppose the challenge for me is how will I focus. I will keep them the
00:12:13	S1	So the <b>12:14</b> meant for pushing them up to hill telling you know and giving them a focus and keeping them there. What would need to happen to keep them there?
00:12:36	52	Well it's going to be stimulating, it's going to be interesting, it's you know what comes to my mind as if — as one of the students saying things about you know what more scenarios. She wants to see the picture that set so, to how I have got to you know help them see the picture. She wants the scenario to apply but some of this is about saying for some depth laying in because you know maybe the students don't necessarily want to go up and down the hill so then maybe they are want back kind of outlook, maybe they want more like ice skating rink you know I mean, and they want to get to a place where they can just skate across of it really and which the other side, which is the passing to the silence and that kind of thing. And sometimes it's

		getting them to focus on the <b>13:38</b> or something because maybe you know maybe for another country.
00:13:44	S1	So you've got ice skating rink, you've got hills and the valleys, ice skating in their journey, was anything else about the journey?
00:14:00	S2	<b>14:00</b> different sort of family not really happy, trying to think about teaching other journey because I was thinking about my teaching rather than the <b>14:14</b> that's what you wanted too.
00:14:17	S1	It could be <b>14:18</b> .
00:14:18	S2	Well okay. I suppose it's two things from which showed the two aspects of the route, the teacher under laying it really because if I think about the journey then I suppose I'm sort of thinking of a lot of things on the way and that's, that's what I mean really. I don't quite now almost go a lot out of that I suppose in some way a little bit, up to the land up to their journey, it's you know, it's whether it's a <b>15:00</b> it's smooth or it's <b>15:04</b> or a direct or what makes a good journey? Comfortable with stretching and able to stretch or you know, I have got to change now because I like changing or driving in cars, I like to see the land as I go passed so I have got stuck with me again but I think – it sort of think about the stimulation and feeling comfortable in some <b>15:48</b>
00:15:55	S1	Stop picking up the pieces15:57 and also cannot see the landscapes and the16:02 the landscape. Is there anything else about – Is there a relationship between the like being bumpy and smooth?
00:16:33	52	I don't think the way really for a journey you know, a smooth rises you can put16:25 it's not necessarily, it's not necessarily very stimulating and a
		bumpy ride can just feel uncomfortable and put you off kind of – yeah. So I suppose there's something16:41_ between and that's something that is comfortable and stimulating pleasurable.
----------	----	--
00:16:53	S1	So there's something comfortable and stimulating, you think16:56 about comforting and stimulating in their journey?
00:17:06	52	I think it is probably something about a freedom in the – getting off the shoes sort of <b>17:16</b> Yeah and what I like about road journeys as oppose to rail journeys is being able to choose your own way so it's sort of you know there's something between – whether you're on what type of journey, you're on one way you've got more choice or whether you know one way you're a bit more constructive I suppose.
00:17:52	S1	So you got more journey, freedom and sort about choice. Is there a relationship between the freedom and choice?
00:18:04	S2	Yeah, it's a very strong really I think. But why you think about the teaching, sometimes the students don't have any choice or the choice is they would make it not about what you would be after <b>V&amp;E</b> .
00:18:26	S1	Okay. So we started off on looking that kind of taking in the youth goal and kind of got all we are teaching being something about freedom or choice and 18:41 going to choice, would that be your kind of the place to kind of live it for now?
00:18:46	S2	Yeah, yeah I think so. However, I might say that the youth club – that is exactly where the youth club should be so yeah. It would be the same thing

		really and so much better to date <b>18:58</b> to residential hearts. It's a kind
		of education that I'm used to provide.
00:19:03	S1	Alright. Okay. Thank you.
00:19:06	S2	Thank you.
00:19:07	S1	Thank you.
00:19:08	S2	Very interesting.

#### Appendix 5 – Pilot Survey sample responses

#### VI 149.

Serry line is no deleyed, but it excepted one a bit. Anythen, i hope this is rait too hate for your confirments paper and that this is of some son.

When I are touching at my basi, it is firm over the stations and upped our for term paper protocologies in the In inner of the househow-charlent intersection. Even is an over and impair the same segue protocologies and there is a really trading all charten and approximation for the station and are the baseline. The baseline is dynamic and the spectra of early finitement was called and approximately for the station and are said and an the baseline. The baseline is dynamic and the spectra mathematic and the second approximation of the second second and are said and the baseline. The baseline are the mathematic and that continuous and approximation are by to be president and are the baseline. The baseline termine are mathematic and that continuous advantages use to by to be president, elicity, abaselined and these baseline are to be by the termine of the second of the second are the second of the secon

the an off day, my tensiden is the over a sub-performance that is out really reading the "antisance" or starbank. The engagement and expressly of tearsing is missing and it is, to denial tensor. The pulling teach's The fey and the planears of beaching and framing are missing.

During the last asseturity year, my teaching was like . . . . many things in life, inclusive of smax higher and some lance, installing chellenger which have required randjusting thirdday and teaching pensions, while; should meaning trying to device charagies to anyage with the statents.

When I works this to caller I think a lat in metaphies and one packably rest your and magazoliciti in de so.

Figure Chine by Classified For South-

Cheers,

Hey!

He problems- love this cast of thing- nearly gote are Shiding ....

When I am beaching at my book, it is like . . . I am on top of the world. I ford anything is possible and that my boundedge is finitions. No like helag a cosk sizz - giving the anticasor fits hast performances of your like. They pay a bet for a taket and support a great performance and to and you at your beam delay what you do have that helay teaching your subject.

tin an off day, my taashing is like . . . Ish and miss. I thei i samest engage properly with the sudience and that the andimum has been found to inter up. I think a let of the time ory boucking in allocated by these and smally be not that I and having an all day, but colour that the multice or be bouing on all days.

This was actually quite hand 20 not sume 17 KS the seat of thing you were siter bue I've tried to do it instinctively rether than over Hindricht

When flam tooking stony basi, it is like ...... fying - it feels really free and energising. These is a connection when isoming it visible and students are engaged, a buzz in the room. It reinforces the solventages of my preferred approach gooductive) to teaching and learning experiences

the an algorithm of the gradual has been algorithm to be the second graduatic second of a graduate of the spin of graduatic of the graduatic for the second with the contract of the graduatic of the graduate with the first of the second of the second graduatic of the second second second of the second of the graduatic second and the second second second

theing the last sectority year, my bracking west like . . . . dullanging, for, land, straiding, anarging, theoghwaveling

\$999 3208 20036

#### **Appendix 6 – ANLP Conference Presentation PowerPoint**



















 Symbols
 Image: construction of construction

 Larly stages of construction of construction of construction
 To identify how Clean Language and Symbolic modelling will help

 Actival Research Strategy
 Setterning of construction

 30-15 regressiont
 Interviewend clean of construction

 Charly stages of construction
 Setterning will help

 Actival Research Strategy
 Setterning will help

 B-15 regression
 Setterning will help

 Charly strages of construction
 Setterning will help





#### **Appendix 7 – Metaphor Coding Categories**

METAPHOR ANALYSIS								
Sourc	CE DOMAIN	TARGET DOMAIN						
Animals (k)	Nature/Environment (k)	Problem						
Building/Constructions (k)	Person/People	Remedy						
Colours	Place / Location	Outcome						
Communications	Plants (k)							
Cooking / food (k)	Senses							
Faith / Religion	Shapes	Identity						
Food	Sizes / Measures	Emotions						
Forces (k)	Text / Stories	Beliefs						
Games and Sports (k)	Tools / Machines	Values						
Good/bad	Transport	Capabilities						
Health / Fitness (k)		Behaviour						
Human Body (k)		Environment						
Light and Darkness (k)								
Materials / Equipment								
Money and Economy (k)								
Movement / Directions (k)								

Identifies 'Source Domains' added as part of coding analysis of interview transcripts.

#### Appendix 8 – Narrative Analysis Coding Categories

NARRATIVE	E ANALYSIS
McAdams and McLean (2013)	Schwartz (2012) Values list
Agency : The degree to which individual are able to affect change in their own lives or influence others in their environment,	Achievement: personal success through demonstrating competence according to social standards
<b>Positive</b> : The degree to which individual are able to positively affect change in their own lives or influence others in their environment,	<b>Benevolence</b> : preserving and enhancing the welfare of those with whom one is in frequent personal contact (the 'in-group').
<b>Negative</b> The degree to which individual are able to positively affect change in their own lives or influence others in their environment,	<b>Conformity:</b> restraint of actions, inclinations, and impulses likely to upset or harm others and violate social expectations or norms.
Coherent Positive Resolution: The extent to which the tensions are positively resolved to produce closure/ending.	Hedonism: pleasure or sensuous gratification for oneself.
<b>Communion</b> : <i>The degree to which individuals demonstrate or experience interpersonal connection.</i>	<b>Power</b> : social status and prestige, control or dominance over people and resources.
<b>Contamination</b> : <i>Scenarios where good/positive event turns bad/ negative.</i>	Security: Safety, harmony, and stability of society, of relationships, and of self.
Exploratory Narrative Processing: The extent of self- exploration and reflection expressed by individuals.	Self-direction: independent thought and actionchoosing, creating, exploring.
Meaning Making: The degree to which the individuals learn or gleans a message from an event.	- <b>Positive</b> independent thought and actionchoosing, creating, exploring leading to positive outcomes
Redemption: The extent to which bad/emotionally negative event or circumstance leads "good" or emotionally positive outcome.	- <b>Negative</b> independent thought and actionchoosing, creating, exploring leading to negative outcome.
	Universalism: understanding, appreciation, tolerance, and protection for the welfare of people and for nature.
	Stimulation: excitement, novelty, and challenge in life.
	<b>Tradition</b> : respect, commitment, and acceptance of the customs and ideas that one's culture or religion provides.

#### Appendix 9 – Clean Language Coding Categories

CLEAN LANGUAGE							
RESPONSE TYPE	LEVEL OF CLEANNESS	TYPE OF QUESTION					
Reflective prompt	Classically Clean	Developing Questions					
Probing question	Contextually Clean	What kind of?					
	Mild/potentially Leading	Anything else about?					
	Strongly Leading	Where is / whereabout?					
		That is like what?					
	Neutral Question	Is there a relationship between?					
		When what happens?					
		When what happens?					
		Intention Questions					
		What would like to happen?					
		What needs to happen?					
		Can happen?					
		Sort and Sequence Questions					
		Then what happens happens next?					
		What happens just before-after?					
		Where would come from?					

#### Appendix 10 - Interview transcript and coding (Researcher 1 & 2)



do wherever we did the last interview but since then five realised that actually my situation built going to to the next 7 years and points to do with band-big and sampling which are sivery about the modernt learning experience wh constrain the quality of what you're delet but an good in it should be or that you're having to world a tange at all orders I do correcting really processe to make it change and functi actually made no hell a the bit more position about their because i heir in taking more control. The beam here 7 years and its we both those that gets compromised to I veolid have probably seld those kinds of things a month or so he just got to confirm that this is going to be a confidential Interview? On the come to realise the fire professionality is being compromised. Because im beginning to feed jast a bitle bit jaded and i heel that i to do and so you feel that rectarge their batteries, replicably. and his and which is quite comfortable but is utimately become and affing, that mean that I had my questing. If it have answered this question when we did our original interview, i probably would have can use the learning happening and you can see people granking that is a real juy for a teacher and and to make some changes in order taged and at that red. So that's probably not the areaser posteer she forward to the next 7 years, i don't know how that's point to come out as in the interview Timeffected shock and und though lise well it had through and downs and I would have tailed about torms mow that five got to make some changes to be a better professional sing terms in the next 7 panel" (suctory phoned or "Law of of your own personal time and the time that everybody har's what keeps up going and the flowin parts are where kry upscift land of changes

Ale are approaching the end of the academic year and there's probably here a lat guing an, a whole load of counserents, a load of bacehing, atakent rapport issues, **(1**) and saft you bit generally reflect for your throughth

fedenting back on this academic year from the produce where I are right now this minute ICs much me

Section that I meed to make some changes.

on the academic year on a whole how would you dear the it?

2.0 Every) surveiting dood helpy risch is a Ni, screening dood sorting in water cherger, hereger, an comport zone had perform and price correction data depaying supects of the just in terms of working with charts and serving people program and grow had then them there? The specificati, There is under according to all indexed francing is the control of and the different proventy and we de constituent proves that a second francing is the control of a production of we control the spectrum by the the control to the branes, this is great place of work brances we're all should serving and francing and the future of it to be branes, this is great place of work brances we're all should serving and developing and the future control to the branes, this is great place of work brances we're all should serving and developing and their hear control to the branes. This is great place of work brances we're all should serving and developing and their hear control to the product and the activity has given me a lot of apportantian to do that Matthiad proves the applier of branes able research partial, the furth theory has the contraction to hear more and first the applier of branes able research partial. There is a brank about the next hear part of my percendity that the hear regioned but because is not have the resonabled from their each part of my percendity is hear regioned because if in supporting them is of other people in their cont part of my percendity is hear regioned because if in supporting the of an editor people in their control of the cont part of my percendity is hear.

its in terms of what you've just said the picked up on three things you've said, as something about being, stack in a plateau and heing stack in a ruck.

## That's ready interesting that language har's 87

1968 An relation to their, fixed things poor/se reservationed. In making changes and weeking to grave accl week to date threas three chemics just go in-jointness first and wee which that means for you and here their fairs poor and then when chemics you week the median week that carrying help publicing poor professionalism relasource parts. I there they are excitent worth you've picked box the box the picken between a picken worth out worth you've picked box they are a the picken thread picket box and it then all using the word pickens. If high are it was picked at picket box are the picket box and picket box are picket box and picket box are picket box ar



It the null being C.M., Induction properation, Classing preparation, got Lands of pengrad work to do over the narrow, then the going to be lands to transledge pletter specie gounger and data and therefores and (S. Brauk publicity) the model of a formula why then person i can give you a work you wight not pet, if a law is according to the model of a formula why this pythological B., duals and not no you under the pertamine an the correng according period period is a stress at the previous accordence, your and followed well for some as the correng according period period be activated in a M. Ariv or exactly true because the coulds of our pion means that unsequenced through heighers, if a ready country, you mean all otherese to the of people bid broad broad according tended, and according and all.

One word?fm\_factuating on it hency starts for a rule and then the being on the platency in lennes of being star It o rul it there are platency more about this particular rul and the munitime? If it too correlateds, find it a graditional because so much should regist in welly fertants and i completely well paid for what i do, i work with a great banch of people, gat a mediy people and they more there's basis of three that are welly welly people does it is 250 any 150 for gat, it's were combined, there's basis of three that are welly welly people does it is 250 any 150 for gat, it's were combined went's so convertine them it mouth a great banch of people, gat a mediy people and they mould went's so convertine them it's a great banch of people, gat a mediy people and they mould went's so convertine them it's and they media is integrated and gate people. To all the them they but the three people to get crust with menuphers new, (b) like ballegin this completely that it is them they completely the gategraph of people is people and the data ballegin this completely people. To the other table the gategraph of the gategraph forms in family, the only of gater and gategraph to people change this, when i dom's work and that well is not, if only, if only if gater and families and families to allow when i get to refer a not bank and it only, if only if gater and the data that and the other, so when's occurring to me in the data when i moute and the data metaphone before, is when's occurring to me is the data theory and the node of the using physical metaphone for metaphone before, is a later of these results on any discustion with you because an tabled about metaphone for a physical metaphone are about provident and if in using physical metaphone for

memphons before, is a bit of these reception are about perception and the unitg physical metaphon for perception, when the heat carefulates and concerty are the attraction (carefulare other opportundes). It can't are any other ways of experiencing. An each heat a the become, well before first memory, the percenpary my mortpage soften action a position to hears work terrections, and before first memory. I we percenphysics mortpage soften action a position to hears work terrections, and before first memory of a softeneous part or short the superior of a position to hears work terrections, if due to deary the interview, it another an experiment might be efficient to a root practical resume for it has been a dependent an experiment might be efficient to a soft the most practical resume for it has been a dependent an experiment might be a bit mean of a risk them.

So just picking up an acree Philg else, you first hereise in terres of heing comfortable and score of the reasons behad that.



### being comfortable but disuttified.

(of 10hild 1htd field phones you used were bitridy and howley that upone and but's where the tender is unring from, their you field comfortable at the same intre-you've got this third and need same upons to briefly and grow. Very end of wher pair's just such a work revisor any more an derithen? (pair) become what event an any applicit gang resold and reconding calleds. I don't brown what this has get to do with reflection but (this), the francescion is in our employed here at the University in an academic, durits when car jubic, that's when we way, receil and enclosed that it is a stacking university in an academic, durits when car jubic, that's when we way to control and that it is a stacking university in an academic, durits when car jubic, that's the francescion is the Hawking the stackers the administrative sets is very bechangered to there is a applicant tracking real and that's large the administrative sets is any bolomoure but my thermosting real and its interaction that's when it and the durit when's when's thermosting real and its instructing real because that's when it and to do. Its hear when they monopy to do that the annual are supporting at the transition that's when it and to do. Its hear when they menope to do they because it are to achieve that if in section it that research. I'm supporting you by doing the interview, i've get the annual are supporting the tran then you're not doing b, the you're helping others doil and it heapy to be helping or when.

(art) move on its the plateau. So yes tack about the plateau and being an the high and an the iteratoryes, what do you see around this iteratorized? This is going back to reflecting can the year. I are assument, lithcox summerses is part of the job and when i way assument i mean musicy wah) also of membi capacity to do anything situa at the end of the day. I are remaping at defens as indeficited, and this is when guide a lot of them is absorbed but not waly catelogued. I are that you of my job i dan't want to be careby do but if we is absorbed but not waly catelogued. I are that you're for any lot i dan't want to be careby do for the non-more products and particularly where you may lot of the first to be with you for the non-however integ as i use musing the catelogued. I are that you're for a build with you're got a student who for whatever integ as i use musing the catelogued. I are the part of my jon's got to be with you for the non-however integ as i use musing the catelogued. I we may teaching rule backgradually mousted an annuplete statistical to be cause that is way here achieved any teaching runcurum gives the top the the non-however integ as i use musing the constraints and rearry indicating the statistical frames plant terms don by regions that a way here achieved reary tablecting runcurum gives that an Understructure for the home terms of the law where the active day rule to artifue us have been for the function by regressing i do the through to be top and rear adjoint and reaching has not Understructure on the reacted on the statement on the parts and reacted reaching users are the top and the rund on by regressing i do the statement and artifue do artifue to restrict users that an Understructure on the statement and artifue and the statement and the statement with the statement on the statement and the artifue to artifue to artifue to the function of the statement on the statement of a statement on artifue users have the statement on the statement on the statement on the statement of a statement on artifue to artifue to a the statement on the statement on the statement on the statement on the statement is and restruct and ar



No is a paraces, that but how I use it as all, issuit as a furdamental part of my job that it registed and I kinds of things, there inc't wally a proper break in the summer, it have got postgrad students who i'm goin outhnow hope holy (2 hed better about the work after than, 10mil Net The peer egging 3, its targing that to have to give some dedicated time through the summer so that's what the bindicage is. Although it may locage as a landscape recher often a hot of jobs Unleases, I use all that ceganizing shalf and I as just the signanoise of it. For going to have to now, what's nears i carb recence argedrare elle becaus l'in quite expensive non for another university to employ yery summer, no not every summer but maple last summer and definitely this summer I suid right fin that's a worry. I've get CABs first, us I've get assument, CABs, Cauring, planning, induction, all those coming up now. Clearing that's the nearest thing, free patternistics around recruitment which is be so and to get some tensarch done this summer but the samener gass to quickly and it's distant up wit and they he only going to each to employ someone who can other the full package as it each they do some de upa for PN I BUR FOW I'M LARKING IT TO FIRM richts that file trying to place

So in here of the institucion that you founded and the list of things, the CABs, the parenersets, the statest apport, the tracking, as reasonsh exact an there.

No if his setsify not premiver and theri hay problem i should be matery that promivers and i break the color colorgian out, lighted derifations they problem is should be matery that promivers and i lore out any pole to the materiane and such that any problem with a support bit type and they conset per the last respecters they can and i see that any problem with a promised prior if also use that an my receil coloration as homen being and (1966k) problem with more color and per all the firm out the last respecters they can and i see that any problem with these colorations are interested in the coloration as homen being and (1966k) problem with these colors and period in the firm out them i can maple do a bit of reading, not the other way around as if the net firm of these colors is a which much important so firm pange to do that and then when the dows that if the an unith a suffic is in proteits also other. is in terms of this policitiation and you tail about potence, do you think there's a relationship between the of and the plateau?

Ten they're the same thing, bud ff's ready multi preventing the way poulte plotting up on this because the other event, the nutlin is contracted space and is fourneephologic whereas its plotters is an open upone where i car steer the bandwape and (car use whet's giving on bud both of them are about copting). It is aloud some, bud without wenting to get som philosophical was are coptions at our lifeting in the just and hy bigger soft and down it any modipute bud is non-some to bes, fit in soft them is the start of the just and hy bigger



Ko year teriological bit about the nut and the plateau them the halo other themen we plated up are were proving and interrupt, as if we aren to put there in a sequence what would core if but, provi and change as change and proving

year. It would have to be a year 1 think because of the way the academic year works, this is my wish, I don't will centred about it I would like to just take a year out like a gap year, fol like to take a year out and fol like you learn through being somewhere, experienting something so I guess a bit of travel but not travel for to pad be satisfied that five got a nise job, some of it fire good at, some of it i could be belier at builthuf's the don't want to cause appresenton for any of my colleaguen who are all very lovely, but #1/m just being read un they are in parallel, because you thereas through greeth and greeth is a thereas, they're kind enserhowith going to huppen is worker thing but this is my wish, for quite huppy here i don't recensed scholarship, That's what PC like me thing. What fo firs to do it I would like to find a strategy where it can just map off major for a ward to beare here but i thek i need to make some kind of radical thange just to shake things up. So my to resist some of the locations where I did my original PED research, and some other piaces as well that buil think for two means, their is get metal of the ray, get me off the plateau, now it regist the that I every year the game must year hit be different, read year fil have more time and it never happens so i the hop back or, or to first do you know what the nut is olday, the planau is of and to confortal with would be to test leave anybody in the lasth and plan sheed iden't want to jurg ship and i contain are insted to frings that live been teading about because itteric the book learning only taken you so far, welvy good if m going right back there has acleased with have seen what's beyond, so that's one muces. cent take, there for a specific purpose and then spend the rest of the year writing up. So basically just off the pletess and ligs partients the witherman and Prick do you know what that plateau was well. to do because I can't do it while I'm here, and because I'm so ignorant I'n token the Phi long to makes turns for all of us, and just count my bleakely abit more to five got those two options going on frees. sking a usbitetical year and i thread i want to do they, i den't know if can this is just a fartury at the inc the only way it can happen in for the to actually, well five got two choices as for as I see R, one is to campletely and redically change my professional experience by just hupping of the haredee wheel und the Pregnesensies and the good Prepared block utaried agains one and

So think about these has choices, if you wen'to itiok ahead to the start of the real acceleratic year and you find to reflect back on these two choices what would you be thinking?

If you're aiking the to reflect forward on to next academic year?



## So just may it's now first week in September

of air colorgues are healing the same way because with seeing people hill by the wayeds, ( don't want to my teaching ham't been way good it have't cared as reach as I ought to have and fruch the first year it's and of how fire but i strugged a lot throughout the whole year and my smergy has been bie which means M. I don't feel that about the rest of my earling career, I feel that I just want to get my energy back. My "In their, them's tink so long that a perior can go on under these discutations, and we know that man make a charge. You maked the looking forward and what fim suging is related, other than looking horan Nocoos. Well that's the response both, introdutio a where the had a guarcharge because ido evice my job overall i enjoy my job a lot but "m answering your way first queetion and this particularyour the registed. I don't want to leave the profession or arything like that, but i thick I could be ready good at to one of these people that but by the wayable to first agreement, the got to take the responsibility to and going of years first looking for word to meeting bots of new maderits, first looking forward to improvin poor but this perturbin year (don't the allever waity gat off the starting blacks and ( this that) buts Altered and your section of the grant and the grant and down and and the tool and you come of the my teaching. I'm just tealing thed about next year already, I'm thraking ough and that lan't a good place aging in that's have i had no i don't had for had been been my job i think i have been maily good at my job but i need recharging. that's what i feel, is fits your i good and i man 14%. I need pl

We is failed about the judices and the low low internand and an 'or reveal on its inking about biring thed and Seling much apply to nee there's a interface/by between not and judices and vaches/big.

# ex there is because firs seeing the same thing, freed an advecture to brighten me

What kind of adventure mapt that het



the reacon why di Kh indegeni but if is consertualised is poly budi to the point that we are employed an condenity, we are employed as people who people incodedge and then's a limited amount of broakedge concerns philip beneficial what fits trying to any is fine come (pathe patheta) reached being as fire and at the deal de a standard at the trying to any is fine come (pathe patheta) reached being as fire and at the deal de a standard is an infraction and an aneater patheta market from the deal to change out the and then come test to the deal and method and houring uption like 1 and 3 years up.

# to it's come out of this and go into the Jornit beyond the landscape

reare so, recruitment for PZ & disping, disping, disping which is deadle edge neard, of course P1 rang te much much beter alvaiver flacts. Sol don't their that religious studies, between you and rea l'minut howing peulith benediered inset to thange solith almost quite looking toward to that charge whatever ope with charge ready well as homen beings to it's correcting that will be survivable and actually it might wrrts of the subject. I don't hell that as an academic discipitive if's either understood or particularly value dead Plenetied, and ward to learn for its own use they want to know they can get a 70 is then eau for but it's gale normal for academics to have subbaticals, I don't think actually what for saying is not an betany stage. I think reflecting on this year as well all the things that I'm biseversed in leading about promiting because that's my reason for being here but at the same time. It's also guin a relief because it uan't and l've tried to get that into constation and if's not happening us l'in left in the position where the unused thing at all, intrivial its a wally normal thing for the nations you and i know they're not happening for un to if my employer worth do it for me five got to do it for repuell. I just don't know how i'm golig to I, change is always highering particularly when we don't know what we're going to change into but we othout doing any work and that is quite exhausting. The other thing that i'm really interested is learning und it or how i'm going to converse. Childres or any of these practical things, up i'm at the Meskings stage convirced that the school gets it and gets what it's about, it's been this quicky thing that's been baland or out their been de because we've had the numbers and so we've pooled on and if's been the but that's he school is not interested in tearing about sol are been a bit diskosted and their itsel is poing too tot ifs sever been properly imaginated and people think we've a bit must, we might be a bit mud but in he undergraduum course goes it means my situation here will have to change and as the discourse it we thege that interest the academics by any 1 interesting to the community that fits working on and dout and teaching is variationability and the school's get no interest in their ski, it pretends it has but it hardy dending now. Also I'm seeing in the studemt that we're recruiting with very few exceptions and his passion, there's no fine in their belies so I don't feel it a fin tracking students who are real had's a lettic bit of an exhausting position to be in as well. That's a combination of staff,



This is my summary of what i thick has been said and feel (her is correct new. So is here af here you (new new bounds) the end of the condensity year so you've been study its a weat here's a which heat of yearcone why that might by, some a bill previous, some of it professions, some of it is terms of your word situation, there's scientified about polations as well and science is the landscape which you are very (here for a weat with and profity much attaced to an well and scientified about you indicates which you are wery (here for some what's there and see here that could (here here and access is the indicates as well. You've put some share bound how that could potentially happen, what you would like to do a inter forent to see what's interplat, do news reacting and share at here, you would like to do a in that forent an well, explore the interplat, do news reacting and share at here, has non-opel and at that forent is and and there and present and both work in porticit with you.

#### That's the gap analysis.

So i suppose what we have and? each to get your thoughts on this is in a serie is supery with how things are and suppose that in a bit more detail and briefly included of how you would like things in he and the metaphens around that as well, dut (i was to ask provider your impresions how would that come out? I reliad their's a very very accurate unerwary. I guess my find connect is shock, my find indicative summers confine things that we've subset shock in the succession of my preferences for, no is me say in difference, it's about my identity soft i think about my identity, is my identity preferences for the discussion that popervoard the or my identity soft if you've do all those priorities if trappine and in the discussion that poated i hund been having now, if wyperferences and percention that appending presentation of my to the or my pervoard the or my identity and its and with the discussion that postand i hund been having now, if wyperferences and include the tapletion and in the discussion that poated i hund been having now, if wyperferences and it think maybe into the discussion that poated i hund been tablet for the strain prior tabletion the tapletion and is no our standhrower to much reflect on this quark after it is a supressive theory does independent while no our standhrower to much discussion that is not a supressive theore it caped with while we does in the to our wately are stalled about directly. If it way difficultier me to supress these their action out is not waite the standard about directly. If it way difficultier me to supress these theorem the standard to the theorem and and standing point which is that it any entries theorem the state after the tension of the state standards to the it is negret from a state while of divertity that theorem is then are stated to the state best in that is any form is would then the what should first a special or the states is a described as the state independent would that a the state at an instates is a priori for a special would be the state of described to what should it point a priorite at the states is a special pure independent would that the state at the state at the special special states.

# impose the puertion for you would be who are ! then?

Who smill ther? Blook? It give the all-that thee time to need, do I need to be in an academic position to werly and endorse my instalencial interests.



01.10

2 100

3

Or could if be done without housing that accelerate bloothy?

Yes but i these if's reportent, the thereaf for your research is reflection and **(chear) testion (c)** you take the reflection, and because if do philosophy invited a loc, for pretry pood or reflecting and if do offici back not at all infragrantly should well actually this while dou't for got should may be? It actually about my identity and who is any, so again that's why if you're not happy with what you're got, thenge it.

• 219 West approve but ballow the as The Ashersoning nyb
nyb · At Sam · Superved at • The Robert Josef •Alexaniy •Redi, Walilaa, Bedgarari w desider-Bridge
And Decomposition
Britson
Charterson
< Nature & Boltzmann Constantly/Chan
 complian · historydae Interplate
Provide
Resplate
Re

end-alizaba

Ödmart Robberheidsbere

z.

	· Castronally Chara		
	· Second	a receiver an entit	
<ul> <li>translats</li> </ul>			
· Course			
<ul> <li>Rests</li> </ul>			
· Adding & Constructions			
	+ 504 The New Ave.		
· DOLMAR INST.			
<ul> <li>Consultation</li> </ul>			
· Building Concilions			
# To kinety			
· Simulation and second			
· Barrick adjustices			
· Malazz			
			" Tak mathin Colomat
			· Park, Walker, Solgand · Optimized Directory
<ul> <li>Couldad out or</li> </ul>			
<ul> <li>The Capabilities</li> </ul>			
		- T1 Englishment	
		- Ti Englishment Robert Barby	
· SQL Whereas a sharehold a			
<ul> <li>Not tracking during and set.</li> </ul>			
<ul> <li>Nor a bootdore</li> </ul>			
a 7			
			* 11 × 10 × 10
			. PL Failings
		• Sec	
		To be a second	
· Infini Property		and a second sec	
			· Knowly
<ul> <li>Control du Control</li> </ul>			
	· Cree Garater		
Kult & Brocks			
	Annual Annual		terminal strength str

#### Interview Transcript & Coding (Researcher 2)

academic heagthy for new chuterics inducting them into the university when they first come here and so its very policy and procedure led. Solit tends to be rather district and therefore quite science and dry. That's they're got placement then they go under freess to practice, it's all that professional-new and so wally thisk that's gate Hearening, it depends what the subject is of course. Some of my lecturing is around den't their that the sort of inducing that would be helpful for Mo's meandy, that's new dideck to prepare them for the fact that they're going to be writing assignments, if they're on courses where with questions and areasen at the and and Paris R, R's way checkel. The other type of lecturing 1 do

It good to aik a question during the lecture then the charces are there are other people withing to aik the that's a strong indicator to me that statients are boarcing deas of with me so it's not a formal acture with ago that one wants to elaborate on or get more information from me and the other k i think if one pensor and arraws section. What I do h as I'm guing this please heal these to ask queriors as I go through it and I therefore is very much based on teacher training in further education lifelong learning. I give lectures on the carriculum design, it links up with historical developments and again? lecture on the professional and staing quantitions, where they are obviously thready things through an I go through my section I think that an opportunity for questions and answers of the end that i like its actually a process interacting with the the tracking and its many modules as and and (their flat the any that) are they work out seels in the their I do that for increasors are because its very easy to forget what the question was 20/2 minutes way that loperate. I don't give a lecture and then at the and of it ask the questions and have a question turne question. I should that's my bastreader, I should where the students are attention, where they are

what I think is a strength of the way that I work with a group of stations. Obviously If it's a smaler group, if ruderts. I their there is a danger with a formal lectarer to be clearal and no interaction at all so that's

Crinet a hundred students for example if it is group of 23 in the soom then obviously that's more

Marten

() (and casi you that question if we throw it on to the table, when you're teaching or incluring at your best when it's working well, just to explore that a little AL. Would you call it treaching at Anthrong' I don't mixed, or presenting, i tand to do more presentations now or whole group presentations taffer fran amail group work these days because of the nature of my job. Nou're likely to give indicators when i know things are going well.

# fin and educh that like for you when that's working well.

MH Karolia

childere contrived isaming where around the activity-ordention by me and them petiting the groups to thick thege and problem value ready. As then't the method that's used in leasting in further education calleges, that makes serve. So thet's more personal and (thick that's essier to manage and again it's the same environment, it's the response from the students, it's the perfugin the any i work with them i like to be a bit humorout, build up a relationship with them perfusion error as we go through the lecture.

do put them at eace and build a relationaby with them and a 1814 bit humanous, the borometer is Hey're attention, assing guerations, thoking through and relevanting with the abulents. And interacting with each other, I think their important so that hell they've not just working through me but working with each other in discussion. And you're interacting with the students and they're interacting with each other. What kind of interacting is then?

In what sense, do you mean how it works or giving mamping?

# When you're interacting with them what kind of interacting is that?

I suppose the formal way of describing it is firm isoding on discussion, fire amount channing things. Perfociating if firs a controunness is main hour wire looking at and if ther year barn on placement and they con relate to their placement, their experience, then it may be their that dog discussion between them and giving them perceptuser giving them examples that they implicit discipled in hours whether they've been working with difficult students and how they constraine it or if it was a challenge to them what strategies they regist use in the fature to support them on their rest placement. So in a sense it thick in an intertaing they regist use in the fature to support them on their rest placement. So in a sense it thick if its party a instance, perhaps starting of the sension, i then their liberane a facture at the convertation from and facture, perhaps starting of the sension, i than their discuss and ideas to their the convertation from and factures.

# to party lecture starting eff. then bolitate, discussion from and choining on discusion.

Yes chairing is probably a very formal way of daing it huir ( perhaps it's another form of facilitation, perhaps saking questions back or axising somebody else what they thick about the discussions that have



been held and what their throughts are. "Syng to give everybook the opportunity to be involved in the process, **not** the peakers are people but to give other people are sportunity to be involved in the where they're on different tables which a tot of my leadness are actually in a diamone environment, to any well what doen table 5 table or what doen the table in the comme thing about these thermoush then been raised, so it's democratic, so everybook's got an opportunity to there ideas if not personally then through the group that they're is.

#### Share idea, facilitation

Yes definitely

## Party incluses alterting off, giving them pupil, examples as that they can came up with laters and democratic. In these amplitude also about historicity like that?

I risid probably, not so much in the instanting but if the mannysis ihuse a pile of handback, not so much in the instant parameter and case studies for the groups to book at. I skineys gave point the instantian that i might have questions and case studies for the groups to book at. I skineys are soon, the sharehout personaly, I don't ges them to attended and any there are a carry program them arrands the studients periodiarly if i don't area to of them and i don't are them regularly every week. I should with the studients periodiarly if i don't area to of them and i don't are them regularly every week. I should be studies at it is do but with a tree group of shuddents i would come strand and perhaps next are to them we interest its para that pourb here, labeling beneard to attaining the annotation and to them be provided at a labora or alternatively its might happens during the annotation and the and coloroush i would say thank you very much the your contribution them this pound and the antitreaght i would say thank you very much the your contribution them this pound and any come book to a laboral junction are also your place and its of you in hope your house the antise and which are are some sident to are a took at this question on the Powerbook and and an antise and to study and to your place of also your placement when you're an tracking praction. I think i use the a to support your and are when you blace then you're an tracking praction. I think i use the a itte to group as a contact with a group of nutlents.

### Duilding up a michlemetry and contact

Yer, Mo and I have both taught in China and I think in Interesting with true Chinese stadents that it goes up at that sort of lost **in a terms the they're physically being given a handood,** they approximate it, they are it as an horizon to receive their learning meterickin from the tation and its newshing that the done for years and



3/12

I just think if's guide nice to doit. I fuil them why i'm doing it as well so if they're waiting a while while fin running round i uny well the reason why i'm doing **this à boosue if's res correcting with each core of you** and hopefully positing you at ease if it's at the start of the lactane and it's part of the group is a serve, not standing at the hoot with a lacter-ortable in host of me as a harrier. It's because i've taught in open plan discorres when i worked in if's in Laws and at the claurooms were open plan areas, there were no wells at all and that was quite interesting and the soly way you could relate to your group that was stauted corrected when i worked in if's in Laws and at the claurooms were open plan areas, there were no wells at all and that was quite interesting and the soly why you could relate to your group that was stauted corrected when is in the noon was the physical contact to give them handouds.

# Physical conduct, agen plan and hell and building up a miletionship

Yes very much to

#### And democratic

Yes so everyone is being treated the sume, it's not as though firs handleg out to those in the host or foune in the first table and saying can you pass that down, every single student receives a handlout, a handlook or some notes or a PowerPoint, resume or whenever, everybody receives it from me not normabody else.

# And is there anything also about interacting when it's democratic, everybody's tracked the same, building that relationship and positivations, functionable

Performant I's body language and f're sawe loads bet everybody ready or a lot of people including yourself and including Me, when you take a season you very quodsy the draw fiberent somebody heats are geing them the resuges in a same or the ingroup very quodsy the draw fiberent somebody heats are geing where it is and (10% if the recent hold out is groups rather than a terms include percent backed where it is and (10% if the recent hold out is groups rather than a terms include percent backed out it there are fiber if the recent hold out is groups rather than a terms include percent backed out it there are fiber if the recent hold out is groups rather than a terms include the percent back do and, it there are fiber you want to check as the doubted, the get are opportunity to say well out and it these fiber and and it they sa with to dradge in that. I there if it ages that day an docardy (10% is their indexed, where at least so the dradge in that. I there if a point this response to all termers are find in the fiber well and it can't take a life out that they possibly can. keponsiveren and perhaps something about loods language and democratic and facilitation. Whereabouts is interactive?



I risks it probably depends on the nature of the lecture, what the lecture is about. Auit usy there is pervery formal lectures where there's no opportunity for interacting, it's policies and procedures, hat perecelly on the leader training programmes or the education index. It a searbh of opportunity whether it's government policy, whether i's behavioural management, whether it's the mains up of the sector, professionalization of the sector, free's lead of portunities for different is an about the student and a severy the formulay of information giving, and then there's that interaction through question and answer, through discussion and debates and managing that so that everyoody fresh as through question and answer, through discussion and debates and managing that so that everyoody fresh as through question and answer, through discussion and debates and managing that so that everyoody fresh as through they're getting something and of the sension. That it's just foot disactic but they're got the and distribuinformation at fructured and on a polyneation leaner. It's on different levels if you line and different argument. I use the word argument is a polyneation leaner. It's on different levels if you line and different enda.

# And nerrybody (web they be petiting something and of the section. A interacting on the inside or the outside?

Now that's interesting, I shok that's wery hard to tell interaction on the inside. Some of that interaction probably is **Not treasureed at the Streak Dicencel be same, careout be sensed at the Bres.** It could be that in the analyment some references is made to correcting that was disconted in the lectures and thought through, supported with wider reading and wider chicutation and examples of caus studies and shought through, apported with wider reading and wider chicutation and examples of caus studies and shought through as a understanding or that their combiner enough to tak about these things but it cames through as a understanding or that they fail combiner enough to tak about these things but it cames through as part of the anigmment periods is the incovinding outcures.

## And is there omything else obout interacting

No I don't think so, I can't think of ampthing at the moment.

# What do you know now about leaching when it I working well

Atteriherens of the group, enthuisen, understanding, tessared by gaestion and arease at the bines. Types of chicosicos, contridense in response and understanding.

## And what difference does knowing that make?

l thick it makes the difference between somebody making notes on something on a group making notes in a Formal leature that they might not understand what it is that they've making notes on whereas in a unafter



group which an tend to have here in the school if's a way to otherizen or if's a way of measuring learning that as the end of the semilon, maybe not everybody **but the majority of people** the picked up andersand and **beil combination, motivated,** when they leave the class to go away and start answring the activities and perhaps proparing to write part of their assignment.

# Aftern tracking it acriting and, you've like what?

Of goodby a conductor of an exchants. Maples that worl of burihers lander high risk, fire working my better and the exchants is in these and incogress the piece of neuro. Then's probably an extreme example but IN the one that's deeps in my head at the moment, it's when some of the music instruments or the maticans are out of tank and rather than itsig is a formal way and then request that piece of music or the part of a piece is actually ask for chart han itsig is a formal way and then request that piece of music or the part of a piece is actually ask for chart than itsig is a torme question, annual that piece of music or the program dentited asmething and then bringing them hack into the orchestra is continue to play the matic.

So brieghy them back into the orchesity optive them stay and repeat, call them questions and pure like a conductor of an archesita assering pour botton, the archesity is in tase and you recognise the piece of music. Ver but rather than stop and say can the brans wettion please go over that last pleas again, the last 3 or 4 burn because pourmade a pign wer of its wolf in really arroyed for you doing that worrpooly wise hous point right, rather than doing it that way to say well shall we just roop there and see if there's amploody that thesis they've out of tune in other worth down amploody werk some darifusition or harves amploody that servences worth to bring up at this point, isomaing that perhaps one arises people are not again say the plant them they don't find feddrack, they don't find as it though to be because i'm saw it wouldn't be just them, i'm sure evenyoody has got that it questions to ask but then because i'm saw it wouldn't be prove even And if is there anybody that thinks they're out of how and they don't feel isolated and there go back into it Nortony.

Yes i thek so, i thick that's the way fim thing of it at this moment in time.

# And the probability in turse and waving the bottom

Yes, I mean there's some difficult pieces within the work and it's good to know that they are not being reconsulty identified as country problems and people heaving to wheave for a lot more fore than they'd



s

planved. It's daing it is a professional marvar and professional way that encoungen then people methol to take mist, method to held confident **that they can set deep questions** an well as superficial questions, hell confident that their voices will be respected, there questions will not be putting them in a position where

Deep questions as well as superfitted questions and take risks, and confident. And not internally langer that you need to: No ther's right, to more an once everydeely? back in texes, I wouldn't say to somebook for example of you obtaiously don't understand what'tre just sait or you don't understand what this should involve siney ristri by saying well ther's a really good quarties, ther's work saiting and going in a little bit more double should whatever it is ther're saling a question, about because is loove hall well that there are obtain propile in the scores what are also singlely out of turey and so it's not just helping an individual it's helping the whole down other and works and so it's not just helping an individual it's helping the whole down the and works.

And go back to harrowy, into harrowy and nother than step and repect, ask them questions

Yes making them hed that they are probations in watches that's why they're there and yes, but like ampoor whe they'put med that encouragement, they just need that confidence that they can by committing and it is down't quite work they're not going to be in thought. Can by asserting, they're not pairy to be in truckle and a conductor of an exchanter. Anything also about that conductor of an exchanted

I think that's probably covered that

When I's not working well that's like what?

It helds arccorefortable, (bennin a server, i dom't lense if this experiences ar whether it's empethy or that correcting's difficul to understand or whether (1) just a soft serve periop, but correcting and quite right, whether (1) the body impauge of the students, whether (1) periop endings, but correcting to engoge and whether (1) a lack of anybody axiding any questions or whether (1), periop endings, but correcting to engoge and debate or discuss or give case studies, it strety huppens had there a fract free i their with instance, i suppose the heat they to devote a it is a that are strety and the soft with instance, i suppose the heat they accord a that are ender a free they a soft fract it as lack the higgest eightmane that they accord a that are not find the soft and it for it to ment and of that the higgest



Account han't never going its huggen that i think there's that here amongst knitheren periodiarly firth an area of work that you known is quite complete that cere has to markly areas the mood of the group because if is it a find of the whole class that that work that is hard to the difficulty. I fick an target prouge when it suggits groups of 5 or 4, which is larity naw now har when the not marklees with small group when it suggits groups of 5 or 4, which is larity has now har when the not marklees with small prouge when it suggits groups of 5 or 4, which is larity has now har a the not marklees at larity entities a shall be a small class that have your now har a the field with a larger date it's probably restores and will chip is and faces that have people is how who as a challenge at larity probably restores and will chip is a small class that the first the face i have first an induce to a first hard with the set of a strend date they could be upon us when it's just of working and then have to use discussion when the strend here and the day could be upon us when it's just of working and then have to use discussion and have a break and then when its discussions to perings and then have to use discussion and have a break and then when its discusses the strengen and difficult of danger the admondpanes and for the of the set includent and hand and a minode final hand a service and have a break and then when its discusses the set have when we have when where we have find hand a service admondpane is were packed in the set when when we have when where we have find hand a service admondpane is were set as the set and the set of the set and find hand a service admondpane is were when the discussion and the set of the set of the set of the set and find hand a service admondpane is were its the set at the set at the set at the set and the set of the set of

### Dispress it quickly, very very quickly.

I must adord I have to usy fith trying to thisk of an example where a known has goes wang. I don't thisk issues go wrong i think parts of them might go wang rather than say a series that's an hou and half or 43 minutes file whole thing was tention, a two difficult bits or what we field probably persists as the bad bits that we remember afterwards but i field there are wery around & i filed we've gotto fields on a feet perhaps and by and think of abere are wery around & i filed we've gotto fields on an are other ways of presenting the information and working with the group.

is adhreedine approaches and think an jour fired and titlere are ways around it and parts go wrong and it fired uncompletizede and whereasticular dans it fired encompletizable? I frick the whole group, I think it's not just the inclurer i think it's probably the class house it hink a group of stackets would pold up on that as well. Fin saw their help folds of the class broke first here we everything a good welly well because classy? I key? Here's new thing that'n got in the wey. It might be personality of the group or personalities or it could be just the could be my personality, it could be personality of the group or personalities or it could be just the could be my personality. It could be prior to the group or personalities or it could be just the context of the lecture, it's one of those tricky become that's were head to econtextly reake interacting that's quite format.



Whole class and is uncompartable, is that on the leade or the outside?

Chil thek IN on the halds definitely.

#### Aftereobouts on the inside?

I think perhaps with the lecturer.

# And whereabouts with the lectures on the halded

You mean mude the lecturer whereabouts?

,

Their's a good one, if their A probably in the head rather than a churching atomsoft, frue resum why i'm saying that is because 19964. It's the head that inspending and trying to change shings to make it work rather than in the stormuch where it, a bit like a correction at the Changew Theases in the 2000s would have that and above between at him or her, i'r, not so much the emotional as perhaps the organize have that and reach to have another approach to this antiperhaps respond as a different way. So responsible and trying its change things and regulities fact it year head and whereabords in your head is capital we food? f suppose it's that sort of will in the new of. The antennae that's supply lock that and sorting, you've got to by something different, this is celletary not working, we've got to think of a way around this.

An embersion and will in the mind and dates arcomfortable feeling have a side or a shape?

No i don't think it doer, not with me. No I don't theik i could come up with a size or shape, I mean the size thing I suppose if it's a large group it's a bighing. If it's a mailer group it's managrade and hophAfry there's the sizilit, the expertise to bring that group round within a secien. With a big group it's probably much harder to do that. Artennae and in your head and metanoding. Dying to change things, diagrame P nery nery quick) and Paink on your heat, attennative approaches. In their amphing else about all of their



21/6

MH Karolia

Perings horeony. What I mean by that is horeony with orrecal actually that warn't wery good, rather than trying to assume that it's somebody elss. (Polsk that't healthy just so that it's going to work out and do than encrosed but well proposed for measions and not take it for granted that it's going to soork out and do than another group previously or scenetizing like that, so it shok that the horeothy with orward's red possiby it durit to own it's not huppeved but possibly horeot with the grouts that for base a very good sealers for the source of that and what do you think has counsed that and to have that do than answer of that and what do you think has counsed that and its has that do the source for a series that helps the group orders than the group polity for instance polity and request and the source for helps the group order than the fourt the fourter polity and red to have the follow in a group. I don't series are when the group polity that the fourter for the rest hand the poly that are then the helps the fourt are requested that that the fourter for the rest in a first the interval or the form any or helps that the doth interval which of counse our that for the first prope, the proper is the rest interval which of counse our that for the proper and the proper is the properties of the terms of the interval which of counse our that for the proper is the term.

# Tareaty with the group and homenty with areauly. What do you know now about leaching when P1 not

#### working well

What do I know now, following the interview?

#### ,

That's a good quanties. I think the last phy i think this homenly is sumwithing that't we work just through about that in that intrastices, i would haps that i would be homent. I think because we're working in education particularly, but it in such it is other subjects as well, we're very much guaring up our intratects to be reflection, reflective particionens and i think we all reflect whether it's almost be success or when diving home how a puriticular lecture has going whet was good abourd. I, what could be draged, whit could we improve and so an and i don't suppose if we thought of it as an homenty thing and i think that is the key to it, being homent with convert and it there a well.

### What difference does knowing that makel

I think because you can do things differently need time, it might be a different group and there might be different extrament but i think is hugh longer inter if you like it's a form of almost early development, that by doing things in a contain way and greaneting things in a particular way and it hant's variand them to think about doing things differently need time activent's han ways of responding to that it think, care in cark's core reflections on that had also getting feedback it han ways of the contained the nution, i hope you've get something out the music hou's series its times a difficult data and perhaps we can just

Cartanically Chem.		
Sature & Schlenework		
manufala.		
Parente and a second seco		
Rosen		
Add Song & Constant Source		
504 TheNevily		
DOI WARKER A		
Cargonal Alars		
	· Westing Conclusion	
To klastly		
The and search		
Barnah adquelara		
Nour		
Radio Madrideos, Capilgar and		
Number and Street and		
Condition of the second		
T) - Capal-Brien		
71 Enderset		
A Distance Barlow		
DQL Were in where there is		
tech all gride as (10)		
· Nor a loost	1a	
Annual Contraction of		
* Vedice		
· Pl. Swipps		
Bernarh.		
T2 Entradior		
before Provide Local and		
(Compared of Compared of Compa		
Sandradia Chana		
fundude daare Dear Gantese Ale Tendro		
ullighted s		

How 10 minutes for any suggestions and first can be very painful of course because you're estably whing for homosty from the students as well as presult has the study and have been better if wird that some instare rooten put up to Uniform a work upor to we could read bits, if a bit of information, it's difficult to understand, dowy that's free sharek you 13 do that and read sense them it meet you 13 makes sure that all the rooten are up them a couple of weeks before the section, lowerbing the that to they can see that it how a variant are up them a couple of weeks before it is unsulon, lowerbing the that but wave possible to their and that the door with great week that everybody wants it to be a success.

Would that be a good place to leave k?

(Note that depends on the neutron, if my going that hypothetizal mode have in a neuron and if it means and an itempartual term is an and an itempartual term is an analysis of the neutron as a fitting and of the neutron is a fitting and of the neutron is a fitting and if it is the posteleneous means in the neutron after the and of the neutron as a fitting item of a table posteleneous after the and of the neutron as a fitting item of the neutron is a fitting and if it is a fitten and are really itemented in normality where people have contributed to and questions and are really itemented in normality itemeted in the neutron and we can't tay in the physical queer any locage, it fitting if if it is a that range and it is a the weat fitting in a table weat with the neutron and we can't tay in the problem any locage, it fitting it is an any set of a can't a set of the neutron and weat table with thengo. It there are a set of the neutron and weat a table and of a set of a set of the set of the answer is well being weat at the neutron and weat a set of the set of

And when it's accelling well it's like a conductor of an archeolog and when it's not working well it finds uncomfortable conditioned with pursuif and with them. Vari i tenk so fre jest geleg back to the conducter bit i septeme in a regutine, i suppose as a conductar feasing that this or cheaters is not pet seady to perform in these of the whole and regutine, it suppose as a conductant pet seath to orchester, it region for the remodulatio, if) not just one part of the whole and resplays the conductar reselts to make sum that the gauss and the score) is correctly aligned with the plans of result that the compose warried it to counce it is contain weig. I show the regions wide plans of result that the compose warried it to counce it is contain weig. I show the regions wide of the production if it can wan to say well it's not the free itset. In just there as the free and year can't blanes mellion but they don't put to the it from time then it's just there as the free wanty may back and off at it's get to be a responsibility of the conductor to make uses that the ortherism that on that and and entertand for a starter of the plans of music, and how it should be planed, can't blanes that to the dom.



#### Appendix 11 – Questionnaire Findings

1. 1	1. To what extent did the interview allow you to explore aspects of your practice?									
								Response Percent	Response Total	
1	Cons	iderably						66.67%	4	
2	To m	ost extent						16.67%	1	
3	Som	ewhat						16.67%	1	
4	Not r	nuch						0.00%	0	
5	Not at all							0.00%	0	
6	Not applicable							0.00%	0	
Ana	Analysis Mean: 1.5 Std. Deviatio		n: 0.76	Satisfaction Rate:	10	answered	6			
		Variance:	0.58	Std. Error:	0.31			skipped	0	

2. T	2. To what extent did the interview help to raise your self-awareness about your practice?											
											Response Percent	Response Total
1	С	ons	iderably								50.00%	3
2	To most extents										0.00%	0
3	Somewhat									33.33%	2	
4	N	Not much									16.67%	1
5	N	ot a	t all								0.00%	0
6	N	ot a	pplicable								0.00%	0
Ana	alys	sis	Mean:	2.17	Std. Deviatio	n:	1.21	Satisfaction Rate:	23.33		answered	6
			Variance:	1.47	Std. Error:		0.5				skipped	0
Plea	Please briefly state which aspects of practice specifically? (4)											
	1       15/05/14 11:35AM       The extent to which my idea of my personal identity relates to my professional identity         ID: 10491275       The extent to which my idea of my personal identity							nal identity				

			Response Percent	Response Total
2	11/07/14 11:06AM ID: 11246565	Although it was a. Erg supportive interview, I did not feel the raised particularly apart from the fact that I was able to use and that I do actually use metaphors in my speech without	metaphors to so	
3	24/07/14 2:15PM ID: 11447418	1. working with dissertation students 2. Admin		
1	02/03/16 8:34AM ID: 32032658	the interface between academic teaching and aspects of te vocational type course; my attitude and assumptions to and	•	

	3. To what extend did the interview help you to explore your thoughts about your practice?								
			Response Percent	Response Total					
1	Considerably		16.67%	1					
2	To most extents		33.33%	2					
3	Somewhat		50.00%	3					

	3. To what extend did the interview help you to explore your thoughts about your practice?									
									Response Percent	Response Total
4	Not much								0.00%	0
5	Not a	Not at all							0.00%	0
6	Not applicable								0.00%	0
Ana	alysis	Mean:	2.33	Std. Deviatio	n: (	).75	Satisfaction Rate:	26.67	answered	6
		Variance:	0.56	Std. Error:	(	0.3			skipped	0

3 To what extend did the interview help you to explore your thoughts about your



4. T	4. To what extent did the interview allow you to explore your emotions and feelings?										
										Response Percent	Response Total
4	Not n	nuch								0.00%	0
5	Not a	t all								0.00%	0
6	Not a	pplicable								0.00%	0
Ana	alysis	Mean:	2.17	Std. Deviatio	n: C	).69	Satisfaction Rate:	23.33		answered	6
		Variance:	0.47	Std. Error:	C	).28				skipped	0

5. To what extent did the interview help you to contextualise your reflections on your practice?

		tesponse Percent	Response Total
1	Considerably	0.00%	0
2	To most extents	50.00%	3
3	Somewhat	33.33%	2

	5. To what extent did the interview help you to contextualise your reflections on your practice?									
									Response Percent	Response Total
4	Not n	nuch							16.67%	1
5	Not a	t all							0.00%	0
6	Not a	pplicable							0.00%	0
Ana	alysis	Mean:	2.67	Std. Deviatio	n:	0.75	Satisfaction Rate:	33.33	answered	6
		Variance:	0.56	Std. Error:		0.3			skipped	0

6. To what extent did the interview allow you to identify changes that could be made to your practice?

		Response Percent	Response Total
1	Considerably	0.00%	0
2	To most extents	50.00%	3

	b. To what extent did the interview allow you to identify changes that could be made to your practice?									
								Response Percent	Response Total	
3	Som	ewhat						16.67%	1	
4	Not r	nuch						33.33%	2	
5	Not a	at all						0.00%	0	
6	Not a	applicable						0.00%	0	
Ana	lysis	Mean:	2.83	Std. Deviatior	n: 0.9	Satisfaction Rate:	36.67	answered	6	
		Variance:	0.81	Std. Error:	0.37		<u> </u>	skipped	0	

6. To what extent did the interview allow you to identify changes that could be made to



7. H	7. How helpful did you find the range of questions asked during the interview?									
									Response Percent	Response Total
3	Some	ewhat							33.33%	2
4	Not n	nuch							0.00%	0
5	Not a	it all							0.00%	0
6	Not a	pplicable							0.00%	0
Ana	alysis	Mean:	2.17	Std. Deviatio	n:	0.69	Satisfaction Rate:	23.33	answered	6
		Variance:	0.47	Std. Error:		0.28			skipped	0

8.	8. Which of the following words best describe your experience of being interviewed?					
		Response Percent	Response Total			
1	Informative	0.00%	0			
2	Confusing	0.00%	0			

8.	8. Which of the following words best describe your experience of being interviewed?							
		Response Percent	Response Total					
3	Revealing	0.00%	0					
4	Unhelpful	0.00%	0					
5	Useful	0.00%	0					
6	Ineffective	0.00%	0					
7	Helpful	0.00%	0					
8	Uncomfortable	0.00%	0					
9	Illuminating	0.00%	0					
10	Difficult	0.00%	0					
11	Valuable	0.00%	0					
12	Awkward	0.00%	0					
13	Effective	0.00%	0					

8.	8. Which of the following words best describe your experience of being interviewed?							
			tesponse Percent	Response Total				
14	Puzzling		0.00%	0				
15	Pleasant		0.00%	0				
16	Lacked focus		0.00%	0				
17	Enjoyable		0.00%	0				
18	Laboured		0.00%	0				
19	Beneficial		0.00%	0				
20	Complicated		0.00%	0				
		а	answered	0				
			skipped	6				

	9. Please briefly describe how the range of questions asked helped or hindered your reflections on your practice?								
				Response Percent	Response Total				
1	С	pen-Ended Questic	100.00%	5					
115/05/14 11:35AM ID: 10491275The discussion was free-flowing, so one particular avenue of my professional was explored in some depth, whilst other areas were not examined									
	2 11/07/14 11:06AM I enjoyed the discussion and the interviewer was supportion was open and non-judgmental. I was therefore enabled by questions to reflect on my practice though not in a way the				the open nature of the				
	3	24/07/14 2:15PM ID: 11447418	enjoyed looking at the cards, A visual representation helped n	ne to focus m	y thoughts				
	4	02/03/16 8:34AM ID: 32032658	the questions were sufficiently open to give me plenty of scop follow up questions pointed me in the right direction to explore	s were sufficiently open to give me plenty of scope to explore my practice estions pointed me in the right direction to explore further.					
	5       24/03/16 4:44PM       The questions provided an opportunity to reflect on aspects of my practice I do not often think about in depth through being tot involved in the everyday minutiae of my role.								
				answered	5				
				skipped	1				

	10. Please briefly describe how the range of questions asked helped or hindered your reflections on your practice?					
		Response Percent	Response Total			
1	Open-Ended Question	0.00%	0			
	No answers found.					
		answered	0			
		skipped	6			

	11. To what extent did you feel the interview identified metaphors that helped you to convey your reflections on your practice?						
		Respo Perce	nse Response ent Total				
1	Considerably	16.67	% 1				
2	To most extents	33.33	% 2				
3	Somewhat	33.33	% 2				

convey your reflections on your practice?										
									Response Percent	Response Total
4	Not	much							16.67%	1
5	Not at all							0.00%	0	
6	Not applicable							0.00%	0	
Analysis		Mean:	2.5	Std. Deviation:	0.96	Satisfaction Rate:	30		answered	6
		Variance:	0.92	Std. Error:	0.39				skipped	0

11. To what extent did you feel the interview identified metaphors that helped you to

12. Please use this space to comment on any other aspect of the interview which the researcher may find helpful and inform future research interviews

				Response Percent	Response Total	
1	С	pen-Ended Questic	on	100.00%	4	
	115/05/14 11:35AM ID: 10491275The use of metaphor was a really interesting way to think about my relationship with my professional practice. I'm curious to know about the metaphors your other					

12. Please use this space to comment on any other aspect of the interview which the researcher may find helpful and inform future research interviews						
			Response Percent	Response Total		
		interviewees used, how I might relate to them, and whether I r practice differently through other people's metaphors.	night perceiv	e my own		
2	11/07/14 11:06AM ID: 11246565	Nothing really to add to what I said following the first interview.				
3	02/03/16 8:34AM ID: 32032658	it might have been helpful to give some examples of how metaphor might be used. I was a little confused to begin with as to whether we were looking at metaphors used in teaching or about teaching.				
4	24/03/16 4:44PM ID: 34157319	It may have been helpful to have been given the questions in advance.				
			answered	4		
			skipped	2		



#### Appendix 12 – NVivo Analysis of Source Domains

lovement and Directions	Senses	Human Body	Tools, Machi	Person	G	iood B
	Nature & Environment	Shapes and art	Entertain	Mater	Faith	Colours
Place or Location		Size and measure				
	Forces		Light and Da	Anim	als Ga	Te
	10,000	Building & Const	Transport an	d Mone	ey Co	iok
			Plants	Food		



#### Appendix 13 – NVivo Analysis of Target Domains



#### Appendix 14 – Analysis of use of Clean Language questions

Categories of questions	Percentage Responses	
Total questions asked: 336		cate inte
Clean Language Questions	39.88%	Que
Reflective Prompts	38.69%	
Probing Questions	21.43%	

Analysis of categories of nterview Questions

Categories of Clean Language questions	Percentage Coded	Ana the
Classically Clean	56.52%	Cle
Contextually Clean	36.23%	Lan que
Mild/leading Questions	7.25%	
Strong Leading Questions	0%	

Analysis into the use of the Clean Language questions



Questionnaire findings of participants experiences of being interviewed